

Loughborough University Institutional Repository

Information flow and product quality in human based assembly

This item was submitted to Loughborough University's Institutional Repository by the/an author.

Additional Information:

• A Doctoral Thesis. Submitted in partial fulfillment of the requirements for the award of Doctor of Philosophy of Loughborough University.

Metadata Record: https://dspace.lboro.ac.uk/2134/6293

Publisher: © Gunnar Backstrand

Please cite the published version.



This item was submitted to Loughborough's Institutional Repository (<u>https://dspace.lboro.ac.uk/</u>) by the author and is made available under the following Creative Commons Licence conditions.



For the full text of this licence, please go to: http://creativecommons.org/licenses/by-nc-nd/2.5/

Information Flow and Product Quality in Human Based Assembly

by Gunnar Bäckstrand

A Doctoral Thesis

Submitted in partial fulfilment of the requirements

for the award of

Doctor of Philosophy

of Loughborough University

November 2009

Mechanical and Manufacturing Engineering Loughborough University © Gunnar Bäckstrand (2009)

ABSTRACT

Information is an important part of the manual assembly process. Information provides the user with the means to fulfil assembly tasks so that the right quality as well as high productivity are accomplished. This thesis addresses issues connected to information and information use in a modern manual assembly environment, and how these issues affect human operators, quality and productivity. The overall objectives of the research were to gain further knowledge on how attention affects the internal reject rate, to investigate these phenomena in industrial and laboratory environments and finally to propose a suitable evaluation method to be utilised at the design stage of an information system.

Studies were performed with the purpose of investigating how the assembly personnel were affected by the information and how it affected quality and productivity. The studies were performed in an assembly plant and in the laboratory. Quantitative data collection included 10 days and nights of production where the information impact on quality was investigated. Connected to this study was a qualitative survey performed among 171 persons from the assembly personnel. The laboratory study took place during three days, approximately eight hours each day. It involved 30 persons, all experienced assembly workers from the reference assembly plant. The focus of this study was how information affected the personnel and thereby the productivity.

The findings revealed that information affected the quality rates and productivity and that this can be linked to *how* the information is presented as well as *when* the information is presented. It was possible to link these findings to the outcome of a successful information search process, and to conclude that a use of an evaluation method or work process during the product lifecycle could have made it possible to avoid some of the problems connected to the information presentation. This is the basis for a proposed pragmatic evaluation method. The method was tested as a support system during the design of a prototype user interface to be used at the pilot plant.

The major contribution of this research is the connection between attention and quality as well as the connection between attention and productivity. Knowledge regarding the importance of presenting the information at the right time must also be regarded as an important and proven contribution.

Keywords: attention, information system, human factors, manual assembly, quality, productivity

ACKNOWLEDGMENTS

This journey started, although I did not know it back then, in 1996 with my preparation for university studies, and it will not stop with the completion of my PhD. I am curious what is coming next. My own expectations on my work has sometimes made me feel insufficient and now and then I have been 100% certain that I will fail to last all the way. Only with the help from my loved ones, my dear friends, supervisors and colleagues have I accomplished something that I will be proud of for the rest of my life, but again, without help and support it would not have happened.

The support from my loved mother Gullweit, my father Dag and my grandfather Bertil (who taught me to catch trout when I was a small child; if you can catch a trout in a brook, you can do anything) has been the foundation for my work. Susanne and Jessika, who stood by my side during this process and supported me, gave me energy, inspired me, cursed me and just loved me when I needed it, thank you so much.

When I started the work with my PhD-studies I was afraid that the physical distance between Skövde and Loughborough would be hard to overcome. Thanks to Professor Keith Case at Loughborough University this has never been the case. Thank you for great supervision, enjoyable gatherings and discussion, both in Skövde and in Loughborough.

Many thanks to Dr. Dan Högberg at University of Skövde, who has spent so much time with me, my research and my thesis. He was the one that first inspired me to do a PhD, although it took some years before I actually started, and he has pushed me through the PhD process. He also became a dear friend of mine during this work. Many thanks as well to Professor Leo De Vin whom, has been a great support during my PhD, for example with my progress regarding the art of academic writing. I would like to thank the management at Volvo Powertrain Sweden as well as University of Skövde who gave me the opportunity to spend time doing a PhD. At Volvo Powertrain I would especially want to mention Kent Eriksson, Jonas Möller, the assembly personnel at the final assembly line, the management team in the assembly plant and colleagues. My relationship with Dana Dobrin started rather tensely, no one had informed her that a research project had started that would affect her department. However, our relationship has developed to a mentorship and friendship over the years. Our discussions have been

an important part of this work and have provided me with so much enthusiasm and energy. At the University of Skövde I would like to mention Dr. Mats Jägstam, whom I believe is the one that from the beginning convinced Volvo Powertrain Skövde to participate in the research work involving my PhD work. I was fortunate to be involved in research projects with Peter Thorvald at University of Skövde, with whom I had interesting and important discussions and cooperation, thank you Peter.

I have so many friends to thank for this so it might be unfair to mention just some of them, but I want to at least name a few: Ronny, Anette, Martin, thank you for all the support that you have given me during this period.

I have the great fortune to have a son, Joel. I have no idea how he has been able to put up with his father's "strange behaviour" during this period, but I am so grateful for having him in my life. To do a PhD is one, at least from my point-of-view, of the most egotistical and self-centred tasks one can get into. I hope that I from now on will be able to give him back some of the time that I spent all those evenings, weekends and nights thinking on my work instead of spending it with him and the rest of my family.

This has been the most interesting task I have ever done, and I have learnt so much during this period. If someone asks if I would do it again, I would probably say yes without hesitating. However, right now, I will take a break from writing and sleep for a while.

Skövde, Autumn 2009.

Gunnar Bäckstrand

Email: gunnar.backstrand@volvo.com (Volvo Powertrain) gunnarback@gmail.com (home)

TABLE OF CONTENTS

ABSIKACI.	DOMENTS	1111 :::		
ACKNOWLEDGMENTS				
TABLE OF C		V		
CHAPTER I	INTEXT	1 1		
		1		
1.2 PRC	JBLEM STATEMENT.	4		
1.2.1	General description			
1.2.2		0		
1.3 AIN	IS AND OBJECTIVES	1 I		
1.3.1	Hypotneses	11		
1.3.2	Kesearch objectives	12		
1.4 IND	PUSTRIAL AND ACADEMIC LINKS AND COLLABORATION	13		
	I LINE OF THESIS	14		
CHAPTER 2	ANALYSIS	16		
2.1 INI		16		
2.2 VO	LVO POWERTRAIN	23		
2.2.1	Physical flow	23		
2.2.2	Information flow	27		
2.2.3	Parameters affecting assembly complexity	31		
2.2.4	Information use			
2.2.5	Quality definitions	33		
2.2.6	Discussion	34		
2.3 OTH	HER PLANTS	34		
2.4 LIT	ERATURE REVIEW			
2.4.1	Background			
2.4.2	Defining data and information	42		
2.4.3	Situation Awareness	47		
2.4.4	Information need and demand	49		
2.4.5	Information Overload	51		
2.4.6	Attention	52		
2.4.7	Human Errors	60		
2.4.8	Literature study discussion	63		
2.5 DIS	CUSSION OF THE ANALYSIS	69		
2.6 COI	NCLUSIONS	78		
CHAPTER 3	INVESTIGATIONS	80		
3.1 INV	'ESTIGATION METHODS	81		
3.2 CAS	SE STUDY 1- INDUSTRIAL ENVIRONMENT	83		
3.2.1	Study outline	83		
3.2.2	Experimental design	86		
3.2.3	Results	98		
3.2.4	Discussion of case study results	106		
3.3 CAS	SE STUDY 2 – LABORATORY ENVIRONMENT	113		
3.3.1	Study outline	113		
3.3.2	Experimental design	114		
3.3.3	Execution	117		
3.3.4	Results	121		
3.3.5	Discussion of case study results	122		

3.4	CONCLUSIONS	
CHAPTER 4 DERIVED EVALUATION WORK METHOD		
4.1	INTRODUCTION	
4.2	COGNITIVE WORK ANALYSIS	
4.3	EVALUATION NEEDS DURING PRODUCT LIFECYCLE	
4.4	PROPOSED WORK METHOD	
4.4	4.1 Prototype User Interface	
4.5	CONCLUSIONS	
CHAPTER 5 DISCUSSION		
5.1	GENERAL REFLECTIONS	
5.2	HYPOTHESIS 1	
5.3	HYPOTHESIS 2	
CHAPTER 6 CONCLUSIONS AND FURTHER WORK		
6.1	THESIS SUMMARY	
6.2	CONTRIBUTION	
6.3	CONSIDERATIONS FOR FURTHER WORK	
6.4	CONCLUSIONS	
REFERENCES		
Appendices		
Appendix A: Layout final assembly		
Appendix B: Case study 2 - Instructions		
Appendix C: Questionnaire survey (translated from Swedish)		

CHAPTER 1 INTRODUCTION

1.1 CONTEXT

In all manufacturing, it is vital that operators and assembly personnel have the right information, at the right time and at the right place (Wilson, 1997; Kasvi et al., 2000; D'Souza and Greenstein, 2003). This information is a key component within a production unit; without it, it is impossible to produce the required products (Sander and Brombacher, 2000). The information has different meanings depending on where it is presented, to whom it is presented, when it is presented and how it is presented. The information flow, "Quality Information Flow" (Forza, 1995), i.e. the stream of information from the production of the information to its consumption, is often very complex and often the information changes its form along the line. This complexity can directly influence the product quality and the production personnel in a positive (the information is a necessity for a successful task accomplishment) as well as in a negative way (Forza, 1995). If there is a problem in the one-to-one mapping between the information presented to the personnel and the information need there will eventually be a negative effect on quality (Losee, 1990; Wilson, 1997; Lou and Huang, 2003). The knowledge and skills that the assembly personnel possess will reduce the effect on quality of poorly mapped information. However, knowledge and skills cost money and time and the need for them can negatively affect flexibility. For example flexibility to respond quickly to changes in the work environment, flexibility to handle changes without new training, etc. may all be reduced. In addition, knowledge is vulnerable to changes in the working environment. That is, if there are changes to the product or the assembly process, the personnel have to acquire knowledge and skills on how to handle these changes, and the old knowledge and skills may not be useful in the same way as before. Therefore, the organisation itself must create an information flow that can support production personnel so that a product can be produced with a minimum of effort and with maximum quality.

The main issue with data and information is that they have to add value to the product itself (Hicks, 2007). This implies that, if data and/or information can increase productivity and decrease quality problems, it can have a positive effect on the product

1

value. This can be accomplished by creating an information flow (to and from the assembler) via, for example, the graphical user interface, that makes it easier for the personnel to fulfil a task, and thereby have a positive effect on productivity, and/or influence the quality in a positive way (Johnstone and Tate, 2004). One of the key issues in the development of information flow is to have knowledge regarding the context in which the information will be used (Fidel and Pejtersen, 2004). This issue puts demands on a work method that can provide knowledge to the people involved in the development of information systems. To be able to propose a work method or to develop one, it is important to understand the needs and demands that the user has on an information system (Wilson, 1997; Bäckstrand et al., 2006a). It is therefore important to consider human limitations as well as human strengths and to understand how humans interpret and use data and information in a specific context (Byström et al., 1995).

This research focuses on humans and their interaction with data and information in a production context where human behaviour, cognition and psychology are research topics that provide understanding of where the problems identified in the industrial settings might have their roots.

The company where the major part of the research was carried out is responsible for the development and production of engines, gearboxes and driven axles within the company group, and is the world's largest manufacturer of heavy-duty diesel engines (9-18 litres volume). It produces engines that have a cylinder volume of 5, 6, 9, 11, 12, 13 or 16 litres and the engines are used in trucks, excavators, articulated haulers, boats etc. Three main processes exist at the plant:

- Casting: Engine blocks and cylinder heads are produced from raw material in the foundry.
- Machining: Engine blocks, cylinder heads, crankshafts and camshafts produced in the previous step are machined to the right dimensions and tolerances.
- Assembly: The parts from the machine shop, together with outside purchased parts, are assembled in two steps:
 - Basic Assembly: All parts that are on the inside of the engine, for example pistons, connecting rods, crankshafts and camshafts are assembled at this stage in the process.

• Final Assembly: All parts that are on the outside of the engine, for example turbo, intake and exhaust manifolds, cable harness and oil pan are assembled at this stage in the process.

The research focus in this thesis is on the Final Assembly process, although some investigations have connections to the other processes. At the outset of this research, the plant management had identified a number of parameters that will affect production complexity in the future. The most important are:

- Increased product variant range. Increased amount of customer-driven variants.
- Higher production volumes.
- Shorter product lifecycles.
- Increased or maintained numbers of short-term employees provided by manpower supply companies.

These parameters add to production complexity and will affect issues related to information flow in all parts of the plant, but how? Today, the assembly personnel at the company have problems using the information in an effective way; the number of internal rejects is an indication that supports this statement (elaborated in Section 3.2). The expected increase in the workload of the assembly personnel, will affect their interaction with the information system. The question is how? Today the common opinion among managers and technicians is that the assembly personnel "have all the information they need" to successfully perform an assembly task. If this is the case, why do errors occur?

A general problem at the assembly plant is that the personnel make assembly errors by assembling the wrong parts or omitting parts in spite of the information available. This indicates that there is a connection between assembly information and quality. At the plant, assembly information regarding variants of a specific individual engine is a key component; if this information is not provided to the assembly personnel or found by them, it is almost impossible to assemble an engine according to the specification.

3

1.2 PROBLEM STATEMENT

Several researchers have identified the correlation between human factors issues and quality (Eklund, 1997; Axelsson, 2000; Drury, 2000; Paz Barroso and Wilson, 2000; Lin et al., 2001; Hong et al., 2007). At the company studied in this research an investigation regarding quality revealed that approximately 93% of the quality problem was connected to "Manpower", with "Material", "Method" and "Machine" accounting for the remaining 7% (Figure 1.1).



Figure 1.1: The relationship between the four main internal reject categories during a *typical time period.*

This shows rather clearly that human factors is a vital part of the quality work. One of the problems with the categories is that they are on an inadequate level-of-detail, i.e. just to categorise a problem as "Manpower" is not enough. A more detailed description is needed in order to aid the rectification of the quality problem. Hence, the "Manpower" category must be divided into more detailed sub-levels (domains), possibly structured

according to "Physical Ergonomics", "Cognitive Ergonomics" and "Organisational Ergonomics" (IEA, 2009). The International Ergonomics Association (IEA) describes topics in the three domains as follows:

- Physical Ergonomics: Concerned with working postures, materials handling, repetitive movements, work related musculoskeletal disorders, workplace layout, safety and health.
- Cognitive Ergonomics: Concerned with mental workload, decision-making, skilled performance, human-computer interaction, human reliability, work stress and training.
- Organisational Ergonomics: Concerned with optimization of socio-technical systems, including their organisational structures, policies and processes.

To aid the quality improvement process in final assembly, it is considered advantageous to break down these three domains into even more detailed levels. If one looks at human factors (in general and cognitive ergonomics specifically) from a human errors perspective and relate it to the output quality, it is possible to use more detailed definitions of the problems connected to cognitive ergonomics. These definitions will be addressed later in the thesis (Section 2.4.7).

The number of internal rejects is a quality measure used to determine the level of the quality issues at the plant together with customer complaints. The focus in the thesis is on internal reject rates and why they occur. An internal reject can be defined as a quality issue found by the department responsibly for the problem and found before it reaches the customer. Depending on the product, these problems cause costs connected to the process of finding the problems (quality control etc.), reassembly of product and waste in the form of damaged product due to reassembly.

1.2.1 General description

The overall issue dealt with in this research is how humans in a production system influence quality and productivity when they use or do not use the information that is provided by the information systems. The main issues at a higher level are to understand:

- 1. Is it the information itself that possibly creates the problems? This concerns the quality of the information; is it possible to use the existing information in a way that leads to the assembly of a product with the right quality?
- 2. The user uses the information provided by the information system, but in addition to this is it possible that the user might be using other information sources that are not a part of the general information system?

It is of great importance to understand the role of the human in the assembly system. The different roles, i.e. the roles that humans in the context possess, will in different ways influence the production, the productivity and the quality. The main focus in the research is on the role "Assembly personnel" and how that role affects quality and productivity.

1.2.2 Specific description

If the assembly personnel have all the information they need, why do errors still occur? These errors sometimes lead to internal rejects. Even more seriously, sometimes errors are not recognised at the plant, which means that products not assembled according to specifications may be delivered. At the plant, the internal rejects are divided into different categories, such as leakages, wrong parts assembled, parts missing etc. A detailed description of these issues is given in Chapter 3. This research work focuses mainly on errors such as "wrong parts assembled" and "parts missing". The error definition is presented in Section 2.2.5.

This leads to the formulation of a hypothetical model that views the different states of an information seeking process (Figure 1.2). This initial hypothetical model aims to visualise an information seeking process in an assembly context. The objective for an assembly worker going through this process, is to get information that can change the person's knowledge, in this case knowledge of what and how to assemble. From a quality point of view, it is important that this process is effective in supporting the assembly worker in carrying out the predicted and proper action. Figure 1.2 will be elaborated on in Chapter 2.



Figure 1.2: A hypothetical model of an information seeking process during different sorts of assembly tasks.

The model in Figure 1.2 is divided into two modes: "Normal, High Volume Flow" and "Variant, Low Volume in a High Volume Flow".

- "Normal, High Volume Flow": In this mode (Figure 1.2), the assembly task is controlled by an engine specification that is connected to a high volume product. Depending on engine type, for example 12 or 13 litre engine, the production of high volume products can be 100% of the production at a specific time.
- "Variant, Low Volume in a High Volume Flow": In this mode (Figure 1.2), fewer engines are handled compared with the "Normal, High Volume Flow". If 90% of the total amount of engines is related to the "Normal" flow, this mode handles (in this specific example) 10% of the total amount. This 10% contains a number of variants, from one to almost infinity, depending on engine type.

"Normal, High Volume Flow". This mode contains three mental states (Figure 1.2):

- *Information Receiver* (Losee, 1997): This is the person that is exposed to the information flow, in this case one of the persons who assembles the engine.
- *Trigger*: A trigger is some kind of signal that creates an awareness of changes (Dix et al., 1998). In this case this relates to a change in the production flow, and thereby a change in the information need. Or interpreted differently, the identification of a knowledge gap regarding the next action to be performed (Wharton and Lewis, 1994).
- *Passive Reception*: The assembly personnel is more or less aware of the information flow but has no intention to act on data/information given (Wilson, 2000). If the assembly personnel for some reason miss or neglect the trigger, they will continue working in the "Normal, High Volume Flow" mode, and thereby increase the risk of an internal reject. While remaining in this mode, products will be produced according to the high volume specification.

"Variant, Low Volume in a High Volume Flow" (Figure 1.2):

At this stage in the thesis, this mode in the model is described on a rather general level. In Chapter 2, the model is further developed and used in the analysis phase as a support when describing current context and problem issues, as well as in the literature study. At this stage, the model basically illustrates that the outcome of an assembly task is dependent on a successful Information Search Process.

Assembly Flow

The assembly process for a specific workstation often starts with the arrival of an engine transported by a carrier (Figure 1.3). This arrival by itself is likely to trigger a data/information search process.



Figure 1.3: A carrier has arrived at a workstation. The computer screen on top of the carrier is part of the information system.

This process continues with the arrival and departure of a carrier/engine in seven-minute cycles. When a new engine arrives at the station, the assembly personnel are expected to start to gather information about what to assemble.

Initial observations have shown that the event "carrier arriving at the workstation" sometimes is not enough to trigger a change in state from passive to active reception. This can increase the risk that the assembly personnel miss changes in products and assembly tasks, i.e. continue working in the high volume area on low volume products (Figure 1.4). If this is the case, it is likely that an internal reject will be created, and the

assembly personnel continue working on the next engine that arrives at the workstation, and the "trigger"/assembly process starts over again.



Figure 1.4: Trigger as one important error source.

If one assumes that the personnel observe the signal/trigger and thereby change from the high volume area to the low volume area (Figure 1.4), is the workers believed to change from passively receiving data/information to actively seeking data/information.

If the mental process is successful, the information is used in a correct and sufficient way and the outcomes from the assembly process are successful, and we are back in the "High Volume"-area and the information seeking process starts over again (Figure 1.2).

One major issue covered in this research is if, and if so why, the personnel do not use the information provided by the information system? Insufficient use of information combined with complexity is already today a contributor to the creation of internal rejects. It is notable that today's production volumes would not be possible without the existing information system and that the work environment complexity is a result of the demand from the customers (Volvo Trucks etc).

From the description above is it possible to develop two research questions:

1. How can we use the information system's potential without facing problems with insufficient use of information, or in the worst case no use of the information at all?

This includes how can we predict how an information user will interact with an information system.

2. How can we minimize the (negative) influence that identified parameters connected to the Information Flow and the Information Use have on the internal reject rate?

1.3 AIMS AND OBJECTIVES

This section describes the research aims and objectives and the initial hypotheses established for the research. It also describes the industrial and academic links for the research and gives an outline of the thesis.

The two research questions presented in the previous section and in the Problem Statement section create the basis for the aim of this research:

The aim of this research is to develop an evaluation working method that can support the design of Information Flow based on product and assembly process demands and to understand the underlying causes of the existing problems.

1.3.1 Hypotheses

As a starting point of this research, the following hypothesis was established:

High production volumes together with Information Overload create a workload that is the main cause for the relatively high number of internal rejects that are due to wrong parts assembled or parts missing.

This hypothesis had to be reformulated early in the research process due to results from a study conducted in the assembly environment (Bäckstrand et al., 2005a). The results from that study influenced the establishment of two new hypotheses that worked as the foundation for the subsequent research activities.

Hypothesis 1 - 'Information seeking behaviour'

The degree to which Active Information Seeking behaviour is supported/triggered has a large influence on the number of internal rejects.

This hypothesis is based on the fact that in most of the cases, i.e. assembly tasks, the assembly personnel have enough information present in their working environment to be able to perform a correct task. Therefore, it is more likely that human behaviour in general, and Information Seeking Behaviour specifically, influence the internal reject rate and not the amount of information.

Hypothesis 2 – 'Use of evaluation methods'

The use of an evaluation work method in the design phase of an information flow will affect the internal reject rate in a positive way.

This hypothesis is based on the assumption that the assembly personnel always strive for a performance outcome that lives up to the quality demands and targets. A common opinion among management and technicians is that the amount and the relevance of the data/information present in the assembly environment is enough to perform a correct task. This indicates that there is a deviation between the need and demand connected to a specific task to be performed by the assembler and the believed need/demand from the designers' point-of-view.

1.3.2 Research objectives

The overall research objective is to study and substantiate the two hypotheses. The objective is divided into the following sub-objectives:

- To understand how the assembly personnel interact with the current information system.
- To gain knowledge of how different attention levels (i.e. active and passive) affect the rate of internal rejects.
 - To understand to some extent how different triggers affect the assembly process.
- To investigate and understand how information affects quality.
- To investigate and propose evaluation methods suitable for workplace design at different phases of the product lifecycle. The focus will be on information flow rather than graphical interface design aspects.

The research is divided into four stages:

Stage One (Thesis Chapter 2)

This stage investigates how the assembly personnel interact with the information present in their working environment.

Stage Two (Thesis Chapter 3)

This stage addresses the connection between different information providers and the internal reject rate. A study was conducted with the purpose of observing if a different information provider affected the internal reject rate. This study is detailed in Section 3.2.

Stage Three (Thesis Chapter 3)

In this stage a case study with the same purpose as in Stage Two was conducted. The main difference was that the case study took place in a controlled laboratory environment. This study is detailed in Section 3.3.

Stage Four (Thesis Chapter 4)

The focus in this stage was to study evaluation methods and their applicability and strengths in a design setting, with the overall objective that they will have positive influence on quality. Finally, a pragmatic work method is proposed including a description of information providers that could be identified by the work method.

1.4 INDUSTRIAL AND ACADEMIC LINKS AND COLLABORATION

From the start this research has been strongly connected to the industry and in this case to the industrial partner Volvo Powertrain Sweden, in Skövde, Sweden. The author has been an employee at Volvo Powertrain since 2004, and is a member of the Research and Development Department at the plant, as well as a member of the research group "User Centred Product Design"-UCPD at the University of Skövde. Volvo Powertrain Sweden develops and produces heavy diesel engines for customers within the AB Volvo-group such as Volvo Trucks, Volvo Penta and Volvo Construction Equipment. Other partners during the research were mainly the University of Skövde, supervision, Loughborough

University, affiliation and supervision, Volvo Trucks, Volvo Information Technology and Volvo Technology.

Most of the research work was conducted at the Volvo Powertrain assembly plant in Skövde and at the University of Skövde.

From April 2007 to June 2009 the author was a member of the EU-financed FP6 project MyCar (Flexible Assembly processes for the <u>Car</u> of the Third Millennium) together with participants from University of Skövde, Volvo Trucks, Chalmers University of Technology and Volvo Technology. The participants were: two PhD-students and three senior researchers from the University of Skövde, technicians from Volvo Trucks, one PhD-student and two senior researchers from Chalmers University of Technology and project leaders from Volvo Technology. The participation in the project has contributed to this research by showing that the problems existing at the Volvo Powertrain Skövde plant exist also at other companies. It also gave input to the knowledge regarding examples on how to present information in an assembly context.

The author was also involved in research and development projects as project leader, initiator or pre-study leader, in projects at Volvo Powertrain Skövde and in AB Volvogroup projects. These projects were all connected to information use and human involvement in production.

The work described in the thesis is entirely the work of the author (with the exception of Case study 2 where co-researchers were involved) and knowledge gained from leading and participating in the above projects.

1.5 OUTLINE OF THESIS

This thesis, as well as the research includes:

- Description of context: Evaluates and describes the context in which this research has been conducted.
- Problem statement: Evaluates and describes the problem area, for example how information can influence an assembly task. This includes the establishment of the research questions.
 - Aims and objectives: Hypotheses and research objectives.
- Sub-objectives, understand, and investigate: divided into 4 stages, see Section 1.3.2 for details.

- Case studies: Two major case studies were conducted.
- Evaluation working method: An evaluation work method is proposed.

Chapter 1 defines and describes the problem area and gives a general and specific presentation of the relevant issues. It presents the hypotheses that are central to the work. The chapter also describes the context in which the work was produced, as well as the conditions for the research, for example industrial and academic collaborations. The aim with the chapter is to create an understanding for the research context, but most of all to present the aims and objectives of the research.

Chapter 2 describes the physical flow as well as the information flow at the manufacturing plant studied. It also presents and describes some of the parameters that affect the complexity of the assembly process. A discussion about whether or not the problem is isolated to the specific context is presented. The discussion is based on the findings from six different assembly factories in two different countries. An investigation regarding some of the general problems connected to information use is presented. This investigation is based on findings from the literature study carried out and is connected to the problem experienced in the research context. The aim of Chapter 2 is to view some of the issues connected to information and quality from a specific as well as from a general point-of-view. The outcome from the knowledge gained in Chapter 2 is used in the studies presented in Chapter 3.

Chapter 3 presents the research methods that have been used during the research and presents arguments for why they have been used. The most important parts of Chapter 3 are the studies that have been performed.

Chapter 4 presents a work method and a proposition on when to use it and why. It also includes a discussion regarding the findings and how they can be applied in an assembly environment. This is done by presenting a prototype user interface which design is based on the proposed work method.

Chapter 5 includes general reflections and a discussion of this research and summarises some of the findings. It also presents and discusses the findings related to the two hypotheses presented in Section 1.3.1.

Chapter 6 presents the conclusions drawn from the research as well as the contributions made by this research. It also presents some of the work that can be performed in further work connected to this research.

CHAPTER 2 ANALYSIS

This chapter presents the current situation at the main research environment, Volvo Powertrain Skövde, and the challenges the company are facing, thereby detailing the conditions for the research carried out. The chapter also shares reflections and experiences gained from visiting other assembly plants and a discussion based on the findings within the research environment. It also establishes connections of those findings to existing research literature (literature survey).

2.1 INTRODUCTION

The analysis process is divided into three parallel flows (Figure 2.1).



Figure 2.1: The parallel analysis flow of Chapter 2.

The parallel analysis flows are presented as individual parts and at the end of the chapter, the conclusions from the evaluation of the research context (Section 2.2) and the literature review (Section 2.4) are summarised in a discussion and an overall conclusion.

In the first part of the chapter, the research environment is described, with the focus on the physical flow as well as the information flow at the production plant. The second part of the chapter presents findings based on a literature survey.

In Chapter 1, a model was presented (Figure 2.2), which visualises a basic behavioural model of an information seeking process. This theoretical model will serve as a foundation for this chapter.



Figure 2.2: Information Seeking Model.

Wilson (1999) describes a model (Figure 2.3), based on characteristics presented by Ellis (1989; 1993), that views the overall steps of information seeking. The purpose of this combined model is to describe the different steps in an information search process, to describe the influencing factors that can affect the outcome and to describe the factors that start the process as well as the outcomes of the process.



Figure 2.3: Information Search Process by Wilson (1999).

Wilson's model is too general and too rough for identifying the causes of assembly errors at the assembly plant. The model is however useful as a foundation for contextual investigations regarding the connection between information and internal reject rate.

If one compares Wilson's model with the information search process presented in Chapter 1, one can identify three phases (Figure 2.4) that will be discussed in more detail below.



Figure 2.4: The three major phases in an information search process.

The model that Wilson presents, based on Ellis, does not deal with issues that can explain the connection between information and internal reject rate since the model is focused on the process and not how the Information Seeking Process itself affects the output quality. To be able to understand the impact of information a more detailed model of the information search process is needed. It must describe Before, During and After the process as well as factors influencing the process. These four matters are elaborated in the following sections.

"Before"

What triggers an information search process? Are there specific goals that trigger a need? Are there signs in the assembly environment that trigger the attention, and is the reaction to that attention the start of an information search process?

In all the tasks that are performed at an assembly plant, as well as elsewhere, an overall goal must be accomplished (Vicente, 2006; Mursu et al., 2007). This is not specific to the assembly plant, and there is a strong connection between goals and actions in the decision making process (Thagard and Millgram, 1997). This overall goal can be divided into problems that must be solved and goals to be achieved, and thereby it is possible to divide the overall goal into subgoals, i.e. a goal hierarchy, achievable by, in this case, the assembly personnel (LaBerge, 1995; Thagard and Millgram, 1997). This striving to achieve a goal can be a trigger to start an information search process but there must exist other events (stimuli) in the environment that capture our attention and thereby trigger a search process. According to Badre and Wagner it is a combination of the two (goals and cues/stimuli) in our environment that makes us search for and focus on relevant information (Badre and Wagner, 2007).

"During"

How do human abilities, such as use of memory, attention (active, passive, shared, divided etc.), awareness etc. influence the outcome of an assembly task? Is decoding of information related to the processing, and is data a part of the decoding process?

An information search process can be affected by disturbances from other information sources ("distractors") in the environment that can affect our attention (Geyer et al., 2008; Narasimhan et al., 2009; Schubö, 2009). The goal of attention, to pay attention (knowing where and when) (Coull and Nobre, 1998), is to, according to LaBerge (1995), support the human in three ways: accuracy, speed, and maintenance of mental processing. However, it also supports the remembering of events in a better way and increases the awareness of occurring events (Pashler and Johnston, 1998). This is done by focusing the neural resources for recognition on a specific area in the context (Olshausen et al., 1993). The interesting parts here are attention and awareness. Is it possible that attention and awareness are two major components during a search process?

"After"

Is there a connection between the internal reject rate and the outcome and results from the information search process? Are the process and human errors connected in a way that influences the number of internal rejects?

It is problematic to divide the action of an information search process from the main process, i.e. to assemble. It is likely that the two exist in parallel. According to Sireteanu and Rettenbach (1994) it is possible for initial serial tasks to become parallel due to a training effect. It is therefore likely that a skilled assembly worker can assemble and search for information at the same time. However, if an assembly task is to be successful, a correct information search process must precede the assembly task, otherwise the outcome of that assembly process could be just pure luck since it is based on presumptions. It is likely that this process is continued during the entire assembly task, and that it starts over and over again. It starts with an observation and ends with an act, but continues once again with an observation.

"Influencing factors"

Human limitations, human demands (psychological and physiological) in the assembly environment can affect the performance of an assembly task in terms of quality and productivity.

"Human errors can only be defined with reference to human intentions or expectations" (Pennathur and Mital, 2003). The expectation of the outcome of an assembly task, or any task at all, must be that the product produced within the assembly process should possess the right quality (Vicente, 2006). There is more than one stakeholder when it comes to defining what is the right quality (Sörqvist, 1998). For the time being, the definition of the right quality is a product that is not an internal reject, i.e. it is assembled according to specifications. The intention of an information search process is to create knowledge that makes it possible for the personnel to perform an action, or at least gain knowledge, that leads to the fulfilment of the expectations (Tang and Solomon, 1998; Vicente, 2006). It is possible, from the previous discussion, to identify some issues that can have an effect on the information search process and thereby the internal reject rate. These issues are:

- Attention¹: How does attention relate to the outcome of an assembly process?
 - Is a goal and/or stimulus a predecessor to an information search process? If yes, in what way?
- Awareness²: Can awareness of an action influence the outcome of an assembly process?
- Human errors: Without doubt, an internal reject can be a product of a human error. However, there are different types of human errors. Thus, the result may be the same (an internal reject) but the cause may vary.
- Information and data: What role does the information itself play? Is it possible that there are differences in data and information and that the two influence the search process in a negative way?

In Section 2.2, the industrial related research context and its needs are presented. The existing problems are addressed and form a foundation for the literature survey. In later sections the process presented in Figure 2.2 will be further developed as a result of the knowledge gained from the literature survey. This detailed process is utilised and evaluated during the case studies (Chapter 3).

 ¹ Attention: "the action of turning one's mind to (somebody/something) or noticing (somebody/something)" Hornby, A. S. (1997). Oxford Advanced Learner's Dictionary of Current English. In Crowther, J. (ed.). Oxford: Oxford University Press.
² Aware: "having knowledge of (somebody/something)" Ibid.

2.2 VOLVO POWERTRAIN

The production lines at Volvo Powertrain Skövde are divided into parallel flows, the *physical flow* and the *information flow*. The number of parts and amount of information attached to a specific product (engine in this case) increases during the refinement of the product. The information provided to the assembly personnel varies and is connected to specific stations and specific tasks. In the following section a brief description is presented with the aim of illustrating the environment and the need for information that the assembly process creates.

2.2.1 Physical flow

The majority of the engines are assembled in a high volume main flow, (Figure 2.5 and 2.6. A larger view can be found in Appendix A). In this main flow a high volume product is assembled, but there are elements of low volume products assembled in the same product flow. High volume products are engine types that are produced in high numbers. A low volume product is an engine assembled with less frequency. These definitions are subjective and can be seen as being defined based on the assembly line characteristics and the products assembled in that specific assembly line. The assembly is divided into two major parts: *basic assembly* and *final assembly*.



Figure 2.5: Engine line, basic assembly.



Figure 2.6: Engine line, final assembly.

In basic assembly, there are elements of manual assembly but the major part of the assembly is automated.

At this stage of the assembly process the variations of the product are few due to economical and production complexity reasons. At the beginning of the assembly line only one engine block variant is used. At the first station an engine block is allocated by a customer order. This is done by identification of the block and a registration for the specific order. At the same time an order is sent to a pre-assembly station for pistons, connecting rods and to the cylinder head assembly line. The variety of pistons and rods are limited as well as the cylinder heads, but they still add complexity to the product. The product is finalised in this first phase and stored in a short-term storage. At this point the product is rather complex but the information connected to the individual engine is limited.

In final assembly most of the work is manual. This calls for differences in the information flow. Compared with basic assembly, in the final assembly much of the information is there as a support for the personnel, i.e. the information is visual on a computer screen or on an instruction sheet (Figure 2.7 and 2.8).



Figure 2.7: Example of one of the products, part of the information system (the computer screen on top of the carrier) and the carrier.



Figure 2.8: Example of a final assembly line.

The assembly line is divided into approximately twenty-three assembly stations (the layout is dynamic over time, so the total number of assembly stations can vary).

The work process for the assembly personnel starts typically with a carrier arriving at an assembly station (Figure 2.9).



Figure 2.9: Assembly work process.

This process starts over approximately every seven minutes, i.e. every seventh minute a new carrier arrives at an assembly station and the assembly process starts over again.

At the beginning of the final assembly line the engine is mounted on a carrier by an automated process. At the same time the data gathered during the basic assembly stage is uploaded to the mainframe, and information regarding the engine is downloaded to the carrier so that the information needed can be presented on the computer screen attached to the carrier. During the carrier's movement in the assembly line the information is updated at each station.

2.2.2 Information flow

The presence of low volume products within the same assembly line demands a dynamic interface and a dynamic flow concerning information transfer for individual engines. The character of this information can for example be parts to assemble on the specific engine, but it can also be information regarding how to assemble the engine.

The amount of information, connected to a specific individual engine, increases during the process of value adding activities. At the beginning of the assembly process, basic assembly, the variants are kept to low and the major reason for this are commonality demands that previous machining processes put on product design from an economical perspective.

The effect of this is the low variant range in the beginning of the assembly process that leads, according to the production technicians at the assembly plant, to the following two major effects:

- Flexibility: By keeping the variants to a few at the beginning of the assembly process a more flexible production process can be created. The parts that create a variant should hence be assembled as late as possible during the process.
- Simplicity: By keeping the variants to low, the production process can be more simplified.

Accordingly, as the product complexity increases at the later stages of the assembly process, the need for more information increases.

The user interfaces in basic assembly are rather simplified. A large amount of the information in basic assembly is not visual for the personnel. The main part of the data and information that exists or is created in basic assembly is connected to the automated process, i.e. robots etc. This information is rather similar to that used in final assembly but is not visualised in the same way. Obviously, the information is used for example by maintenance personnel, but they are not considered in this research. Typical information in basic assembly is mainly:

• Robot programs: Robot trajectories are information used by the robots. What trajectory to be used is controlled by the product variant. After the system has identified the variant, the robot decides what program is to be used for that particular engine individual.

- Control instructions: Control instructions are used by the personnel as a support in, for example the quality work.
- Assembly instructions: What to assemble. Information for the assembly personnel. This is the information used by the personnel to identify the variant and to identify what part to assemble on the individual engine.
- Data from process: This data is not visual to the personnel, but can be viewed on demand. Typical information is process data, for example cycle time, process status, error codes, etc.
- Product data: Data created during the basic assembly process. The data is stored in an escort memory³during the process. The data is read from the escort memory early in the final assembly process and is sent to a mainframe. This data is connected to a specific individual engine and follows the engine during the whole assembly process.

In the final assembly area the data and information have a different character. In basic assembly the focus is to provide the automated process with the data and information needed, whereas in final assembly the focus is on the assembly workers and their needs. The data/information is mostly viewed via the computer screens, in some cases attached to the carriers, and in other cases situated within the workstation. Instructions on how to assemble the parts on the engine are distributed in paper form and each station has its specific instructions. These instructions only exist in the assembly context in paper form.

³ An escort memory is a data storage device attached to a product. It is possible to read and store data from and to the device. This creates the possibility to store data connected to a product during the manufacturing process. This data can then be uploaded to a database and be used for example when traceability is needed.
Header	Qty.	Instruction text	OP1	OP2	OP3
21020476 SERVOPUMP	1		0		0
977030 O-RING	1				
946671 FLÄNSSKRUV M8*30	3				
Kolla flankspel på servohjulet		Enl. instr. 430arif8	0		
21023285 HUS, BR. FILTER	1		٢	×8	
946671 FLÄNSSKRUV M8*30	3				
20866609 BR.RÖR, MAIN FILTER-CYL.H	1	Bakom insug			
949747 KLAMMA	3	Till br.rör			

Figure 2.10: Example showing the information provided via the IT-system.

The main information provider is the IT-system, i.e. the computer screens (Figure 2.10). The information provided via the IT-system has three main purposes:

- Support the personnel in the decision making process regarding *what* to assemble in a specific assembly station and on what engine. That is, what specific part does an engine require.
- Support the personnel regarding the assembly process. *How* an individual engine should be assembled.
- Support the personnel regarding *when*, in what order, they should assembly the parts.

The information provided to support the decision regarding *what* to assemble focuses mainly on providing part numbers and part names (Figure 2.10). The information on the computer screen is updated as the engine arrives at the workstation. The part numbers are presented to the personnel with the purpose of aiding them in the decision making process regarding what to assemble. The part number is used on the material shelf as well (Figure 2.10) so that the personnel can find the right parts. The information provided by the information system does not support the assembler in finding the part. This is part of

the workstation training and the personnel need to learn by heart where to find the part and the part number is there to help them verify that they have found and picked the right part.



Figure 2.11: Example from a material shelf front. The part number is present on both the box containing the part and on the material shelf front itself.

The information on *how* to assemble the parts is also provided, in some sense, at the same time as the part number is presented (Figure 2.10). The system provides information to the personnel on what order the parts should be assembled and in what order the tasks, for example tightening of bolts, should be performed. Not all the steps in the processes are presented and some process information is provided via paper-based instructions located near the personnel. An example of information not presented is the orientation of hose clips as this is something that the personnel need to learn. They can also use the paper-based instructions during the assembly process to refresh their memory.

2.2.3 Parameters affecting assembly complexity

Some parameters have been identified that influence the production complexity. For some parameters is it possible to see distinct trends that support this. The data shown as trends in Figure 2.12 and Figure 2.13 was gathered at the specific plant. This is common production and human resource data that the plant management use in their daily work.

In Figure 2.12 the increased trend for two important parameters are plotted. The number of engines produced and the number of design change notes $(DCN)^4$ are parameters that can directly affect the assembly personnel and their workload. The number of DCN per year creates a much more complex work environment by its nature. The trend for the two parameters has developed in the last ten years and the management predict that this trend will continue into the future.



Figure 2.12: Trend regarding number of engines and design change notes per year.

⁴ A design change note (DCN) is a change in the engine setup, e.g. a new or a changed part that will produce a new part number or a new part. These changes can sometimes create a need for a new presentation of the assembly parts in a specific assembly station. These changes can more or less affect the assembler, in some cases the changes are so small that the assembler is not affected in any way, and sometimes the change can cause a rebuild of parts of the assembly line.



Figure 2.13: Trend regarding engines produced per employee per year (Values on the vertical axis are absent due to company confidentiality).

Figure 2.13 shows how productivity has increased for the five years 2000 to 2005. The trend in productivity is calculated by the management to continue or increase in the future. Different means, for example new technology, will make this possible but it will also create a higher demand on the production systems and thereby on the information systems as well as on people.

In an initial study performed at the company some parameters were investigated and analysed (Bäckstrand et al., 2005a). Three of them were:

- Volume.
- Time and event.
- New employees.

"Time and event" and their influence on information use and quality will be elaborated further in Chapter 3.

2.2.4 Information use

An advanced information system, combined with more traditional information providers such as paper instructions etc., is a powerful tool when it comes to providing data and information to the end user and is sometimes a necessity in a complex assembly environment (Reed, 2001; Greenough and Tjahjono, 2007). It should rapidly provide the user with correct information/data so that the user has the support he or she needs to fulfil a specific task with the right quality (Dencker et al., 1999; Greenough et al., 2001). As it is, the information use is not optimised in the specific plant. The indicator of this is the number of internal rejects caused by "wrong part assembled" and "part missing", see Section 1.2. These two quality categories are believed to be influenced by how the assembly personnel interact with the different parts of the information system.

2.2.5 Quality definitions

A fundamental part of the quality work is mapping of quality problems. This can be done in different ways but there are some stages that have to be part of the quality process. They are (Sörqvist, 1998):

- Preparations.
- Planning.
- Execution.
- Follow-up.

These four steps represent a standard way of working with quality. The interesting part for this work is not the process itself; instead it is the different types of quality problems and their definitions. These definitions are a necessary part of the quality work and create a possibility to trace the quality problem back to its origin. Once the origin is found, the work of finding the root cause can start.

There are approximately 75 different types of quality faults defined related to final assembly at the specific company. Three of them are of particular interest and they are:

- Component missing (fault code: 141).
- Wrong component assembled (fault code: 142).
- Extra component assembled (fault code: 144).

These three faults are believed by the author to be connected to an assembly error that is related to a situation where the assembler does not perceive and use the information presented via the information system in the way expected. They also together are the third highest cause of internal rejects at the plant. This will be further elaborated in Section 2.4.3 and 2.4.7.

2.2.6 Discussion

The work environment at the plant is rather complex, similar to many other assembly plants. The interesting part here is the combination of product complexity, changes in the work environment and the internal reject rate. This combination creates a demand on the information in the work context and this demand is of relevance within this work. The conclusion from the analysis of the work environment is that the information flow to and from the assembly personnel is complex and it is affected by continuous changes due to process and product changes.

2.3 OTHER PLANTS

Early in the research project eight factory visits took place. These visits were very interesting and an important part of the research, although it was not possible to conduct any in-depth investigations, for example regarding the internal reject rate. When one compares different manufacturers a reflection is that handling of information in an assembly environment is a general problem. During two of the visits there were clear events where errors (internal rejects) had occurred due to wrong part assembled. This indicated that the problems investigated at the reference plant were not isolated events. A conclusion from the visits was that the quality risk connected to the handling of information use present in all factories, as well as the problem of an inefficient use of information.

In at least in two of the plants an interesting event was observed. To be able to handle the parts to assemble the personnel had created new information on their own.

Case 1:

The task was to examine what type (thickness) of shims a specific assembly required, and to assemble the same. The assembler used a template made of steel to measure the thickness needed. The process was divided into:

- 1. Use the template to measure the thickness needed.
- 2. Find the shims on the shelves in material storage.
- 3. Assembly the shims.

The results from the template measurement were presented to the assembler in the form of parts of millimetres, i.e. 0.05; 0.10; 0.15 etc. This information then had to be transformed to a part number. This required an instruction sheet. After a while, the assemblers will of course use their knowledge regarding what measurement connects to a specific part number. The interesting aspect in this case is that instead of using the instructions the personnel had created new information providers. The solution was that the personnel made their own information by providing marker pen markings on the plastic containers that contained the shims. By doing this they would not need to use the part numbers provided by the instruction sheets. This can be seen as a sign of creativity, but the problem is that by creating their own information flow the official information system was not used and thereby a quality risk was created. It is difficult to trace possible quality problems when the personnel use an unofficial information provider.

Case 2:

Assembly of air tanks. This case has similarities with the above. As in the previous example the personnel felt a need to create new information and information providers. The assembly process in this case was:

- 1. Gather information of the final product.
 - a. Find assembly number.
 - b. From assembly number, find part numbers included in the assembly.
- 2. Gather parts to assembly.
- 3. Deliver final assembly to next station.

Before they could gather the part, the assemblers created new information in the form of hand printed tape strips that they put on the product. In the next step of the assembly process some of the assemblers had created their own information in book form. In that book they wrote information and instructions that could not be found in the work environment.

From a quality point of view this behaviour is a quality risk. From an economical point of view it is a waste. From a flexibility point of view this behaviour creates an inflexibility due to the fact that it is difficult to move or introduce a person to the assembly area. From a productivity point of view it can reduce productivity because it can affect the time it takes for a person to reach full productivity.

The problems are not always obvious to the management, but can be rather easily spotted when one studies the environment with a focus on information handling. A problem that all the plants have in common is that the information is presented in a way that does not support humans. That is, it is very difficult to understand what to do and what to assemble based on the information presented.

2.4 LITERATURE REVIEW

In this thesis, the focus is on the graphical interface which is one of four main information providers that present or transfer information to the assembly personnel. The information is presented to (or transferred to) the personnel via:

- A graphical interface, a computer monitor: When a carrier arrives at a workstation, the computer monitor is updated with information regarding what part should be assembled at the specific station.
- Paper instructions: Paper instructions are accessible in connection to the workstation. An example of instructions could be the assembly order for different parts.
- Direct communication between colleagues and managers.
- The context, for example parts already assembled on an engine.

The graphical interface and the information present informs the personnel about, among other things, what to assembly, but how does the information affect the persons that are supposed to use it? Earlier in this chapter, an information search process model was presented (Figure 2.2). As previously discussed, these three major components are not enough to describe or to understand the information search process. The main "ingredient" should be the information itself and its definition. It is possible within related literature to find several definitions regarding what is meant by *information*. One relevant issue is the distinction between *data* and *information*, and the importance of such. Is it possible that information presented on a computer screen is data and not information, and if so, does it matter? This is elaborated in Section 2.4.2.

The literature review presents and discusses components within an information search process. The main purpose is to try to find existing theories that can explain the problems connected to the information flow and the internal reject rate.

2.4.1 Background

As mentioned earlier, a common opinion among managers and technicians is that the assembly personnel have all the information needed in their work context. One can always argue if this is a fact or not. On the other hand, if this statement is true there

should not be any errors made because of a lack of information. Still, errors occur that seem to be an effect of a lack of information. How is this possible? The literature investigation started from a standpoint that states:

- All information needed is available in the work context.
- The personnel always strive to assemble the product in a correct way.
- The personnel have the skills needed to assembly the product in a correct way.

The first part of the standpoint had to be investigated within the final assembly area to confirm that this was indeed the case. The investigation performed by the researcher showed that yes, all information existed. The investigation also revealed that there were some questions regarding how this information was handled by the assembly personnel. By handled, the researcher means that if a process exists that defines and explains how a human makes decisions leading to an execution of an action then the process should explain how the decision-making process is supported by the information presented to and used by the assembly personnel. To be able to make decisions based on information a basic criterion is to understand. To be able to draw any conclusions from information one has to understand the meaning. This made it relevant to investigate what is *information*. Is it possible that what is information for one process, e.g. assembly, quality control etc., is not for another, and if so, if it is not information what is it?

An interesting research field identified is Information Literacy. The term literacy can be defined according to Oxford Advanced Learner's Dictionary (1995) as the "ability to read and write" and information literacy can be described according to Julien and Barker (2009) as "the set of skills required to identify information sources, access information, evaluate it, and use it effectively, efficiently, and ethically". Information Literacy originating from the library profession in general and bibliography in particular, and it has produced conceptual models for information seeking and retrieval (Marcum, 2002). One common standpoint for these models, for example (Kuhlthau et al., 2008), is that they are based on:

- A subjective opinion (by the one that searches for information) that a gap in knowledge can be filled by information.
- That a person actively searches for information; that a need exists.

Carol Kuhlthau presents an Information Search Process (for example (Kuhlthau et al., 2008)) that she applies to students and their search for information connected to specific student assignments. This search process can be applied to this work as well. It is divided into five stages, briefly presented here:

- Initiation: At this stage an assignment is announced.
- Selection: A topic is selected by the student.
- Exploration: Explore information to learn about the topic.
- Formulation: At this stage the student forms a focus, for example a hypothesis or an angle.
- Collection: In stage five the student collects information on their focus.
- Preparation: This stage prepares the students to write.

The process includes parts that are of interest, but it can not be directly connected to the output quality from an assembly process. It includes all the components necessary, although on a very high level, for a successful search but it only handles the "During" stage of the model presented in Figure 2.4.

The quality definition presented in Section 2.2.5 for one of the internal rejects, "Wrong components assembled", indicates that the personnel have:

- 1. Not observed the information presented.
- 2. Misinterpreted the information.
- 3. Used the wrong information sources.
- 4. Intentionally neglected the message.

"Not observed the information presented" should be an effect of a non-started information search process or at least a search process that relies on information provided from an external source, and not an internal memory source. And a non-observation of external information should increase the risk that one assembles the wrong part.

This reasoning indicates that before one can start an information search process, one has to be aware of the fact that the need for information exists and that this information is crucial for the output quality of an assembled product. If one fails to observe the information presented, the other problems (2-4 listed previously) will not occur, so the focus at this stage is to understand what starts or does not start the information search process.

This conclusion highlights that there must be something missing in the process described by Kuhlthau, at least from a general perspective. The purpose, at this stage, is not to evaluate if the Kuhlthau process is correct or relevant in this context, it describes very well how students work. The purpose is to describe the reasoning, conclusions and assumptions that have been made as a preparation phase for the literature review. A reflection after studying information search processes was that it seemed that these processes all relied on the idea that a user actively searches for information, for example a student searching for information for an assignment, a person trying to use an ATM (automated teller machine) etc. Based on this argument, it seems that this way of using information is not applicable in the research environment. The reason for this is that although the information is sometimes hard to understand, it is present in the work context. The conclusion is that there must be something that starts an information search process and if this is not present the affect is that no information is "consumed" and this can lead to an internal reject that had the effect that parts are missing on the product, or that the wrong parts were assembled. It seems that there are some keywords at this stage that have to be understood better:

- Active information searching and the opposite, passive information search.
- Attention. Can attention affect how a person perceive, observe, use etc. information? And can this be connected to a quality output?

It seems that being active when one searches for information can be a key issue, but how does one get active, and is it possible to be active all the time? Is it possible that an assembly worker searches for information during an eight hour shift, and if so, are there any drawbacks? The first step in the literature review is to investigate if there is a difference in actively and passively searching for information (a remark is that the conclusion was that if active search for information existed, then passive search should exist at least to some extent) and if there is a difference in how the two affect the information search process. This investigation leads to a review of Information Behaviour which discusses passive attention, passive and active search, and ongoing search. Based on this review, some interesting suggestions were made:

- Passive attention can be seen as an activity where information acquisition may take place without intentionally seeking it, such as watching television, listening to the radio or driving a car.
- Passive searches are "those occasions when one type of search (or other behaviour) results in the acquisition of information that happens to be relevant to the individual" (Wilson, 1997). The conclusion from this remark is that in passive mode one does not search for information, but that the acquisition of relevant or interesting information is a happening, and that it is not from the beginning of an active activity.
- Active search is when an individual searches for information that is important or relevant to them.
- Ongoing search is when an active search is established and one has a basic framework regarding knowledge, ideas etc. but where occasional continuing search is carried out to update or expand one's framework. An example could be getting more knowledge regarding a consumer product were one planning a purchase and searching the internet for more information regarding that specific product.

These points suggest that there are active and passive ways to attend to information. This might be rather obvious to the majority, but nevertheless it is important that it is known and defined. The interesting part here is that active and passive indicate that something must start an information search process, some kind of *trigger*. This trigger might change the search and attention from passive to active mode and thereby start the information search process. If this trigger starts a search process what happens when it is absent?

When the literature survey started, there were some keywords identified that were the output of the investigations that, one could call a pre-literature study, that started with the information search process defined by Kuhlthau (2008). This definition was later abandoned for the definition provided by Wilson (1997). This together with the question of how it was possible to make errors, internal rejects, despite the fact that all information existed in the work context was the foundation for the literature survey.

The keywords for the literature study were:

- Information and data.
- Active and passive information seeking.
- Active and passive attention.

- Triggers.
- Quality, the effect on.

2.4.2 Defining data and information

Data

The sentence below is a hexadecimal version of ASCII (American Standard Code for Information Interchange) code.

"68656C6C6F20776F726C64"⁵

This code is a representation of data and has to be transformed into information, in this case into words, before one can understand the meaning of the sentence. This type of code is used in computers and is transformed into words, for example presented on our computer screens. This is just an example to show the difference between data and information.

It is obvious that the data needs to be processed before it is understandable. This process is central for the transformation of data into meaningful information.

Data can be seen as:

"A representation of facts, conception or instruction in a form suitable for transfer, interpretation, or processing by humans or machines (Nationalencyklopedin, 2005). (In this case, Information can be seen as the meaning of data, which requires a receiver with the ability for data interpretation).

According to Meadow et al. (1997) data can be been seen as "a less meaningful message" and more meaningful messages "are called information".

Meadow et al. (1997) have a notation for data similar to the one regarding information. According to that, data can be seen as:

1. "A set of symbols in which the individual symbols have a potential for meaning but may not be meaningful to a given recipient."

⁵ "68656C6C6F20776F726C64" transformed gives "hello world".

- 2. "A set of symbols in which the individual symbols are known, but the combination is meaningless: the semiotics are known, the syntactics are not."
- 3. "Understandable symbols rejected by the recipient as being of no interest or value, typically because they are redundant or disbelieved."

Altar (1999) supports the opinion of Meadow et al. by stating that "*Data are facts, images or sounds that may or may not be pertinent or useful for a particular task.*" In this thesis data will be regarded similarly to the notation by Meadow et al. presented in this chapter. That is, data can be seen as a set of symbols that have a potential for meaning but the symbols must be processed into information to create meaning for the receiver.

Information

To be able to discuss information it is important to have a clear definition in mind. When one reads some of the literature regarding the definition of information one realizes that there are a number of different definitions of information.

In an early statement regarding information, as well as communication, Hartley (1927) states that there must be symbols "which by general agreement convey certain meaning to the parties communicating". Therefore, it is important that the parts that communicate, for example sender and receiver, "talk the same language", but also that they have a common concept (Kilov and Sack, 2009). Hartley writes, "for two persons who speak different languages the number of symbols available is negligible as compared with those persons who speak the same language". If this thesis were written in Swedish and not in English, it would be very hard for a person not capable of reading Swedish to understand it. This can also be a human cognition ability according to Lakusta and Landau (2005) and is a fundamental aspect of our capacity to represent events in our interaction with the world. And of course, if we send information to a receiver who does not understand the symbols, it has no meaning to the receiver (Miller et al., 2001), so this cannot be regarded as information. But, it might be defined as *Data*.

This is one important part of the definition of information and it is applicable to both human and machine. However, there must be something more than just "talk the same language". To visualise the information flow a basic model can be used, see Figure 2.14. This model provides a view of a high level information flow at the final assembly plant.



Figure 2.14: Visualisation of information flow from sender to receiver.

This model will be used in this section as a foundation for the discussions regarding the definition of data, information, knowledge etc. and is influenced by Shannon's (channel) model, see Figure 2.15, of a communication system (Shannon and Weaver, 1963; Losee, 1990) and Hartley's distortion figure (Hartley, 1927).



Figure 2.15: Shannon's channel model. Based on Shannon and Weaver (1963).

As discussed earlier one important part of information is a mutual foundation of understanding, for example language (Millikan, 2001). In the paper "Measuring the Impact of Information: Defining the Concepts." Meadow et al. (1997), state that information contains a message that is understood by the receiver and thereby changes the receiver's knowledge base. A statement is also supported by Brandt and Uden (2003). The keywords in this statement are "understood", "changes" and "knowledge base". In the same paper Meadow et al. present four definitions of information. According to their

definitions there are some things that must be discussed one-by-one and can be related to for example "Data", "Situation Awareness" etc. They are:

- 1. Information: "Messages that exist but are not necessarily sent or received by a given recipient, such as books, unread, in a library, yet deemed significant by someone. In many minds there is no difference between data and this meaning of information."
- 2. Information: "A message sent to a destination or received by a destination, but not evaluated or understood. The distinction between this and the first definition is small. This definition implies that the message is in some way called to a user's attention, but not assimilated by that person."
- 3. Information: "A message understood by the recipient and which changes that person's knowledge base."
- 4. Information: "Information is the process of converting received messages, data, signs, or signals into knowledge."

If one looks at Figure 2.14 and moves in the figure from right to left it is possible to find some similarities and differences compared with the Meadow et al. definition of information. Meadow et al state that information is deemed significant for someone. In Figure 2.14 information is important because it creates a change in the knowledge base that can cause a reaction/action from the receiver. One of the problems in Meadow et al.'s statements is that in part 4, they state that "Information is the process of converting received messages.....", and in part 1 they state that "Messages that exist but are not necessarily sent or received by a given recipient....". So, in part 4, information is a process and in part 1 it can be a physical object that no receiver has yet received. Is this important? Yes, it can be of great importance, at least in this context, in a manufacturing environment. If one looks at Figure 2.14 again it is possible to see that if the receiver has not yet received any information it is not possible to change its knowledge base and therefore it is not likely that there will be any reaction/action. In a worst case, an action will take place, but is this action a right action? This calls for a definition of information that depends on, at least, two perspectives, a sender perspective and a receiver perspective. If one reads part 3 of Meadow et al.'s definition, there is another problem that is important to have in mind, "understood". This implies that to be able to change someone's knowledge base it is important that this "someone" understands the

information observed (Losee, 1997). Understanding is one thing that is part of what Meadow et al. state as; "speak the same language" which can create understanding between sender and receiver. But there must be more than just the language that create and influence "understanding". In part 4 of Meadow et al.'s (1997) definition they write about a process, that Information is a process, for converting data into knowledge.

For a receiver to change its knowledge base some fundamental conditions are important. These are:

- Awareness of a need for information. A receiver must be aware of that he/she should actively search for or receive information (Kelly and Fu, 2007; Yoon 2007).
- Perceive and observe. Perceiving the presence of information is a step towards observation (Losee, 1997).
- Understanding of the information:
 - o Language (Hartley, 1927).
 - In the context (Sawaragi and Murasawa, 2001).
- Interpretation of what action the sender had in mind when it created the information. If the interpretation is wrong, it is likely that the action carried out by the receiver is not the one that the sender/creator predicted. Machlup (1980) and Gustavsson (2000) call this interpretation *a creation of knowledge* in the receiver's mind.
- Update knowledge base. At this stage it can be vital to talk about "receiver believed knowledge" (Machlup, 1980). Receiver knowledge is not necessarily the same as sender knowledge, although the receiver might think it is.
- Action. Act or react according to the information presented (Endsley, 2000b; Sawaragi and Murasawa, 2001). Obviously, at this stage the knowledge that the receiver possesses might lead to a non-reaction.

From the discussion above is it possible to define information as:

Information: A message that when **received**, **read**, and **interpreted** by a recipient **creates** knowledge or **changes** the receiver's knowledge base so that an **action** can be committed by the receiver that is **predicted** by the sender.

2.4.3 Situation Awareness

When looking at Figure 2.2 it is possible to see that there is some kind of receiver interface present. This interface is the connection between the sender and the receiver. Before looking at the specific problem itself it is important to discuss "understanding" and "awareness". If one looks at part 3 of Meadow et al's (1997) definition of information:

Information: "A message understood by the recipient and which changes that person's knowledge base." and the definition in the later part of Section 2.4.2, a rather important part of information "handling", for example receiving, can be identified. The "receiving" of the information is of course an important part of an information flow. According to Figure 2.14, the receiver must interpret the information so that an appropriate action can take place. That is, one has to comprehend the meaning of the information so that an action can take place, and one has to understand how this action will affect the future status of the product.

Endsley (1996) has formalised/defined this as Situation Awareness (SA): "the perception of the elements in the environment within a volume of time and space, the comprehension of their meaning and the projection of their status in the near future".

Before it is possible to receive information, one must perceive it (Rookes and Willson, 2000; Sheridan, 2002; Feng et al., 2009). Endsley, in the paper "*Automation and Situation Awareness*" (Endsley, 1996) stresses a very important issue regarding information and information receivers. There are problems connected to production systems as well as a pilot's work environment, i.e. the cockpit, and this is active and passive attention. This together with active and passive information seeking, passive and active reception are believed to be the main contributors to the problems in the specific work environment (Machlup, 1980; Wilson, 2000; Bäckstrand et al., 2005a).

In short (a more detailed description of attention is presented in Section 2.4.6), active and passive attention can be explained as:

- Active attention: Is connected to actively processing information.
- Passive attention: For example a process operator whose function is to observe an automated process.

According to Endsley (1999) passive attention is a state that can affect the ability "to correctly integrate or comprehend information". One interesting part in her discussion is that in a study regarding an automobile navigation task she found that perceiving information/data correctly was affected negatively if a receiver was in a "passive attention" state. The receiver was still aware of presented data, still capable of perceiving data, but was less effective when it came to comprehending the meaning of data and relating that to the goal of a task or operation.

The next step in SA, if we assume that the receiver has perceived presented information, is to integrate the information in the context or as Endsley (1996) states: "....the comprehension of their meaning and the projection of their status in the near future".

In Figure 2.16, SA Error Taxonomy, Endsley (2000b) describes problems related to the processing of information by a receiver. If one has perceived the information, the next step is to process the information/data, and thereby change our knowledge base. In level 2 of the taxonomy Endsley states that it is important that a receiver must be able to derive relevant meaning from the data perceived in level 1. There are a number of different reasons why a receiver misinterprets information/data or has problems with finding the most relevant parts among all the data.



Figure 2.16: Situation Awareness model (Endsley, 2000b).

One reason, according to Endsley, can be a lack of a good receiver Mental Model (Endsley, 2000a). Moray (1998) writes "*A mental model is a mapping of the properties of the task to its representation in the mind of the worker*". According to Besnard and Greathead (2003) an operator maintains a mental representation of different processes in a system. These incomplete mental representations of the reality, due to the limitations of a human memory, need to be fed and updated by knowledge and available data in the work environment. Other issues concerning mental models include Team Mental Models where Langan-Fox et al. (2001) hypothesise that if a team share a similar mental model, they should be more successful. Using this hypothesis together with Figure 2.16 Endsley's suggestion regarding mental models could be extended by talking about Team Mental Models, and state that it is possible that sender and receiver in the assembly environment should be seen as a team, and thereby they should/must also share the same team mental model/s.

To summarise Situation Awareness, and using Endsley's definition it is possible to say that SA deals with:

- Reception of data/information.
- Passive and active attention.
- Mental Models, team and personal.
- Project future state of a system.

2.4.4 Information need and demand

"Structured translation of data into action" (Johnstone and Tate, 2004) to reach a specific goal must be the main focus for the assembly personnel (Wilson, 2006b). This requires that the information is available at the right time in the right place and that the assembly personnel have identified a need for the specific information (Endsley, 2001; Wang and Forgionne, 2006).

In an evaluation of some of the workstations at the specific plant (Bäckstrand et al., 2005a) there were indications that there are problems connected to the distribution of information/data that can be related to what Bäckstrand et al. choose to define as "Delivery Vs. Demand" of information/data.

In a discussion regarding Information Overload, Wilson (2001) refers to a journal paper written by Cacioppo and Petty, "The Need for Cognition" (Cacioppo and Petty,

1982). In this discussion Wilson describes a "computer jargon" that is rather similar to "Delivery vs. Demand" and that is Information Push and Pull. Information Push and Pull or dissemination and acquiring (Rich, 1997) is according to Wilson one factor in the discussions surrounding Information Overload, but it is also a key factor regarding the information flow in the specific assembly plant. In the following discussions the terminology "Delivery" and "Demand" will be used.

Information/data delivery, is according to this author, the event that occurs when a specific type of data/information must be accessible in a specific work environment, for example an assembly workstation, and a *Demand* for information/data occurs when an "object", in this case one of the assembly personnel, has identified a need for information/data. This need originates from a need to fulfil a goal (Figure 2.17), and maybe the satisfaction of fulfilling the goal (Wilson, 1981; Losee, 1990).



Figure 2.17: Demands and goals (Bäckstrand et al., 2006a).

One of the conclusions, from a evaluation performed by Bäckstrand et al. (2005a), was that events in the work environment started the Information Delivery process, and not an

identified information need from the personnel. These events (some of them) occurred every seventh minute, and this leads to an Information Delivery flow that was/is unique for that specific event. This event by itself would be expected to create an information need, but there are some indications that this is not the case. As discussed earlier in the text a precursor to demand could (must?) be need. Wilson (2006a) states that "need is a subjective experience that occurs only in the mind of a person in need". According to one of many dictionaries on the internet, need is a target-oriented behaviour that is connected to for example appreciation and security (Nationalencyklopedin, 2005). This raises the question: to be able to create a demand originated from a need a goal must exist. How is this goal formulated and from whom is it originated? Should it be a personal goal? Alternatively, should it be a goal defined by the organisation? If it is a personal goal, how does this goal affect the information demands (Losee, 1990)?

Bäckstrand et al. (2005b) describe in a preparatory work why there is a need for cost calculations in the specific area. This need has created a demand for more data/information. Is this connected to reality for the assembly personnel? According to this author it is, because it builds on the same foundation, or at least it should. The demands for information should follow the same process, despite the difference in goal.

2.4.5 Information Overload

If one assumes that the flow from Sender to Receiver exists, one can also assume that like in a river, the flow of Information can overflow in a specific situation, i.e. the amount of information a receiver has to process is so large or takes so much time to process that it is not possible to process it in time (Himma, 2007). Wilson (2001) defines information overload on a personal level as:

" a perception on the part of the individual (or observers of that person) that the flows of information associated with work tasks are greater than can be managed effectively, and a perception that overload in this sense creates a degree of stress for which his or her coping strategies are ineffective." (Wilson, 2001)

Information Overload can, according to some researchers, occur when a receiver of information receives so much information that it is not possible to manage the information in an effective way (Losee, 1990; Brewster, 1997; Sheridan, 2000; Himma, 2007). This can create a feeling of lack of control and thereby stress (Wickens, 1992;

Bellorini and Decortis, 1995; Edmunds and Morris, 2000). Wickens (1992) as well as Mital et al. (2004) state that there are connections between stress and errors. It seems that stressful situations can create absentmindedness as well as inappropriate coping strategies (Norman, 1990). An important effect of stress can be what Wickens and Hollands (2000) call *"increasing selectivity*". This selectivity can according to Wickens and Hollands create a perceptual tunnel effect that makes it harder for a human to observe information on the periphery, and thereby makes it more difficult for the user to handle information outside that perceptual tunnel. This filter effect seems not to be an effect entirely of a reduction in spatial visual envelope. Subjective importance and/or priorities seems to be an important part of the filtering.

According to Meadow et al. (Meadow and Weijing, 1997) a message has to have *"been received and understood or appraised"* to be referred to as information, which *"means that we can have data overload, but not information overload."* They also state that :*"if a message must be received in order to constitute information, there cannot be an overload of unreceived data."* This creates a minor problem in the specific case, the assembly plant, or rather with the hypothesis that tries to explain the problem. According to Meadow and Weijing it is not possible that a person can be overloaded if he or she has not received any data. Still, the assembly personnel have a problem with the handing of data and information, and thereby (sometimes) act in an incorrect way. It is possible that this problem, handling of data/information, has nothing to do with Data/Information Overload, but until a further investigation proves otherwise, overload will be regarded as one of several parameters affecting the handling of data and information.

2.4.6 Attention

Blasch and Plano (2002) present key theories of attention, Four of them are (briefly):

- Early Selection Theory (or the Filter Theory): According to this theory we only take up a small amount of information and this implies that we use limitations and selectivity as a filter. These two considerations are according to Duncan (1995) the core of the filter theory.
- Attenuation Theory: All messages are weighted and priority is given to one that will be received in full and the other messages receive only partial processing (Blasch and Plano, 2002; Hoshino, 2003).

- Late Selection Theory: All messages attended or not, undergo semantic analysis and downstream analysis. The messages are weighted and the key message is determined due to the limitation that makes it possible to respond to only one message at a time (Hoffman, 1998; Pashler and Johnston, 1998; Blasch and Plano, 2002).
- Object Based Theory: Hoshino (2003) states that according to this theory it is not possible to simultaneously perceive more than one object at a time. The visual scene is dissected into distinct objects.

Within this research no effort will be made regarding the theories or the likelihood of their substantial value. The purpose of presenting some of the theories connected to attention is to show that there are different beliefs regarding how humans handle attention and how attention can affect the information search process.

Earlier in this thesis it was stated that "to reach a specific goal must be the main focus for the assembly personnel". According to Yantis (1998) as well as Blasch and Plano (Yantis, 1998; Blasch and Plano, 2002) there are two ways to control visual attention: top-down (goal driven) or bottom-up (stimulus driven). Attention is said to be goal-driven and stimulus-driven. Yantis (1998) exemplifies the behaviour difference between the two by presenting how a person looks for a particular product in a supermarket:

- Goal-driven (active attention): The product (in Yantis's example, the product is a cereal) is known to have a specific marker, or a packaging, and one searches for that. This product is likely to be selected by our attention and recognized.
- Stimulus-driven: All the products in a part of an aisle tend to have the same or similar packaging, except one, that seems to "pop-out of the background and draw attention automatically".

There is more than one issue within the attention field that is of interest to this research. One issue is how cues in our environment affect our attention. Richard et al. (2003) compare goal-driven and stimulus-driven attention and how they are affected by cues and how cues can control our attention regarding both time and space (Table 2.1).

Stimulus-driven Attention Control	Goal-driven Attention Control			
Associated with direct cues (a.k.a.	Associated with symbolic cues (a.k.a.			
peripheral or exogenous cues) that occur	central or endogenous cues) that indicate a			
directly at a potential target location	potential target location			
Initiated by sensory event, such as the	Initiated by cognitive operations, usually			
abrupt appearance of a cue or other	following the interpretation of a symbolic			
transients	cue's meaning			
Effects are rapid and transient, peaking	Effects are more gradual and sustained,			
approximately 100 msec after a cue's	peaking approximately 300 msec after the			
appearance	cue's appearance			
Are involuntary because they are	Are voluntary because they are			
minimally influenced by:	diminished:			
1. instructions to ignore the cues	1. by instructions to ignore the cues or			
2. the predictability (validity) of the cues	2. if the cue is not useful to the subject			
	(for example, low cue predictability)			

Cues are of interest for more than one reason:

- Their effect on the way that they can help us direct our attention to a location in space where our attention is needed.
- Their ability to help us, during some circumstances, to shift our attention to a place faster and more accurately than uncued locations. That is, if a cereal box in the example has a marker, i.e. a cue, that stands out from the other boxes, it is more likely to be chosen (LaBerge, 1995; Proctor and Dutta, 1995; Yantis, 1998; Richard et al., 2003).
- Their ability to reduce the numbers of stimuli to one (Vierck and Miller, 2005).

Active (Goal-driven) and Passive (Stimulus-driven) Attention

In a previous part of this section, a short definition of Goal-driven and Stimulus-driven attention was presented. There is some difference though between Goal-driven and Active attention and Stimulus-driven and Passive attention. The difference, at least compared with how Yantis (1998) presents it, between goal-driven and active attention is minor, but the difference between Stimulus-driven and Passive attention is of more interest.

Active and passive attention are two ways to classify attention. Active attention is a state that is required of us when we need to detect information in the surroundings for our own purpose. Passive attention is the ability to identify dangerous situations in our surroundings (Kida et al., 2006). The ability to actively detect information in our environment is of great importance in an assembly context. The interesting aspect is whether or not there is a connection between attention and internal rejects. Some of the research regarding active and passive attention has been conducted within neurophysiology, for example about how a somatosensory stimuli⁶ influenced active and passive attention (Kida et al., 2006). In this example the researchers recorded Electroencephalography (EEG) results during the study. The purpose was to investigate if a change in the somatosensories stimulus frequency during active and passive attention tasks affected the EEG. The most interesting part of this research is not the results, higher amplitude (for example a higher brain activity) during active attention tasks, it is parts of the method used in the study. During the study an electrical impulse was given to the subjects left hand fingers via attached ring electrodes. Depending on the task, active or passive (defined by the researchers) the effect on the EEG's amplitude differed. The interesting part is that the electrical impulse triggered an effect on the EEG. This indicates that:

- A trigger, in this case tactile, creates a reaction in our brains.
- Depending on what stage one is in, active or passive, the reactions shown on the EEG are different, higher amplitude for active and lower for passive.

This knowledge can be used in this research in the sense that it is not necessary to study the reaction in the brain while performing active or passive attention tasks. It is also possible to in some sense define active and passive tasks:

⁶ Somatosensory stimulus, in this case was an electronic stimuli delivered to the left hand.

- Active: For example counting or button pressing.
- Passive: For example reading a book.

Kaplan et al. (2006) define attention as " the process whereby an agent concentrates on some features of the environment to the (relative) exclusion of others." According to Kaplan et al, this process can occur in two situations: when a salient event, such as a loud noise, triggers the attention, or when one is involved in an intentional directed process, such as climbing a mountain, where one actively selects particular features in the context.

An assembly task according to this definition is an active attention task. This task is a part of a longer assembly sequence containing a number of tasks or activities. The sequence in the research context can typically be divided as showed in Table 2.2. This example is a part of a method instruction for a specific software, AVIX (Solme, 2008), used by the reference company. The definitions in this example are the same as used by the company.

Table 2 2. Typical	l tasks/activities in ai	n assembly seauence	(Solme 2008)
Tuble 2.2. Typical	iusns/activities in ar	i assembly sequence	(Sounc, 2000).

Activity	Explanation	Comments
Take	When you take things	Take part from e.g. rack
Place	When you prepare things	Place part on product for
		later mount
Handle	When you prepare things	Sort/prepare before
		assembly. E.g. handle
		pipes before assembly
Affix	When tightening, glue,	Tighten screw
	clipping	
Adjust	When something is	Only mount, not
	corrected	tightening
Assemble	mble When you assemble in	
	right position	tightening
Inspect	When you check an	
	assembly	
Return	When you return things	Return machine
Instruction		Can be first task for each
		position
Administration	When you document	Write/stamp info
	information	
Wait		When you have to wait
		for something before you
		can assembly. Must be
		between task not as end
		station
Stop Watch	Used when analyse is	Machine time, etc.
	impossible	
Stop Watch Movements	Used when analyse of	When use lifting tool
	movements is impossible	

These activities can be divided into sub activities such as "Walk", "Pick", "Bend" etc. (all the activities can and should be seen as skills, something that we possess).

Some of these activities will be performed automatically, such as walking, reaching, etc. and sometimes information processing is automated (Järvelin and Wilson, 2003). This automatism is strongly connected to active and passive attention. LaBerge and Samuels (1974) state that the "criteria for deciding when a skill or subskill is automatic is that one can complete its processing while attention is directed elsewhere". This ability, to do something automatically, that the human possesses is interesting and might be a necessity due to the limitations that exist regarding how we use the different attention modes. According to Reason (1990) attention can be seen as work performed on discrete elements in a restricted work space, our working memory. The work uses energy from a

strict limited "pool of attention resources" (Reason, 1990; Wickens, 1992; Downing, 2000; Endsley, 2001; Crawford and Cacioppo, 2002; Niebel and Freivalds, 2003), therefore is it important to use this energy in an optimal way.

If one presumes that passive attention draws less energy from the "pool of attention", one of the goals with information must be to support the personnel in a way that makes it possible to continue working in a passive attention mode, where the use of skills and subskills is automatic. Norman (1990) describes this kind of automation as a useful advance that can replace tedious or unnecessary tasks and monitoring. This must (if one presumes that the Reason and Norman theories are correct) result in less use of resources. Less use of resources by one process, for example the Information Search process, should make it possible to use the limited resources more effectively (Downing, 2000).

When one is in an active (goal-driven) or passive (stimulus-driven) attention mode, attention is summoned exogenously⁷ (stimulus-driven) or deployed endogenously⁸ (Rafel and Henik, 1994; Coull et al., 2000). Rafel et al. present four properties distinguished by Jonides (1981):

- 1. Endogenous orienting of attention is vulnerable to the effect of following memory load, whereas exogenous in not affected by such cognitive demands.
- Endogenous orienting can be suppressed voluntarily, whereas exogenously triggered orienting cannot. This is statement is supported by Richard et al. (2003), see Table 2.1.
- 3. Exogenous orienting is not dependent on the likelihood of a peripheral signal, whereas endogenous cueing is strongly influenced by the subject's expectations.
- 4. Exogenous signals appear to produce stronger orienting effects (as measured by the size of the validity effects of the precue).

Later in this thesis a case study will be presented that in some senses uses these properties (Section 3.2). The study itself will be presented from a quality impact point-of-view later, but it might be useful to present the connection between attention and that study at this stage.

During that study, an alternative user interface was used. The existing interface (Figure 2.18) uses text as an information provider. It shows to the personnel what part they should assemble and in what sequence they should assemble it.

⁷ Exogenous: from outside, in this case outside the body.

⁸ Endogenous: from inside, in this case the brain.

Assembly assurance											
	Order id: 99258542				VOLVO IT, SKÖVDE						
Head	lor		No.	Instruct	on text	Bcode	Contirm	Tight.	3u Tìgh [≜]		
364025			1			\checkmark		\checkmark	хЗ		
946671 SCREW M8*	30		3								
20411113 CONSOLE			1					0	x 2		
1547252 SPACER			1								
1546531 LOCKING S	ни		1								
977977 CLIP			1								
946440 SCREW M8*	16		2								
STATION COMPLETED)						\bigcirc				
									,		
STATION COMPLETED :											
(F2) Status CTRL (F3) Send Res.	(F4) Man.Input	(F5) Re Do	(F6) Man.C	oer. (F7) Reject	(F8) Msgs	(F9) OK	(F9) OK (F10) Station list				
16:28:45:Batch: FType: PgmSetBatch Id: 1 Name: Result: Ok					Ven	Version: 1.5.6 16:28					

Figure 2.18: Example of existing user interface.

The alternative interface used during the case study did not use text as an information provider but instead it used colours. The meaning, the information provided by the different interface, was the same with the difference that the interface using colours did not provide information regarding smaller and common parts, for example screws, it just provided information regarding the different main part, in Figure 2.18 it is the "CONSOLE" that is the main part.

The main goal is unaffected by the different interfaces, i.e. to "assemble parts on engine". The difference between the two interfaces is how they affect the personnel before, during and after the assembly task and thereby the attention mode and the use of our working memory and the "pool of attention". So how are the two associated with the different attention modes? Well, this is a little bit tricky due to the small differences that exist regarding how to define active/goal-driven/ endogenous attention and passive/stimulus-driven/exogenous attention. It is however important to understand how different interfaces are connected to different attention levels.

In this case it might be proper to look at the examples stated by Yantis (1998) presented earlier in the text. His example presents an "everyday" situation that is familiar to anyone that has been to a supermarket. He connects active attention to a goal-driven search for a specific known marker, and passive attention to a stimulus response to a "pop-out" effect produced by a difference in a marker (in the example, the marker is

product packaging that makes the product differ from the others on the shelf). According to Jonides (1981), one of the differences between active and passive attention is how they create a cognitive demand, in this case a difference in memory load. Wickens (1992) states that a failure to carry out an action can be connected to a failure of memory. This shows that a connection exists between attention, memory and the ability to perform a proper action. This connection is vital when it comes to the preparation of Case study 2 presented in Section 3.4.

2.4.7 Human Errors

In 2005 a new assembly line was built at the plant. There were technicians as well as assembly personnel working within the project. When the author worked with the preparations for the paper "Parameters affecting..." (Bäckstrand et al., 2005a) there were some discussions regarding Mental Models and if a non-uniform Mental Model could negatively affect a work place design. This discussion led to an interview session with the engineers and assembly personnel within the project. The conclusion from this session will not be discussed in detail in this section, but some things are worth mentioning.

Two of the questions in the interview were (translated from Swedish):

- 1. "Is it possible to use shortcuts in an assembly task and thereby lower the task time?"
- 2. "In what way, if any, can shortcuts influence quality?"

The answers to the first question indicated that it was possible to use shortcuts. A shortcut in this case is for example an unscheduled or unforeseen (by the production preparation engineers) preparation "act" to prepare for a scheduled assembly task. All of the interview subjects indicated that these "acts", taking a shortcut, seldom lowered the total assembly time, and as an answer to question number two, as a rule, the shortcuts can influence quality negatively. The reason that it can influence the quality negatively is that the assembly process is defined from a quality, a productivity and a physical ergonomics perspective. If the personnel do not follow the described process, the quality risk increases. This can be seen as an example of Intentional Error, but it can also be a misinterpretation of work instructions or data and thereby it can be an Unintentional Error.

According to Pennathur and Mital (2003) there are, in general, two types of human errors:

- Intentional: Intentional errors are caused by actions deliberately conducted with a belief that they are correct or better, i.e. faster, safer, etc., than the prescribed action.
- Unintentional: Unintentional errors are actions performed with no prior thought, for example assembling the wrong parts, misinterpretation or forgetting to read an instruction, etc.

Wickens (2000) proposes four categories (Figure 2.19) and visualises where these fit in the information processing process of human errors. They are:

- Mistakes.
- Slips.
- Lapses.
- Mode Errors.



Figure 2.19: "Information Processing Context for Representation Human Errors" (Wickens and Hollands, 2000).

Mistakes: A mistake can be seen as a "failure to formulate the right intentions". Knowledge-Based mistakes are connected to decision-making and are caused by, according to both Endsley (2000b) and Wickens (2000), decisions based on an unsuccessful interpretation of a situation. There are connections between Situation Awareness and Knowledge-Based mistakes, although Wickens (2000) describes it as "a failure to understand a situation (i.e. incorrect knowledge)".

Rule-Based mistakes occur when for example the assembly personnel believe that they know the situation and the environment where he or she operates and plan an action based on this knowledge, but overlook signs in the environment that can indicate other types of actions (Rasmussen et al., 1994).

Wickens (2000) describes rule-based actions with an "If-Then" notation where the "If"-condition controls the actions and a "Rule-Based mistake" occurs when this condition does match the environment conditions. At the specific plant the "environment conditions" change "scheduled-chaotically" due to variations in production, and to changes/updates on the product. At this point it is not possible to draw any conclusions regarding what types of errors occur. This is believed to be important; identification of what types of mistakes that have occurred in the past must be an input to the solution to the problem. If this identification has not been done, there is a risk that a solution focuses on solving a Knowledge-Based problem when it actually is a Rule-Based mistake/problem.

According to Wickens, both types are characterized by an intention that is not appropriate for the situation (again, there are strong connections to Situation Awareness). But there are important differences that can call for different actions when it comes to finding a solution. Wickens states that:

"Rule-based mistakes will be performed with confidence, whereas in a situation in which rules do not apply, and where knowledge-based mistakes are more likely, the operator will be less certain."

Slips: Slips are characterized by an action where the intentions are correct, but the action is carried out incorrectly. Wickens (2000) uses a naïve metaphor that explains in a good way the "essence" of slips. Wickens writes:

"Pouring orange juice rather than syrup on the waffles while reading the newspaper" and states that this is a perfect example of a slip. This example is very

interesting because when Wickens (2000) describes mistakes and Knowledge-based mistakes he mentions that attention is one of many shortcomings connected to knowledge-base mistakes. The example above describes a typical situation of shared or divided attention (Wickens, 1992).

Lapses: *"Failure to carry out any action at all"*. Wickens (2000) divides Slips and Lapses into two related but different categories, while Reason (1990) discusses them as one category. The main difference between slips and lapses might be that, if one looks at the orange juice example, slips can be caused by a lack of attention whereas lapses involve a failure of memory.

According to Reason, mistakes are a mismatch between the intention of an action and the consequences caused by the action, and slips and lapses are a discrepancy between the intended action and the action that actually takes place.

Mode Errors: "*Pressing the accelerator of a car to start at an intersection when the transmission is in the reverse mode*" (Wickens, 1992). Mode errors can occur when for example the assembly personnel commit an action that can be used in both correct and incorrect circumstances. According to Wickens, they are related to slips and lapses; Slips due to the right intention incorrect performed and lapses due to memory failure.

At this point it is assumed that a connection between data/information flow and internal rejects exists, but at the time, it is not possible to find the error source. Wickens (1992), Wickens and Holland (2000), Reason (1990), Pennathur et al. (2003) all discuss different types of errors, for example lapses, mistakes etc. The cause of the errors is different, and therefore one can assume that the solution for prevention to a specific error is different. This can call for not only finding the error source, but it might also be of importance to categorise the (assumed) different error types.

2.4.8 Literature study discussion

The literature study has identified a gap between theories and practical or proven use on a factory level. This does not have as much to do with the specific literature, as it is connected with the specific scientific fields. The findings when reading the literature are interesting and support the hypotheses regarding the information search behaviour, especially within psychology and attention. Attention, active and passive, seems to have a strong connection to memory and stress.

The attention theories: In this thesis, four different attention theories are presented: Early Selection Theory, Attenuation Theory, Late Selection Theory and Object Based Theory. The four theories all differ from each other in at least one important aspect and that is in how they handle the information in the work environment. Depending on which of the four one believes is the one that best describes how humans handle information, the impact on the design of an information system, or the result (the existing information system) differs. For example, according to the Early Selection Theory, see for example Duncan (1995), humans only take up a small amount of information, i.e. humans use limitations and filters at an early stage of perceiving information and thereby minimize the amount of information that needs to be processed. A possible conclusion drawn from the study of the Early Selection Theory is that the amount of information presented in the context will not interfere with the information search process, and thereby the amount of information will not influence the success of such a process. In Section 3.3, Case study 2 is presented, and the results from that study indicate that the Early Selection Theory might not be applicable within the context of the thesis. The Late Selection Theory on the other hand states that all messages attended to or not, are weighted and the key message determined due to limitations that makes it possible to only respond to one message at a time (see for example Hoffman (1998)). This indicates that all information in the work context studied are attended to and processed in some sense in our brains. The conclusion from this is that more information in our work place creates a need for more limiting and filtering by our brains. This raises a rather interesting question of thought. If humans use filters, late or early in the process, how are these filters created? Who decides how one should filter the incoming information? Is it the initial training and practicing that influence this filtering? It is not within the scope of this thesis to answer these questions, but the theories and their substances influence the experimental design of the studies presented in Chapter 3, and one must assume that they also influence the outcome of an information system design; the problem is to understand how. This is an important discussion but it has to be continued at a later stage, in this case, in future work connected to this research.

Definition of Active and Passive attention: In the literature there are at least three different terms for attention: (1) *active and passive attention*, (2) *goal-driven and stimulus-driven attention* and (3) *endogenous oriented and exogenous oriented attention*. They are rather similar except for the fact that there are different views regarding the
definition of them. For example, Kida et al. (2006) exemplifies passive attention with reading a book, while Kaplan et al. (2006) state that an intentional directed process, such as mountain climbing, is an active attention activity. This is rather confusing, since reading a book must be considered as an "intentional directed process", although not as dangerous as mountain climbing. Despite some of the differences in the definition of attention it is obvious that attention could be the root cause for the specific type of internal rejects that are under investigation within this research, i.e. wrong parts assembled or parts missing. As mentioned, one problem with the existing literature is the lack of literature that is directly associated with assembly. Much of the literature focuses on the process industry, where the main focus for the human is to control and observe, for example paper mills, nuclear power plants and cockpits. This is not that strange as these environments are critical and the outcome of a mistake can be catastrophic. The outcome of an error in an assembly environment such as the one studied in this research, might not cause a disaster, but may have large economic consequences. There is a major difference in the tasks though, if one compares the work in a control room with the work on an assembly line. The main difference is that in a control room the main task is usually surveillance, a passive task, while in an assembly environment the main task is the active task of assembly. This difference puts different demands on the information systems, and on the humans using the systems. One difference is the amount of information the person has to handle, but the most important differences are in the tasks. If one focuses on the need of information for the assembler, and not on the comparison of work environment, one realises that there are some differences that the discussion regarding active and passive attention can explain; active attention is triggered by an exogenous event that cannot be suppressed or ignored by the assembly personnel (or by anybody else for that matter). This event is the trigger that should start an Information Search Process, and nothing else. Therefore, an information system that triggers the information search process is essential and when the information search process has started it should support a passive attention mode. This means that the need for an active (active from an attention perspective) use of the information must be kept to a minimum.

Situation Awareness (SA): Much of the work within the SA field has so far focused on military, aviation and process industries, for example nuclear and chemical plants. The emphasis is on process monitoring, supervisory control and not on active task performance. A rather interesting conclusion can be made when one studies SA, and that

is the absence of the important issue regarding triggers and their role regarding how triggers create change of state, from passive to active attention and from active to passive attention. For example in Endsley's three level SA model (Endsley, 1999), triggers should be a predecessor to the model, or a part of level 1. Endsley (1999) describes "Critical cues related to key features" as an important part of a system design. Endsley also states that due to several factors, although the information is directly available it is not observed or included in the scan pattern, due to for example not looking at the information, attention narrowing etc. This is according to Endsley the main contributor within SA level 1 to SA related errors. This is where this author believes triggers can be a major contributor to preventing this type of human error. The trigger in this case is the event that creates awareness of the presence of important data or information and draws the personnel's attention to the correct data or information source. A remark is that the trigger can be different, an example is within the engineering field where 51% of the respondents answered that solving a problem triggered their need for information (Kwasitsu, 2003).

Endsley is the main contributor to the SA science field; still there are some important issues that Endsley does not discuss including:

- 1. How can triggers contribute to error prevention connected to SA level 1 problems?
- 2. How can a good SA be maintained over a time period, with a minimum of work effort, i.e. how can a proper attention level be maintained so that a good SA exists and continues to exist over time.

Human Error and Stress: The literature review has briefly included human errors and stress. However, it is neither possible nor interesting to exclude human errors or stress as error sources that have an influence on the internal reject rate. A reflection is that it seems that automatism perceptual tunnel affect, increased selectivity, stress and filtering, all have something in common and that is that they all can negatively affect the ability to observe information. However, Wickens and Hollands (2000) write something that is extremely interesting. According to them, a study conducted by Houston (1969) showed that the stress-produced perceptual tunnel influence, although this influence usually affect the task performance negatively, in some circumstances it facilitates performance when focused attention is desired. This gives rise to the question if it is possible to use this filter effect in some way and is it possible that, depending on the stress levels

experienced by the assembly personnel, this can dynamically affect their performance? It is clear that a (too) high stress level can affect the performance negatively, but is it also possible that a (too) low workload can influence the performance in a negative way? Sutcliffe (1995) states that non-stimulating tasks should be avoided. According to Sutcliffe the reason for this is that it can be hard to enforce attention on an uninteresting task. This statement is connected to a discussion by Sutcliffe, where he addresses stress and fatigue, where stress can cause fatigue. The discussion is interesting for many reasons, but especially where he states that:

- Fatigue may result from continuous mental activity in overlong, mundane tasks.
- Fatigue may result from intensive concentration on tasks demanding difficult mental reasoning.

However, according to Sutcliffe it is not the case that task complexity always leads to fatigue. The number of stressors that can affect the outcome of an information search process are many, for example noise, vibrations, heat and psychological factors such as anxiety, fatigue, frustration and anger (Wickens and Hollands, 2000). According to Wickens and Hollands they all affect information processing, see Figure 2.20.



Figure 2.20: A representation of stress effects according to Wickens and Hollands (2000).

These factors, such as stress, fatigue, the ability to handle information and their affect on the information search process, are extremely interesting. However, the focus for the further work within this research will be on attention as this is believed to be the main issue regarding the start of a successful Information Search Process. That is, if the assembly personnel are unaware of changes in the information flow, the likelihood of an upcoming quality problem increases.

The influencing factor from stress is of great interest, but in order to understand how it influences quality separate studies should be performed and such studies are not possible to conduct within this research.

Attention is an extremely interesting area from a manual assembly perspective. A problem, or rather an opportunity, that the industry is facing is how to get as much as possible out of the information and the information systems. Much of the literature regarding attention presents findings that have their origins in laboratory experiments within the field of psychology. One of the problems with this is that it is rather hard to state "the problems we are facing in the assembly plant have their origins in the lack of attention support". To better understand how attention affects assembly systems, studies should be performed within or in close contact with these kind of systems. These studies should better support understanding regarding the influence that attention might have on the assembly process.

The conclusion is that attention (passive, active etc.) is an important component that influences quality output; it is obvious that if an assembly worker is not aware of a need for information a correct information search process will not be started. This will result in a quality risk that can influence the internal rejects severely.

In the literature addressing cognitive aspects, much effort is used to describe how humans use and process information and very little on how to use cognitive science to create effective work environments from an information demand point-of-view. There are several cognitive related evaluation methods, for example "Cognitive Work Analysis" CWA (Rasmussen et al., 1990; Vicente, 1999; Bäckstrand et al., 2005a) which can be used to evaluate work environments. One problem with CWA is that it is very time consuming and thereby, from a cost-benefit or a safety perspective, only useable or applicable in evaluation of complex information systems, for example helicopter cockpits, process control interfaces etc. In the literature survey it was recognised that one major component was rarely covered and that is how important information awareness is to successful goal fulfilment. The discussions regarding information, data, interface design etc. are often focused on how to design an interface so that a user can quickly can find and use information. The real problem is how to create awareness of a need for

68

information, for example active attention with the purpose of achieving a specific defined goal. Endsley (2001) presents an interesting guideline, "SA-oriented Design" that can be useful for the evaluation of work environments. Unfortunately it seems that this guideline is also lacking the major component of how to handle active and passive attention, but nevertheless it is interesting enough to evaluate and develop in the future.

2.5 DISCUSSION OF THE ANALYSIS

In this chapter a presentation has been made regarding the industrial related research environment and literature closely connected to human-information interaction, for example attention, situation awareness, human errors etc. The main purpose with the analysis work was to try to find information that could explain the problems that occur in the assembly environment, and if this was not possible, try to understand how information and the use of information can influence humans in both negative and positive ways. In addition, an intention was to try to make assumptions that can support the hypothesis and the research objectives.

The conclusion from the analysis is that there are two main areas that are affected in the assembly plant: productivity and quality. The analysis revealed that it is, at this stage in time, not possible to measure the impact that the information flow has on the productivity in the assembly plant. This is not entirely true since it depends on how one measures productivity. On a general level productivity is the ratio output divided by input. Output could for example in this case have been selected as quality output, and input as hours worked, or output could have been selected as the value of the assembled product and input as the specific product cost. So, if one can measure the quality, one can in at least one way measure the productivity. In this case, the quality is more interesting than the productivity, so therefore the possible impact from the information flow can be evaluated against the quality output. Productivity, in the form of engines produced per worker, and how the information flow affects that productivity, would have been very interesting to measure, but in this case is it not realistic to identify this correlation due to too many parameters, for example personnel changes, order intake etc., that can affect the productivity output.

So, can the quality output be affected by the presence (at this point, one must assume that the information provided is of a quality that makes it possible to use, i.e. we assume that insufficient information can affect the quality in a negative way, but this is considered not being the case) and the use of information?

In chapter one, it was stated that the common opinion among management was that the personnel did not use the information provided in an appropriate way and thereby created some of the quality problems that occur. It was not possible to find explicit evidence in the literature that pointed to lack of information as a negative parameter regarding quality. However, an absence of information can create a quality risk. The interesting part is not the information itself, instead it is the combination of in what form, for example text, colour patches, symbols etc, the information is presented and how one interacts with the information provider and how this interaction affect humans.

At this stage in the thesis it is important to try to understand how the different theories presented in the literature can be connected to specific problems at the assembly plant if these problems can cause quality problems and if the theories can be applied in the specific research environment.

Earlier in this chapter Figure 2.14 was introduced as a help to visualise the complexity of the information flow. This figure can be used as a support when one tries to understand the effect the different theories can have on an information flow. The final goal of a search process must be to fulfil an action. To produce that goal, a number of steps must be taken, such as, from right to left: update of knowledge base leading to a decision and to an action, interpretation of information lead to an update of the knowledge base, transformation of data leading to information.

To be able to define information four different parts must, according to Meadow et al. (1997), be fulfilled:

- Existing message.
- A message sent received by a destination.
- A message understood by a receiver, which creates change in that person's knowledge base.
- Information should "process" data into knowledge.

The first and second criteria seem reasonable; to be able to use the information, the message including the information must exist and it must be sent to a receiver. One could assume that in an assembly environment, the message containing the information is always sent and should therefore be available to the personnel when needed. The system

that has sent the message has identified a need for information, should a demand for information occur at the same time.

In the paper *Attention, Interpreting, Decision-Making and Acting in Manual Assembly*, Bäckstrand et al. (2006a) four situations are presented that can occur and that are connected to the above criteria. The purpose of any rational action should be to achieve a goal (Wharton and Lewis, 1994; Wilson, 1999; Johnstone and Tate, 2004). This should normally create a demand and a need for information. In principle, four situations can occur regarding information need versus demand:

- 1. There is a need but no demand. In this situation, an error will sooner or later occur.
- 2. There is a need and a demand. This situation is the preferred one. In this situation, the error risk due to a lack of information is lowest. However, this situation still requires that the information available matches the need. Furthermore, if there is a mismatch between need and demand, then a residual state (1) or state (3) will eventually emerge.
- 3. There is a demand but no need. This situation can be frustrating for the personnel. They have identified a need and have a demand, but the context of for example the assembly station does not provide them with the (subjectively) needed information. This is a potential future error source; frustration can result in the absence of active information seeking behaviour in the future, even when an information need is identified.
- 4. There is no need and no demand. This situation is more or less trivial.

Figure 2.2 presented earlier in this chapter visualised an information seeking process. In that process, a decision has to take place. This decision is important for the outcome of the information seeking process and is likely to affect the quality output. The decision is triggered (or at least should be) by a need from the system that an information search process should be started, but it can also be triggered by a demand from the human. If a message is sent to a receiver, and this receiver fails to identify the information need, the risk of a quality problem increases.

In Meadow's (1997) definition of information, the remaining two parts that must be fulfilled are: (1) message understood by a receiver so that a change can be made in the knowledge base, and (2) the information should make the receiver "process" the data into knowledge and from that make decisions. That is, the decisions are driven by the

information as presented (Miller et al., 2001). A note has to be made regarding transmitted and received information. According to Losee (1997), a situation where there is a message deemed as having no value to a receiver should be seen as no information having been received nor transmitted. This can be rather important because it indicates that a receiver's subjective opinion regarding an information demand can create a situation where no information is received and thereby no change in the knowledge base will occur.

According to Hartley (1927), two people that speak different languages have a limited amount of symbols and thereby the possibility to transfer information into knowledge is limited. This indicates that there is a need for some kind of mutual knowledge base among the user/receiver of specific information. This knowledge base should be acquired by training and education, and in some sense it will be acquired/updated by experience. In the assembly plant, it is impossible to sustain quality without training, but the focus should be on having an information system that can be managed by the user in a way that minimizes the training and the quality risks. According to Nembhard and Uzumeri (2000), a manufacturing worker is constantly forced to learn new skills, technologies and processes in order to keep up with the increasing demands within the industry. Therefore it is important to design information systems so that they are easy to use, do not need constant updates and do not need advanced training. The opposite would be an information system that is advanced, in need of constant update due to for example product updates and an information system that constantly is changing where the changes are connected to part numbers. There are several problems connected to this and the one of most interest is connected to minor changes, such as new part numbers on parts that still look the same, that create a change in the product set up. These changes sometimes make very little impact on a user interface layout and that is a problem. According to Marsden and Green (1996) one of the main reasons why systems fail to perform well is that their desired objectives are poorly designed. The effect of a combination of small changes in product set up and a poorly designed interface might be seen in the quality output.

It might be important to reflect on "Human Errors" and the four error categories presented in Section 2.4.7 and how the quality problems at the assembly plant are related to these categories. According to Park and Lee (2007) there is a problem in obtaining empirical data within industry, which forces a heavy reliance on judgement from experts. It is not possible to understand how much empirical data Park and Lee believe it is necessary to have to be able to do investigations of the root cause for an error. This research is not conducted by an expert in the field of Human Errors, therefore the focus will be on the human error definition *Mistake* and *Rule-Based Human Errors* and the reason for this is that it is possible without any expert skills to find quality problems that are related to this definition. The definition of Rule-Based Human Errors is:

Rule-Based mistakes occur when a person believes that they know the situation and the environment where they operate and plan an action based on this knowledge, but overlook signs in the environment that can indicate other types of actions (Rasmussen et al., 1994).

The question is, is it possible to apply this in some way to data/information, Situation Awareness, and Attention? To summarise some of the problems that are connected to these three aspects:

- Data/information:
 - Failure to identify information need.
 - Failure to identify information demand.
 - Have the knowledge but overlook signs/change in the information flow.
- Situation Awareness: Endsley and Robertson (2000) describe and divide the different causal factors between the three levels.
 - Level 1: Failure to correctly perceive the situation.
 - Information not available.
 - Information difficult to detect.
 - Information not observed.
 - Misperception of information.
 - Memory error.
 - Level 2: Failure to correctly comprehend the situation.
 - Lack of/incomplete mental model.
 - Incorrect mental model.
 - Over-reliance of defaults values in the mental model.
 - Level 3: Failure to correctly project situation.
 - Lack of/incomplete mental model.
 - Over-projection of current trends.

- Other.
- Attention:
 - Missing exogenous signals.
 - Suppressed endogenous signals.

It is possible to identify two major causes from the listing above: *knowledge* and *attention*. Knowledge and how to gain it is an important part of manufacturing, but in this case, attention is the subject of interest. This is because of the strong connection between attention and the investigated internal rejects.

The problems connected to attention are, summarized from the text above:

- Failure to identify information need and demand.
- Users do not perceive the information in their environment.
- Missing exogenous signals.
- Suppressed endogenous signals.

What type of error (quality problem) can be connected to attention? A interesting part of Jones and Endsley's causal factors is that 76% of the errors in situation awareness are in their study caused by problems in perception of needed information (Endsley, 2000b). If one assumes that there is a connection between "problem in perception" and attention, one can argue that there is a high risk of omission due to changes of state in the information flow. This omission can be caused by a framing (that creates boundaries) effect that limits the possibilities to choose information (Beach and Connelly, 2005), or a failure to observe provided information. The framing can be connected to a suppression of endogenous signals so that the information for some reason is not observed, i.e. a person misses the exogenous signals in the work environment.

The answer to the question "is it possible to connect data/information, Situation Awareness, and Attention to Rule-Based mistakes someway?" that was made earlier in this section, must be yes. In Section 2.2.5, a description was presented that describes the different internal rejects and their denomination. The interesting ones in this work are internally at the plant called:

- Component missing.
- Wrong component assembled.
- Extra component assembled.

These three have one thing in common and that is that they all are an effect of a Rule-Based mistake, at least according the definition presented by Rasmussen et al. (1994). Therefore, the conclusion is that a connection exists between them and that the common part is attention.

Based on the findings from the literature review, it is now possible to extend the information seeking process model that was presented in Chapter 1 in more detail, see Figure 2.21.





The model still contains the main components, *the trigger*, *passive* and *active reception*, and it has been updated with two *Situation Awareness* components, a failure due to *non-fulfilment of a demand/need* and *Reception Overload*. The important part of the model is the trigger and its effect on the following search process. Attention is not viewed in the model as a single step. Instead, it is a part of Situation Awareness as well as in the trigger.

The conclusion is that SA largely influences the outcome from a search process, and that attention is a part of SA as well as attention is a part of triggers. The three levels of SA: "Perceptions of Elements in Current Situation", "Comprehension of Current Situation", and "Projection of Future Status" all include parts that can influence the process and thereby the quality output.

Information Demand/Need, Fulfilled is the next step (Figure 2.21). Is it possible that there is a connection between the demand and the fulfilment of the demand and internal rejects? How does one know if a demand has been fulfilled? Is it the processor of data that determines if the demand has been fulfilled, or is it the orderer of task/action that has that responsibility? This is connected to goals and how they affect the performance and will not be addressed further in this thesis, but it still is an important part of a successful information search process.

As mentioned in Section 1.3.1, it is unlikely that an Information Overload is present, but there can be an overload from a perception point-of-view. Is it possible that all impressions from the work environment such as talk, music, heat, chill and noise can create an overload that prevents the object's possibility of perceiving, processing and acting on data/information in its surroundings? In a table presented by Wickens (1992) "Noise" is a parameter that can negatively affect, amongst other things, the short-termmemory. "Noise" can be both sounds and a data flow (with a purpose to be informative). It is possible that this noise prevents us from perceiving vital data in the work environment, making the use of our short-term-memory less effective, and creating stress (Wickens, 1992; Dubischar et al., 1999). Therefore, it is likely that Reception Overload (Figure 2.21) can exist in spite of the conclusion that it is unlikely that Information Overload is present.

If the *process* is successful the information is used in a correct, sufficient way and the outcomes from the *assembly process* are successful, we are back in the "High Volume"-area and the process starts over again (Figure 2.21).

The main issue is why the personnel do not use the data and the information provided by the information system. This together with the predictions based on historic facts and trend analyses, which state that the working environment within the assembly area in general and at the plant specifically, indicates that the assembly area will be facing an increase in complexity. Insufficient use of information combined with complexity is already today a contributor to the creation of internal rejects. It is notable that the production volumes produced today would not be possible without the existing information system and the work environment complexity is a function of the demand from the customer.

The knowledge gained from the analysis phase together with questions raised by the same was used in the case studies presented in Chapter 3, and in Chapter 4, where a work method is presented.

2.6 CONCLUSIONS

The conclusions from the analysis stage are that the amount of information, the way the information is presented, and most importantly, attention are the main influences on the internal reject rate of the described categories as well as productivity.

The amount of Information

The theory regarding Information Overload is supported by the theories of Late and Early Selection and show that it is likely that information is weighted and filtered before a human processes it (Section 2.4.6). The conclusion is that the quality of information affects the assembly personnel in a negative way. The study of the assembly work context together with the findings in literature supports this conclusion, and it also supports the Information Search Process that was developed in more detail and presented in Section 2.5.

Information Presentation

The conclusion is that the way the information is presented can affect the use of the short-term-memory in a negative way. This is connected to the statement above, but is more focused on the way a specific message is designed. If a message is designed in the wrong way, the working memory can be affected in a negative way.

Attention

Before an actual Information Search Process, a trigger must be present to start it. If this trigger is absent or not observed by the assembly personnel, the quality risks (influencing the internal reject rate) increases.

This conclusion is based on the findings in the analysis chapter, and is the base for the next step in the research, the investigations (Chapter 3). These investigations will include tests of how different types of information trigger a search process and how this and the information itself affect the quality and productivity.

CHAPTER 3 INVESTIGATIONS

This chapter presents the investigations performed, the conclusions from them, and the possible solutions to solve the problems identified.

The conclusions from Chapter 2 state that there are three major issues related to:

- the amount of information.
- the way the information is presented.
- and most importantly attention.

These are believed to affect the internal reject rate of the specified categories.

They are connected with attention and the amount of information in different levels of the information search process and need to be investigated further.

The investigations are divided into:

- Case study 1: The purpose of the study was to evaluate how and if a drastic change in the user interface would affect the assembly workers and their ability to handle attention and information, and if this would affect the rate of internal rejects.
- Case study 2: The purpose of this study was to evaluate, in a laboratory environment, to what degree the user interface would affect the quality as well as productivity.

In this chapter some of the data connected to internal reject rate will be recalculated from a percentage to a value compared to a baseline. There are two reasons for this. Firstly, some of the data connected to quality issues are classified and cannot be used outside the company. Secondly, the recalculation of the internal reject rates so that they can be compared to a baseline makes them more readable and understandable.

3.1 INVESTIGATION METHODS

The research methods used have varied depending on the studies performed and differences in the research environment. The general approach is that the studies are performed from a mixed methods research perspective. Mixed methods research involves in general research that collects, analyses and interprets qualitative and quantitative data in a single study or in a series of studies investigating the same phenomena (Leech and Onwuegbuzie, 2009). The methods utilised in the research will be described in this section.

Triangulation

Triangulation is described as "the combination of methodologies in the study of the same phenomenon" (Denzin in Beach, (2001). According to Beach (2001), there are two forms of triangulation, within-method and between-method. For example Case study 1 presented in Section 3.2 is a between-method study where quantitative and qualitative methods are combined to interpret and study the data retrieved. Triangulation implies a fixed point in space that is scrutinized from different angles/directions (Patton, 1980; Decrop, 1999; Stake, 2000; Bratthall and Jörgensen, 2002) or a point in space located by the knowledge of two angles and a side in a triangle (Nationalencyklopedin, 2005). According to Decrop (1999), there are four different types of triangulation:

- Data Triangulation: uses a variety of data sources in a study.
- Method Triangulation: uses different types of methods, for example Quantitative and Qualitative methods.
- Investigation Triangulation: several researchers interpret the same data.
- Theoretical Triangulation: uses a number of perspectives to investigate a single set of data.

Quantitative and Qualitative

According to Danzin and Lincoln (2000) the difference between quantitative and qualitative research is that while quantitative research focuses on the measurement and the analysis of the connective relationship between variables, qualitative research focuses on the processes and qualities of entities that are not experimentally examined or measured. In quantitative methods the data are characterised in terms of, for instance, quantity, amount, intensity or frequency. Some of the common methods of collecting

qualitative data are by observation, interviews and/or examination of documents (Lyons, 2002). In qualitative methods the data are characterised in terms of, for instance, process, social structure, inquiry, direct observation or interviews.

Interviews and Questionnaire Surveys

Interviews can be used in all stages of a research process (Breakwell, 2002). The main use for interviews is when there are no clear answers or numerous answers and the population is limited (Andersen and Schwencke, 2002). According to Andersen and Schwencke (2002) the method is interesting when it comes to creating trust between the interviewer and interviewee, which is very hard to accomplish with quantitative surveys. This can be of great importance if the data one wants to retrieve is of a more sensitive nature, for example how persons act regarding work quality.

In this research work, two types of interview methods are used:

- Structured Interviews: the interviewer uses a fixed set of questions in a fixed order.
- Unstructured Interviews: the interviewer has a number of topics but the questions and their order are not fixed.

According to Patton (1980), the purpose of an interview is to find out what is in a person's mind. The important part is that it is not possible to observe everything, for example feelings, thoughts, etc. Therefore, interviews are a possibility for the interviewer to enter the interviewee's world.

In this thesis, the qualitative data was gathered by interviews, both structured and unstructured, and questionnaire surveys.

One of the purposes with a questionnaire survey is to gain knowledge about peoples' thoughts and feelings (Shaughnessy et al., 2003). The results from a questionnaire survey are often used to describe peoples' opinions and attitudes. Major benefits with surveys compared to interviews are according to Andersen and Schwencke (2002):

- That it makes it possible to get answers from a large population.
- That the influence of the interviewer on the interview subject is minimized.
- That it makes it possible to reach people that are geographically spread.

The drawbacks can be according to Andersen and Schwencke (2002):

- Problems with low response rate.
- Problems with knowledge regarding the interpretation of the questions, i.e. how did the person answering the survey interpret the questions asked?

3.2 CASE STUDY 1- INDUSTRIAL ENVIRONMENT

This section describes Case study 1 and the different parts within the study, how and why they were conducted. It also describes the outcome and results from the different parts and how they contributed to knowledge.

3.2.1 Study outline

The aim of the study was to find evidence supporting one of the hypotheses "generated" from the research aim (Section 1.3):

The aim of this research is to find or develop a prototype work method that can support the design of Information Flow based on product and process demands.

The hypothesis:

Hypothesis 1 - 'Information Seeking Behaviour'

The degree to which Active Information Seeking behaviour is supported/triggered has a large influence on the number of internal rejects.

EXPECTED RESULTS

The expected results are practical knowledge regarding how active and passive attention affect the internal rejects, knowledge regarding how the assembly personnel interact with the information flow present in their work context and how the information affects quality via the assembly personnel.

PERFORMANCE INDICATOR

To evaluate the knowledge value it is important to identify performance indicators. A definition of a Performance Indicator is: "Performance indicators compare actual conditions with a specific set of reference conditions" (Dantes, 2008). In this case the main reference performance indicator is the historic data regarding the internal rejects.

The historic data covers the period 13th March 2006 to 22nd December 2006 and includes approximately 33000 records. This data includes (the most relevant presented here):

- Date and time: Every reject registered gets a date and time stamp. This is used by the company and its quality department within the quality work to trace the error source. It will be used in the study to connect the rejects to a specific time period, i.e. when the actual study takes place in the work environment these time stamps will connect the internal reject to the study time frame.
- Engine family and engine variant: The data covers the five engine families, 6, 7, 12, 13 and 16 litre engines. The engine families exist in different variants (named in the historic data as "engine type"). Approximately 500 variants exist. The focus is the 13 litre engine, but data for 12 and 16 litre engines was a part of the preparation and evaluation work.
- Effect number: This number is a code used for identifying different types of rejects.
- Effect description: The study focused on effect numbers 141, 142 and 144 described as:
 - o 141: Component missing.
 - 142: Wrong component assembled.
 - o 144: Extra component assembled.
- Part number: A number connected to a specific individual part used for identification.
- Free text field: This field can/should be used by the quality assurance personnel to describe the cause of rejection. Unfortunately the personnel do not use this option in a reliable way; therefore this data is used with caution within the study.

DATA RETRIEVAL PROCESS

The historic reject data process (Figure 3.1) starts with a discovery of a divergence of the assembled engine compared to the order specifications. This is done at the end of the line and in the test zones. If a divergence is discovered, the personnel make a note in the computer system and save it in a database. After registration, depending on the severity of the divergence, the engine is transported to a reject area where the problem is solved. The manually registered data is stored in a database.



Figure 3.1: The reject report flow, from reject finding to historic data presentation. The numbers 12, 13 and 16 refer to the engine size.

The most common way of retrieving data from the database is via a Business Objects (BO) inquiry. The result, the data, can be copied and inserted into an Excel spreadsheet. This data has to be prepared for viewing, i.e. the retriever must process the data into information and the data is compiled and viewed in for example a Pareto diagram (will be further addressed and explained in Section 3.2.2).

PREPARATION

In this case, the data can be sorted in a way that makes it possible to identify:

- Engine family: 12, 13 and 16 litres.
- Station: For example, the 13 litre engine line is divided into basic and final assembly (two parallel flows). This makes it easy to connect a reject to a specific assembly station. In this work most attention was focused on four assembly stations (2 stations \$0800 and 2 station \$1100 on parallel assembly lines):
 - S0800: Main component is servo pump.
 - S1100: Main component is cable harness.
- Date and time when the quality problem was detected.

PRESENTATION

Three different applications were used to create usable information from the historical data. The applications are:

- Excel: Used to compile data and create diagrams.
- GB-Stat: A tool created for Volvo Cars and used by Volvo Powertrain to create statistical diagrams.
- Minitab 14 (Minitab, 2009): The application was used to calculate and present suggestions regarding the power and sample size required.

3.2.2 Experimental design

The experimental design was based on a quantitative field experiment in an assembly environment. The analysis part contains both quantitative and qualitative evaluations.

EXPERIMENTAL ASSEMBLY ENVIRONMENT

The experiment was conducted in the actual production environment. That is, the experiment took place in the assembly line and only small changes were made to the environment. The production at the assembly line continued and the conditions were more or less the same as on a regular production day. This included such things as carriers, forklifts, hand tools, material racks etc. The personnel involved in the study were all employed by the company and performed their regular duties.

The case study was presented to the personnel on a total of four occasions during their regular information meetings. During these meetings, the personnel were introduced to the experiment and had the possibility to ask questions and discuss the study.

The experimental environment and the assembly environment, as it was before and after the case study, differ only in the part that is included in the study, in this case a *trigger*. The triggers were magnetic rubber sheets with a size of approximately 300x300x0.85, 300x50x0.85 and 60x60x0.85 millimetres (length, width and thickness) attached to the carrier (Figure 3.2 and 3.3) and 400x300x0.85 attached to the material racks (Figure 3.4 and 3.5). The trigger had two purposes: firstly, to create awareness that a different engine variant was to be assembled: secondly, to give information regarding what part should be used.



Figure 3.2 and 3.3: The location of sheets on the carrier for the triggers.

The sheets were attached to the pallets and shelves in the material racks with a simple metal hook. This made it possible to quickly move the sheets when new parts arrived at the station.



Figure 3.4: The location of the rubber sheets on the material racks in assembly station S0800.



Figure 3.5: The location of the rubber sheets on the material racks in assembly station S1100.

At the beginning of the assembly line, see Appendix A for the layout, a station was built specially for case study purposes. This station was manned day and night during the study period, 12.00 24th May 2007 to 00.00 6th June 2007, and was responsible for engine identification, attachment and detachment of the triggers to the carriers (Figure 3.2 and 3.3). The changes made to the assembly stations can be seen in Figure 3.4 and 3.5. The only updates to the four stations were the rubber sheets that gave information regarding the position of specific parts.

COLLECTION OF QUANTITATIVE DATA

The results from the study were obtained by comparing the performance indicator before and after the study. As mentioned earlier in the text, this indicator is a part of the reject handling system and was recorded for approximately 9 months during 2006.

HISTORICAL DATA

The data gathered during the period of 13^{th} March 2006 to 22^{nd} December 2006 shows that the performance indicator for the period has a mean value of 1.46% (Figures 3.6 and 3.7). However, depending on the sorting of the reject data two more values must be considered with a higher mean, in this case 2.77% and 3.12%.



Figure 3.6: Performance indicator sorted per week.

Performance indicator, value:

- 1.46% (from now on known as $H_{\alpha 0}$): The result for this indicator includes only rejects with the effect number 141, 142 and 144 (for a detailed description, see Section 2.2.5).
- 2.77% (from now on known as $H_{\beta 0}$): The result for this indicator includes $H_{\alpha 0}$ and rejects that have a non-assembly related attribute. A "non assembly related attribute" is an attribute that is used by the quality inspection personnel with the aim of categorizing the reject to a specific part of the assembly process. Attributes can be for example method, material or machine.
- 3.12% (from now on known as $H_{\gamma 0}$): The result for this indicator includes rejects from $H_{\alpha 0}$, and $H_{\beta 0}$ as well as rejects connected to categories 300, 400, 420, 440 and 450. These types of reject effect numbers indicate leakage of oil and water. Parts missing or wrong parts assembly (141, 142) can, if not found earlier, lead to a leakage in the final test zone. If this occurs, the reject registration will not be in the category "parts missing" or "wrong parts assembled", it will be "leakage" or "deviation from specification".



Figure 3.7: Performance indicators, $H_{\alpha 0}$, $H_{\beta 0}$, $H_{\gamma 0}$, *sorted per assembly station.*

A Pareto diagram is used to view the most important factor. This is done by simply applying the 80/20 rule that suggests that generally 20% of causes in any given situation will account for 80% of the problems. Pareto analysis ranks categories of items in order of occurrence or severity and the form of the bar graph is referred to as a Pareto Diagram or a Pareto Chart (Greenall, 2004; Galaktionova et al., 2006).

A Pareto Chart can be used to:

- Display the relative importance of factors.
- Compare the extent of problems before and after improvements.

This helps to:

- Select the starting point for problem solving.
- Identify major causes of problems.
- Prioritise improvement actions.
- Facilitate resource allocation decisions.
- Monitor improvements (Greenall, 2004; Galaktionova et al., 2006).

In the Pareto diagram in Figure 3.8, the left vertical axis shows the number of internal rejects connected to the different assembly stations in the assembly line during a time period. The right vertical axis shows the percentage of total internal rejects associated with each station. The horizontal axis lists the station name in order, starting with the station with the highest internal reject frequency on the left moving to the least frequent on the right. As a general rule, the tallest bar should be tackled first, moving on to the second once the first problem (station in this case) has been successfully reduced.

According to the Pareto diagram (Figure 3.8) and the above, the focus should be on stations S0800, S0500, S0250, S1100, S1040, S0200, S1400, S0700, S0600 and S1300. The analysis of the historic data in the Pareto chart⁹ (Figure 3.8) shows that the internal reject rate on station S800 represents almost 20% of the total number of internal rejects.

⁹ A Pareto chart can be used when the relative importance of all problems must be graphically represented with the purpose of viewing the most important factor. (Galaktionova, Y., A., Ibraqimova, E., K. & Bekisheva, E., V. (2006). On product quality control. *Fibre chemistry*, 38 (1), 79-82.



Figure 3.8: The Pareto diagram for the internal reject rate during the reference period. All data are of the $H_{\alpha 0}$ *category.*

The historic reject data for station S800 were further investigated and the results for the station can be seen in Figure 3.9. In the figure, the results for two periods are presented, the reference period and the amount of reject during the month of June 2006, a period within the reference period. The purpose of investigating and presenting data for June 2006 is to determine if a major difference could be present during the month of June. The pilot study was partly performed during that month. The baseline¹⁰ (Hornby, 1997) for the reference period is set to 100. All results or measures regarding the internal reject rate will be compared to that baseline. In Figure 3.9, the internal reject rate for the period of June 2006 is 185 in the relation to the baseline.

 $^{^{10}}$ A line or level used as a base for measuring or comparing. The values is calculated by dividing the measured results, e.g. 68 with the value of the baseline e.g. 32, the and then multiply the result with 100 (baseline value). In this example, would the results be (68/32)*100=212,5.



Figure 3.9: Results from the investigation of station S800 and its internal reject rate.

Station S1100 actually has only the fourth highest rate of internal rejects but nevertheless was chosen as the second station in the study. This decision was based in part on the internal reject historic data, but in this case interviews with the production technicians also influenced the station choice. According to the production technicians there were some problems connected with this station that were rather severe. The problem was that if an internal reject occurred it could cause a high level of rework on the engine. Figure 3.10 shows results for two periods, the reference period and the reject index for the month of June 2006, a period within the reference period.



Figure 3.10: Results from the investigation of station S1100 and its internal reject rate.

PARTICIPANTS

Participants and their roles in the study were:

- Experimenter:
 - Designed experiment, defined scope and frame for case study.
 - Informed participants, production leaders, factory management and technicians about the study and how it could affect the daily work.
 - Created and gave instructions to participants regarding how they should act during the study.
 - Trained and assisted participants that had an active role in the study.
 - Compiled and assembled work orders used by assistants at the beginning of the line during the study.
 - Gathered and compiled data before and after the study.
 - Analysed data, reject statistics, before and after the study.
 - Actively worked within the study with the experiment.
 - Conducted interviews with participants after the study.
 - Created and distributed information regarding the results.
- Assembly workers:
 - Assembled the engines according to the defined assembly process.

- Reject data registration was conducted by assigned persons from the assembly personnel. This is part of the regular assembly process.
- Assistants:
 - Applied and removed the triggers on the production equipment. Registered engine numbers and colour coding, i.e. connected the specific engine individual to a specific trigger colour setup.
- Management: The assembly plant managers had an indirect but important part in the study. Their role was to support the work and thereby create the possibility for the study to be conducted.
- Technicians: Their role was to support the study with their expertise regarding the assembly process.
- Material handling personnel: The material handler was indirectly an important person, as his/her main responsibility was to deliver material to the assembly line. The handler also moved the colour coding material from the empty material containers to the new full containers.
- Quality coordinator: Expert regarding the work quality in the assembly line.

The experimenter, case study leader and author were the same person, and performed the roles based on previous experience and training.

The experimenter had knowledge regarding:

- Manufacturing environments:
 - General: General knowledge regarding manufacturing from previous experience within the industry. Work close to the assembly process as a maintenance engineer and builder of automated assembly machines for a period of approximately three years. Work as a consultant for approximately two years within the field of systems engineering at the plant where the study took place.
 - Specific: Work at the assembly plant for a period of more than three years in the Research and Development department.
 - Projects: A total of five projects have been conducted at the plant that have contributed to the preparation of the study.

It should be noted that a majority of the assembly personnel responded very positively to being a part of the experiment. Typical reactions were:

- Interesting to see if the colour coded interface would affect their work situation in any way.
- Positive to the fact that someone was working with problems that they had to handle every day.
- That it might be easier to introduce new personnel to the assembly environment.

FIELD EXPERIMENT

The field experiment was conducted day and night during the period 12.00 24th May 2007 to 00.00 6th June 2007

Assembly process, before study

This process is presented in Section 2.2.1 and in Figure 2.9.

Assembly process, during the study

The process during the study is in a sense the same as before the study except in one important aspect, and that is the trigger. The trigger is a predecessor to "1. Identification of parts....."

- 0. Trigger.
- 1. Identification of part to assembly. This identification process is not necessary on 100% of the stations as some of the assembled parts are identical on all engines.
- 2. Retrieval of part to assemble. The personnel gather part to assemble.
- 3. Assemble part on engine.
- 4. Repeat steps 1 to 3 until all parts are assembled.
- 5. Confirmation of task. The personnel confirm that they have done all assigned tasks at the station; this is done via an IT-system.
- 6. Carrier departs from the assembly station to the next station.

Equipment used

The experiment was performed on site in the assembly factory, in a production assembly line. Therefore, the equipment used was the same as during regular production. This includes such things as:

- Carrier.
- Information System.
 - Computer interface.
 - Paper instructions.
- Lifting aids.
- Etc.

No prearrangements were executed that influenced the production context, except arrangements for the handling of the triggers (the magnetic sheeting). A workstation at the beginning of the assembly line was built (see Appendix A: Layout assembly). This workstation contained a table, notebook computer and a chair.

Accessories

Accessories used within the assembly environment specific for the case study:

- Magnetic rubber sheets.
- Notebook computer.
- Production schedules.
- Paper instructions specific for the study.

Case study influences

In all experiments there is a possibility that the experiment itself influence the results. In this study four experiment groups were used with the purpose of creating an understanding of how this particular study might influence the personnel and thereby the results. The different experiment groups were separated by two different variables:

- A: Interaction: Interaction will be varied between A1:"High", experimenter actively interacts with the assembly personnel at their workstations (ask questions, starts discussions about their work etc.), A2:"Low", very little or no interaction between the experimenter and the assembly personnel.
- B: Trigger: B1:Present or B2:Absent. The triggers are the magnetic rubber sheets attached to the carrier mentioned earlier.

Control group A2B2: The results from this group are the historic data from the period 13th March 2006 to 22nd December 2006. These results will be used as a reference value for the experiment groups.

Experiment group A1B1: High interaction and trigger present.

Experiment group A1B2: High interaction and trigger absent.

Experiment group A2B1: Low interaction and trigger present. The experimenter will be situated at the beginning of the assembly line.

3.2.3 Results

Quantitative evaluation of the results

The evaluation of the data showed:

- Quality: A decrease in the number of internal rejects from assembly station S800 and S1100 during the period 12.00 24th May 2007 to 00.00 6th June 2007, compared to the reference period 00.00 13th March 2006 to 00.00 22nd December 2006.
- Productivity: No effect could be found regarding the station time, i.e. there were no differences, before, during or after the study regarding the assembly time.

The data showed that the rejects during the study decreased to 59 (baseline = 100, Section 3.2.2) for station S0800 (Figure 3.11) and to 0 (baseline = 100) for station S1100 (Figure 3.12).



Figure 3.11: The internal reject rate for station S0800.



Figure 3.12: The internal reject rate for station S1100.

The decrease during the study for station S0800 was approximately 41 points compared to baseline (Section 3.2.2.) while for station S1100 the decrease was 100 points, i.e. no rejects occurred during the case study period.

Qualitative evaluation of the results

A qualitative approach was added in the later part of the study. The main reason for this was to gain a better understanding of what caused the results - the decrease in the number of internal rejects.

There were two parts to the qualitative evaluation, a questionnaire survey and an interview. The survey was conducted among the personnel that were directly affected, those who work at station S0800 and S1100, and indirectly affected those who in some way were a part of the manufacturing system in question, the assembly line for the 13 litre engine. The interviews focused mainly on the persons from the support organisation, for example line manager, quality technicians, production technicians.

Questionnaire Survey

The questionnaire survey was conducted among 171 persons that worked or work in the manufacturing system in question. The questionnaire was in Swedish, but a translation to English can be found in Appendix C.

- 171 individuals participated in the questionnaire survey.
- The response rate was 100%.
- The questionnaire contained 36 elements.
 - The elements were formulated in two different ways:
 - As a statement: for example "The use of Colour Coding worked very well", and the alternative (in this case) was on a 7-point scale, from "Agree Entirely" to "Disagree Entirely" with "Neither or" in between.
 - As a question: "Do you feel secure with the Colour Coding that was used in station S0800 and S1100?". The answer alternatives were; "Yes", "No", "Sometimes", "Sometimes not". The two latter alternatives had follow-up questions where the individual was asked "When?" and "Why?"
- The personnel were encouraged to write as many remarks as possible. The last page in the questionnaire was a blank page which the individuals could use for feedback to the researcher.
- The questionnaire survey took place on four occasions, one time for each shift and once per subject. The researcher was present during the time when the personnel filled in the questionnaire.
The questionnaire was divided into sections. In the first section the personnel were asked to state year of birth, gender, main assignment, years at the company, years in the assembly department, shift (day, evening, night, weekends) and if they had worked on station S0800 and/or station S1100. If the answer was No on the last question they where directed to question 2.8 in Section 2. The data collected regarding year of birth, gender, main assignment, years at the company, years in the assembly department and shift where not used further in this thesis, but might be used in the future as foundation for experimental design purposes.

Section 2 started with specific questions for those who had worked at station S0800 or/and station S1100. There were seven questions that were directed towards the Colour Coding. One of the purposes with the questions was to establish if there was some connection between the results from the quantitative evaluation and what the assembly personnel had experienced, but also what the personnel felt when they worked with the colours. The results were interesting in more than one respect; the first question that the individual had to consider was if the Colour Coding had any positive effect on the quality and 87% answered that it had affected it and 23% stated that it affected it "to a very high degree".

The answer alternatives to the question (Figure 3.13):

"Do you believe that the Colour Coding used on station S0800 and S1100 influenced the quality in a positive way?"



Figure 3.13: Answer distribution for question 2.1 in Appendix C.

When the assembly personnel answered the questionnaire, they did not know for a fact that quality was the performance indicator used in the study to measure the results. This can be of importance if one believes that a placebo/Hawthorne¹¹ effect can in any way influence the participants in a study.

On the question (Figure 3.14):

"Did you feel secure with the Colour Coding that was used in station S0800 and S1100?"

¹¹ The Hawthorne effect: an increase in the output due to experimental manipulation. For more information see e.g. Holden, J. D. (2001). Hawthorne effects and research into professional practice. *Journal of Evaluation in Clinical Practice*, 7 (1), 65-70.



Figure 3.14: Answer distribution for question 2.2 in Appendix C.

When they were asked why they did not feel secure or trust the colour, the answer was that they did not trust the persons applying the Colour Coding to apply the right colours. Some made a remark in the questionnaire at the question and stated that the colour coding applying process was a manual process and that this could be negative from a human error perspective.

A core issue is whether or not the information presented supports the assembly task in a correct way. The results from the survey (Figure 3.15) show that a majority of the personnel is of the opinion that the amount of information that they receive via colour coding was enough for them to successfully perform an assembly task.



Figure 3.15: Answer distribution for question 2.3 in Appendix C.

The remarks and the answer distribution show that the information provided via the colour was sufficient from a task performance perspective. Examples of the remarks made by the assembler (descriptive comments in brackets)¹²:

... facilitates a lot on Station 1100 [cable harness]. It [the colour coding] provided a much better feedback on which cable harness I had to assemble. Very positive to the reduction of information regarding what to assemble on the engine...

...I think that colour coding was a superb way to view information; you are spared from unnecessary information...

...colour coding makes it possible to quickly see what to do...you are a lot more attentive [with colour coding], can focus on the important...

...harder [without colour coding] to pick parts, many part numbers are rather similar, easy to read on the wrong row [graphical user interface]...

To minimise losses connected to assembly it is important to eliminate waste. An important part of information presentation is to present the information at the right time, when the information is needed. There is more than one reason for this, in this case it is interesting from the two perspectives of quality and productivity.

¹² The quote translated by researcher.

In the questionnaire the personnel were asked if the colour coding could contribute to their knowledge regarding what to assemble next (Figure 3.16).



Figure 3.16: Answer distribution for question 2.6 in Appendix C.

Some of the remarks, oral and in writing, state that a problem exists regarding when the information is presented to the personnel. In the existing system the information is presented to the personnel when the carrier arrives at the assembly station, but according to the personnel the need actually arises when they are finished with the current task. Since the information is absent until arrival of the carrier the personnel are forced to either:

1. Wait until the carrier arrives at the station. That is, do nothing until the carrier arrives.

- 2. Walk to the previous assembly station and by looking at the engine on that station try to, by using their knowledge, investigate what to assemble at their station (the next station).
- 3. Try to predict what is to come and prepare for it.

The first and second alternatives create waste and contribute to the productivity in a negative way. The second and third can directly affect quality negatively. According to the personnel having the information present at the right time is important for an effective assembly process. A suggestion showing a user interface that contributes to the work process with an early information presentation will be presented in Chapter 4. The study has shown that an information system which supports early information presentation is beneficial for productivity as well as quality, and is an important support for the personnel.

3.2.4 Discussion of case study results

The information process and the colour coding used within the study enhanced the information search process by:

1. Presenting the information so that the assembler can see the information from their station (Figure 3.17).



Figure 3.17: Example of what the personnel can see regarding what type of variant is approaching their assembly station.

- 2. Making it easier to identify from a distance what to assembly.
- 3. Providing the information so that the information is presented when a need has occurred.
- 4. Eliminating the need to learn by heart where in space the parts that should be assembled are located.
- 5. Creating a less demanding workload due to the elimination of noise (i.e. redundant information).
- 6. Triggering an information search process when it is needed.
- 7. Supporting part changes, for example when a design change note is initiated. When this occurs, it is possible to make the change without changing the colours in the work environment. If a new part is added new colour can be added. By using the existing colours or just adding a colour, the need to learn by heart where the parts are located, the productivity losses during the learning and the training needs are lowered in a significant way.

These enhancements are directly connected to the quality and are also believed to considerably affect the productivity.

An interesting addition to the productivity evaluation was initiated by a remark from one of the participants. According to this statement the assembler could decrease the station time from seven to three and a half minutes during the study, i.e. a productivity increase was an effect of the use of colour coding.

Therefore, an evaluation of the station time was conducted, and was a direct effect of Case study 1, but was a part of the results evaluation process.

This evaluation included a Business Object question that made an inquiry based on the date and station, in this case 24th May 2007 to 6th June 2007 and stations S0800 and S1100. The results from that inquiry were compared with inquiries for a period before and a period after the study. The results showed no affect on the station time during the study. In an interview session with the production technician, quality coordinator and production manager one of the quality coordinators described this as (descriptive comments in brackets)¹³,

...I have worked on the station [cable harness station] in question before I became a quality coordinator, and the first thing that struck me was that I felt more

¹³ The quote translated by researcher.

relaxed and that I work much more calmly ...the pre-work [pre-work is a way for the personnel to prepare themselves for the next task. If the pre-work is used in a wrong way, the quality can be affected] changes, I did not feel a need to prepare myself in the same way [compared with the way he did when there was only data/information provided by the computer screen]...the number of pumps [servo pumps] that I needed ["need" in this case is a subjective conclusion that differs from person to person and is not a part of the described assembly work process] to prepare could be "lowered" because by just looking at the previous station and the carrier at that station I knew what was to come...

The comments from the quality coordinator indicate that the pressure from the assembly tasks on him was less when he could use the colour coding. It also indicates that the task could take less time, but in this case the person used the time savings to change his work process in a way that lowered the work burden. The missing evidence that time was saved could be explained by the fact that the personnel did not use lowered station time to assemble more engines, but instead they used the time to lower their work burden. Another issue that it is important to keep in mind is the design of the report system and the production system. In this case the assembly line layout did not support an evaluation of the assembly time due to a queuing affect before and after the assembly stations. This statement is supported by the results from a question in the survey. In the survey the personnel that used the colour coding were asked to reply to:

"It was faster to use colour coding when I assembled"

They could reply to this statement on a 7-point scale that covers from "Agree Entirely" to "Disagree Entirely" (Figure 3.18).



Figure 3.18: Answer distribution for question 2.5 in Appendix C.

38% "Agree Entirely" that they could lower the assembly time by using the colour system.

The conclusion is that although it is not possible to show quantitatively that the cycle time was affected, it is possible to qualitatively suggest that it is plausible that such an affect exists. The statements made by the assembly personnel as well as the statements made by the quality coordinators should be weighted highly due to their great experience regarding the assembly context.

The results from the study clearly indicate that information has a strong affect on quality as well as the ability for the assembler to identify information needs. They show a possibility of increasing productivity. This will be addressed further in Section 3.3.

The study supported the conclusions from Section 2.5 where it was stated that the amount of information, information presentation and attention influence the internal reject rate as well as the productivity. There are however some interesting issues that were not possible to see during the literature review. It was not possible to understand

that the timing of the information had an important influence on the assembler during their assembly tasks. The way that the information system was designed revealed that the feeling of control was an important aspect for the assembler. The system used during the study made it possible for the assembler to get information regarding what was to come and thereby it was possible for them to prepare both mentally and physically (gather parts etc.).

There are issues in this study that are hard to measure. The reaction among the personnel directly involved is one of those issues. One common reaction was that the reduction of the amount of information was one of the main benefits with the colour coding. This made it possible for the personnel to quickly find relevant information without noise from redundant information.

The feeling of control that the personnel expressed is also an important result from the study. It is difficult to measure how this might influence quality and productivity, but it is still an important input that supports hypothesis 2 (Section 1.3.1).

The main conclusion is that the study supported the substantiation of both hypothesis 1 and 2. By showing that a trigger, in this case a change of colour, affects quality in a positive way and that issues like the amount of information as well as the timing of the information might have been identified if a evaluation method had been used during the design of the information system.

During the analysis of case study results, four interesting issues were discussed with the personnel involved.

- The case study effect on the internal reject rate, according to the quality coordinator (Figure 3.19).
- 2. An effect on the internal reject rate directly after case study finished (Figure 3.19).
- 3. The use of case study information in other parts of the line.
- 4. Case study affecting other types of quality problems at station 800 and 1100.



Figure 3.19: Case study results including reject rates for a two day period after case study finish and quality coordinator statement.

The quality coordinators (in this case there were two) are personnel with long experience of the assembly line. They need to have knowledge regarding how to assemble an engine as well as how quality work should be performed. Their knowledge was used in the analysis phase of the study. According to their analysis of the internal reject rate for the case study period some of the problems counted should not be a part of the results. Therefore, according to the quality coordinators, the actual internal reject rate for the case study period should be as low as 15 point (baseline 100). Their conclusion should be highly weighted in this discussion due to their expertise, but the results from the study will still be presented as 59 points.

During the analysis phase, data from periods directly before and after the study were investigated. The data before the study do not reveal any major differences compared with the reference period. However, when investigating the data for the period 00.00 6th June 2007 to 23.59 8th June 2007 directly after the case study, some interesting findings were made. During this period the internal reject rate increased from 59 (15) to 420. It decreased in the following days to a level similar to that before the study. This increase is significant. An investigation was made regarding the manning of the stations, to see if any major changes had been made on the assembly station and if any new products/parts had been introduced during the period. This investigation revealed that no such change

had been made during either the case study period or the period directly after. No evidence could be found that could explain the significant increase. However, a reflection regarding this is that the personnel might have adapted to the colour coding during the study and that the high reject rate for the period after the study could be connected to the fact that they had to go back to using the information system used before the study. This indicates two things:

- 1. It is a demanding task to learn or refresh knowledge using the existing information system.
- 2. The learning curve and its influence on the assembly time can also affect the quality.

These indications are in some sense supported implicitly by results from Case study 2 (Section 3.3.4 and 3.3.5), but will not be addressed in depth in this thesis.

In discussions with the assembly personnel during and after the study it became clear that the information provided by the colour coding was used on assembly stations that were not part of the study. In particular one station, assembly of oil pan, used the information in a way that made it possible for the personnel to prepare the assembly task before the carrier with the engine arrived at the assembly station. This made it possible, according to the assembly personnel at the station, for them to lower their workload significantly. The task of preparing the oil pan is time consuming and can be a bottleneck if any problem occurs during the preparation work. Colour coding made it possible for the personnel to use the information earlier and thereby was it possible to use the time they had between engine departure and arrival in a more effective way.

The quality coordinators had also identified other effects that according to them were a spin-off effect caused by the study. The effects that they identified were on quality issues other than the ones that were a part of the study. According to the coordinator a quality problem connected to a cut o-ring (used as oil sealing between the servo pump and engine) decreased or vanished totally during the case study period. Their conclusion is that the time saved by the use of colour coding can lower the assemblers' workload and this can affect the way the assemblers perform their tasks.

These four issues are all an effect of Case study 1. Issues 2-4 were not part of the study initially, but they are an effect of it.

3.3 CASE STUDY 2 – LABORATORY ENVIRONMENT

In Case study 1, presented in Section 3.2, no effect on productivity could be measured. This is not that strange when one studies the production line. There are different ways to move a product through a production line:

- "Stop and go": the personnel for example press a button that indicates that he/she has finished with the assembly and that the carrier can move to the next assembly station. If the following station is occupied, the carrier will have to wait until the station is cleared.
- Paced line: all the carriers move at the same time.
- Continuous moving: the carrier is slowly moving forwards at a constant speed.

On the specific production line a Stop and go system is used. The station time is balanced for an assembly period of seven minutes, i.e. it takes approximately seven minutes to complete all tasks at an assembly station. This system makes it difficult to measure the time it takes to finish all tasks at a station due to uncertainties regarding the trustworthiness from such a measure. There are too many parameters that can affect the results. Therefore a laboratory experiment was initiated with the purpose of investigating if different types of presented data and/or information could influence productivity in any way.

Case study 1 showed that information could influence quality in a positive way. Unfortunately, it was not possible to evaluate if the same affect could be found connected to productivity.

Case study 2 used different user interfaces for the same assembly task and the hypothesis was that different interfaces could affect the quality and productivity.

3.3.1 Study outline

The aim with the case study was to find some evidence that supports the two hypotheses presented in Section 1.3.1:

Hypothesis 1 - 'Information Seeking Behaviour'

The degree to which Active Information Seeking behaviour is supported/triggered has a large influence on the number of internal rejects.

Hypothesis 2 – 'use of evaluation methods'

The use of an evaluation work method in a design process of an information flow will affect the internal reject rate in a positive way.

EXPECTED RESULTS

The expected results from the study were further knowledge regarding how information affects productivity and quality and whether or not the outcome supported the hypothesis.

PERFORMANCE INDICATOR

In Case study 1 presented earlier in the thesis the main performance indicator was the historic data gathered at the assembly plant. Due to the nature of Case study 2 no such data existed. The performance was measured between the three groups, described in detail in the experimental design, Section 3.3.2.

3.3.2 Experimental design

The experimental design was based on a quantitative laboratory experiment.

EXPERIMENTAL ASSEMBLY ENVIRONMENT

The experimental environment was specially designed for the experiment. It was located in a controlled area where only the participants had access.

PARTICIPANTS

Participants and their roles in the case study:

- Experimenters:
 - Designed experiment, defined scope and frame for study.
 - Informed participants, production leaders, factory management and university personnel (those affected) about the study, and how it could affect the participants work during the study.
 - Created and gave instructions to participants regarding how they should act during the study.
 - Gathered and compiled data before and after the study.

- Analysed the available data before and after the study.
- Actively worked within the study with the experiment.
- Conducted unstructured interviews with participants after the study.
- Compiled and distributed results.
- Assembly workers:
 - The assembly personnel assembled the product according to the defined assembly process.
 - Assistant/experimenters:
 - There were two assistants/experimenters during the study. Their main tasks were quality control, process control, timekeeping and disassembly.
 - Pilot tryout participants:
 - Tested the set up before the case study was started. The purpose was to find problems etc. with the set up before the actual start of the case study. These persons were not involved in the later stage of the case study.

The experimenter, one of the assistants and author were the same person and performed the roles based on previous experience and training. In total there were two experimenters, three pilot tryouts and 30 participants. A total of 35 participants were directly involved in the study.

Assembly process

The assembly process during the study:

- 1. Identification of part to assemble.
- 2. Pick part to assemble. The participant gathers part to assemble.
- 3. Assemble part.
- 4. Repeat steps 1 to 3 until all parts are assembled.
- 5. Confirmation of task. The participant moved the finished product to the quality control area.
- 6. The participant uses the order form on the computer to find out what to assemble next and the process starts over from 1.

Equipment used

The experiment was performed in a temporary laboratory (Figure 3.20) that was equipped specially for the case study. The equipment used was:

- Material front: metal set of shelves, three shelves high.
- Material boxes.
- Lego: within the study, Lego was used as parts to assemble.
- Information System:
 - Computer interface. Three different interfaces were used.
 - Paper instructions.
 - o Microsoft Excel.
- Assembly area.
- Video camera and video stand.
- Timer.

Study execution

The study was divided into four major phases:

- 1. Preparation: The preparation phase included design and building of a laboratory environment (Figure 3.20).
- 2. Test: A pilot test was conducted with the main purpose of evaluating the setup and to get feedback on the same. Three persons, one student and two employees at the university performed the test.
- Execution: The study was conducted by 10+10+10 assembly workers, i.e. three groups with ten individuals each, all workers from the reference assembly factory. Each worker started with a two minute training session followed by a 20 minute assembly session.
- 4. Follow-up: During the study the experimenters performed quality control of each final product and noted the assembly time for each finished product. This data was studied to evaluate if there were any differences in the assembly time and quality between the three groups.



Figure 3.20: Laboratory environment.

3.3.3 Execution

The study was divided into three sessions over a three-day period. There were 30 assembly workers that performed an assembly task and they were divided into groups of ten, ten each day. The task was to assembly a Lego design, see Appendix B, created specifically for the study.

The study started with a two minute training session where the assembler used paper instructions, see Appendix B, to guide them through the assembly task. As a support during the assembly task there was an assembly order presented on a laptop computer (Figure 3.21).

Rubrik	Antal	Instruktionstext
• 56109282	2	
• 55724016	2	
• 56109282	1	
• 57046203	1	
• 55724016	1	
• 57046203	1	
• 60668495	1	
• 56109282	1	

Figure 3.21: Example, user interface.

Different interfaces were used for each of the three days. They all informed the user on what to assembly and when, but they each had a different appearance. The framework for the design of the interface had its origins in the interface used at the assembly plant. The information presented in the interface was what part to pick, assemble, how many of the specific part. The numbers of parts in the assembly was the same but the parts could vary. This variation of parts (they differ from each other by colour and part number) in the assembly made it possible to create a real assembly situation.

The variants and their assembly order were randomised. The main purpose for this action was:

- 1. To simulate some of the framework existing in a real assembly environment where there are many product variants (Engstrom et al., 1995; Jinsong et al., 2005).
- 2. To create an environment where automated behaviours could be used (or rather be created) and the outcome from such behaviour could be measured.
- 3. To evaluate if variants influenced the productivity.
- 4. To evaluate if the quality of output was connected to specific variants.

After the assembler had been trained, the assembly process started. The first step was that one of the experimenters gave the user access to the interface so that the assembly process could be started. The assembler started gathering parts to assemble according to the order presented in the interface. As mentioned earlier, there were variant orders present that were randomly ordered, so the assembler had to always refer to the interface during the task to make sure that they assembled the correct final product. It was not possible to learn by heart what product to assemble.

After the assembler had finished assembly of a product he/she sent the product to the next station, quality control (refer to Figure 3.22 for layout and process flow).



Figure 3.22: Layout and process flow.

The quality control station managed two tasks, quality control and assembly process control. The experimenter studied if the different interfaces effected the process and if so how. The assembly process was documented via a video recorder. This made it possible to study the assembler and the assembly process in the analysis stage.

When the product had passed quality control, it was moved to the next station, timing and disassembly. This station measured the time it took for the assembler to assemble the final product. The experimenter started the timing when the assembler updated the interface, so that assembly of a new product was ordered. The timing stopped when the assembler sent the final product to quality control. The disassembly task includes disassembly of the final product into parts and putting the parts in the right place in the material rack.

The process was the same during each of the three days of the study. The difference, as mentioned earlier, was the user interface. In figure 3.21, the first day's interface is shown. The three interfaces had one thing in common and that was the information. During the three-day study, the final product that the worker had to assembly was the same. A pilot test was conducted where one of purposes was to evaluate if a change of interface affected the assembly process. The results from that test showed that the ability to assemble the product was not affected by the different interfaces, although the time it took to assembly the product varied.

Day 2 had an updated user interface, Figure 3.23, where the difference was the presentation of information.

Rubrik	Antal	Instruktionstext
• 56109282	4	
• 57046203	2	
• 55724016	3	
• 60668495	1	

Figure 3.23: User interface day 2.

The same product was assembled with the same number of parts. The interface was updated so that it was possible to minimise the amount of information.

Day 3's interface, Figure 3.24, was updated with a change from numbers to symbols, this was also the case with the material fronts.

Rubrik	Antal	Instruktionstext
• •	4	
• 3	2	
• 😳	3	
• 5	1	

Figure 3.24: User interface day 3.

3.3.4 Results

The results, Table 3.1, from the study were measured by the outcome of the two parameters quality and the change in productivity.

Table 3.1: Results from	case study.
-------------------------	-------------

	Day 1	Day 2	Day 3
Quality, faults detected	3	0	0
Productivity (average) Time/assembly 	48.3	34.5	31
Productivity (average) >30s/assembly 	33	113	217
Number of assembled	240	330	371
Average time for the assembly of first part	126	66	56

The productivity time average is the time it takes for the assembler to assemble the product. The productivity >30/assembly, is the number of products that were assembled in less time than 30 seconds, and numbers of products assembled during a period of 10 times 20 minutes.

3.3.5 Discussion of case study results

The results show that the effects on the quality were negligible. However, during the first day some of the assemblers used an assembly strategy that created a quality risk that was not measurable. This risk was connected to the fact that some of the assemblers tried to pick the parts first and put them on the table, and after that assemble the product. Some of them lost track of where in the interface they should look for information, and that led them to forget what parts they had picked already. The strategy "pick all parts first then assemble" was successful during day 2 and 3, but was not the right strategy during assembly day 1.

The conclusion is that the case study supported the one made in Chapter 2, i.e. that the amount of information, the way the information is presented, and most importantly, attention are the main influences on productivity. It did not support the hypothesis from a quality perspective in a direct way. However, by showing that information can substantially affect the productivity where the amount of information (totally and in each message) is the main issue it is possible to conclude that a large amount of information indirectly affects the quality. Thereby the case study supported the hypothesis indirectly by showing that the information flow affected the assembly personnel, especially during day 1, in a negative way and made them use more assembly time to find and process information. The results from the case study gave interesting and valuable insights into how the presentation of information can affect productivity and quality. It is rather clear that the total amount of information in a user interface affects productivity, as well as how the information in each message affects the productivity. An interesting fact is that the quality was affected negligibly by the changes in the interfaces. This is believed to be connected to the product that was, from a quality perspective, a rather easy product to assemble. However, during the study variants were assembled.

That the productivity was affected by the changes on the interface is obvious. But is it possible to connect the results to the hypothesis?

In Section 2.5, it was stated that there were three aspects, according to the analysis, that influenced the quality and the productivity; the amount of information, information presentation and attention.

In the case study it was not possible to evaluate if attention was an influencing factor on the outcome of the results. According to the information search process presented in the Section 2.5 attention affects the ability to enter the search process due to automaticity. It might not been possible to produce this automaticity during the short assembly time.

However, there are some interesting indications that automaticity did occur during the study and that this affected the productivity positively. As mentioned the product was easy to assemble and the variant only varied regarding the colours on some of the parts in the assembly. The size and where to assembly the part on the product were the same.

The data gathered during the study included assembly time for the high volume product (Figure 3.25) as well as for the variants, low volume products (Figure 3.26).



Figure 3.25: High volume product.



Figure 3.26: Variant, low volume products.

There are indications that productivity was affected negatively when the assembler assembled the variants. By comparing the assembly time for the product assembled before the variant with the time it took to assemble the variant (Table 3.2) is it possible to see that in some cases there are increases in the assembly time.

	Day 1	Day 2	Day 3
Average assembly	1.033	1.152	1.064
time index for			
variant			
Number of	52	79	88
variants			
assembled			
Number of	26	58	57
product variants			
assembled with a			
index >1			
The distribution	50%	73%	65%
for product with a			
assembly time			
index >1			

It is not possible to draw any final conclusion from this, but the indications show that there might be an effect on productivity. The indication (Table 3.2) is that the variant average assembly time is 3-15% higher than the assembly of the volume product. The distribution for the number of variants with an assembly time index higher than 1 varies from 50% to as high as 73%.

It is possible to see that 50%-73% of the variants assembled took between 3.3% to 15.2% longer time to assemble than the previous high volume product. It is believed that this might be connected to the handling of information and it also might be connected to attention. The reason to believe this is that the product itself is easy to assemble and this is not affected by whether or not the product is a variant. It is also possible to see an effect on the high volume product assembled directly after the variant. It seems that the assembly time on that assembly is affected as well. This negative affect levels away after a while.

The study of how variants affected productivity was not part of the case study, but is interesting and will be addressed in future research together with attention.

The decrease of the amount of information presented in the interface had a large influence on productivity (Table 3.1). The results show that an initially large amount of information can affect productivity significantly, in a negative way. It also shows that an interface design that considers human limitations and strengths, such as working memory, is more efficient.

One issue that was not a part of the initial case study scope was if different designs in the interface could affect the training. It is not possible to determine how the information flow affected the outcome of the short training session that proceeded the assembly session. The fact is that an affect can be seen on the results where the first product average assembly time decreased by more than 50% during day 2 and 3. This is very important from a training perspective and will be addressed in other projects connected to this research but it will not be addressed further in this thesis.

3.4 CONCLUSIONS

The conclusion from the studies conducted is that quality as well as productivity are affected by how well an information search process is initiated and performed, and how the information search process is support by the IT/IS-systems (including paper instructions, computers etc.). The results from Case study 1 shows that the initiation of a correct information search process directly influences quality in a positive way. It also shows that there are other effects that can be difficult to measure, but are still important contributors. The affects that the personnel experienced during the studies, and expressed in interviews and surveys, are important results. The feeling of control, decrease in presented information, the feeling of a productivity increase, the belief that colour coding influenced the quality positively, support for the personnel in their strategic planning, are all examples of results that can be hard to measure from a quantitative perspective, but are obvious from a qualitative perspective.

The results from Case study 1 support the two hypotheses, see Section 1.3.1:

1. The number of internal rejects decreased during the study. The reason for this is that the personnel interacted better with the information as a correct information search process was initiated by the colour coding. The changes in the colour made the personnel more aware that a variant was approaching their assembly station. That is, the personnel's attention was affected by the colours so that they could interact with the information system in a better way.

2. If an evaluation method had been used during the design of the information system, some of the design flaws might have been found. Examples of flaws are the amount of unnecessary information the timing of the information, and the problems with finding relevant information on the computer screens.

Case study 2 clearly showed that information and how it is structured and viewed have a large influence on the productivity. This obviously is the most important part of the outcome from that study. However, the study also indicates that there are more, less obvious, aspects that are of interest in future research. As mentioned in Section 3.3.4 it is likely that there is a connection between productivity and product variants. This might seem obvious, as if the product variant had been different in more significant ways and the difference would influence the assembly process more. In this case the only difference in the products was that the colours changed. This change influences the assembly process so that the assembler must adapt his/her information search behaviour. The adaption creates a need for an active search for information that can guide the person to the right part on the material fronts. It seems that although the assembler uses the information in a much more effective way during day 2 and 3, an affect can still be found on the productivity during variant information handling. A conclusion from this discussion is that one of the reasons for this is that during day 1 the need for an active (including active attention) search process was much higher and that caused the higher assembly times.

CHAPTER 4 DERIVED EVALUATION WORK METHOD

In this chapter an evaluation method will be presented and a description of how this method can be used for analysis of information need. A proposal regarding how evaluation work should be performed during a product lifecycle (PLC)¹⁴ is also presented. The aim is to draw attention to some of the underlying problems of evaluation work in different product lifecycle stages. The discussions are supported with results from a study where an evaluation work method was used.

4.1 INTRODUCTION

Two important conclusions have been made when it comes to the need for information in the assembly line. The conclusions are that two different types of information are needed and they are connected to *decision support* and *process support*.

- Decision support: Whilst decision support is needed during the whole assembly process, what is meant in this case is the support that makes it possible for the assemble personnel to pick the right part variant. For instance, by supporting the worker to pick the right variant out of a selection of five different servo pumps.
- Process support: Process support, in this case, is the support that guides the assembly worker through the assembly task. The focus is on how each part should be assembled on the engine.

In an assembly line, like the one in the reference plant, a number of product variants are assembled. To choose the right variant the assembler needs to consult the information in the assembly context, i.e. the assembler needs decision support. A successful outcome from an assembly task is based on a successful information search process. In earlier chapters a conclusion that has been reached is that attention is an extremely important part of the initiation of an information search process. There is a connection between

¹⁴ The product lifecycle is rather often thought of as the stages that a product go through, from "Introduction" to "End of Life" from a sales perspective. The stages are for example "Introduction", "Growth", "Maturity", "Decline" and finally "End of Life" Ryan, C. & Riggs, W. E. (1997). Redefining the Product Life Cycle: The Five-Element Product Wave. *Journal of Product Innovation Management*, 14 (3), 227-228.

variants that create a decision point, attention and the quality output. Therefore, it is important for a production technician to identify this decision point.

The parts included in a variant can be visually almost identical. They may have the same design and have the same numbers of connections (electrical, hydraulic etc.). But, despite the fact that they possess the same functionality, the design can differ in a way that influences the process, and thereby creates a process variant.

The part variant and the process variant demand different types of information. The part variants create a decision point where the assembler needs information of *what* part to assemble, the process variant creates a need for *how* to assemble according to a specific process that is a result of, for example, a part variant.

4.2 COGNITIVE WORK ANALYSIS

Usability Inspection Methods (a generic term) (Mack, 1994), Human-Computer Interface Design (Sutcliffe, 1995), (Cognitive) Systems Engineering (Rasmussen et al., 1994), Usability Evaluation Methods (Liljegren, 2006) all have one thing (at least) in common, and that is that they all use rather complex analysis methods. These analysis methods and cognitive system designs are well defined, for example in Cognitive Work Analysis (CWA) (Rasmussen et al., 1990; Vicente, 1999; Bäckstrand et al., 2005a), and is an important part of the design phase during the development for example of a new user interface. Examples of the use of CWA for the design of user interfaces can be found in domains like nuclear power plants, aviation, military use and medicine (Vicente, 1999; Sanderson, 2003; Liljegren, 2006).

During an investigation in the early stage of this research a study (Bäckstrand et al., 2005a) was conducted at the reference company with the purpose of evaluating if the workload from the information presented could be one of the sources that affected the internal reject rate. In that case study Cognitive Work Analysis (CWA) was used as a work method for analysing an assembly workstation. In short, the actual CWA was preceded by a 21-hour passive observation where the work on one of the assembly stations was analysed. There were two purposes:

1. To find indicators that showed if CWA could be used in the pre-design phase for an information system, not from a designer's perspective, but rather from a procurer's

perspective. That is, could it be possible to use CWA to evaluate design suggestions submitted via tenders.

2. To determine if it is possible to find and predict tasks, strategies etc. in an existing system and thereby find tasks and strategies that could be helpful for the personnel to adapt to, or which in a worst case could create quality problems.

The results from the study indicated that it is likely that CWA could be used for the purpose described above. This is obviously one of the reasons for the creation of a work method such as CWA. Sanderson (2003) discusses the connection between CWA and first-of-a-kind system. This is interesting because many of the systems in an assembly plant are of that type, first-of-a-kind or "the only"-of-a-kind. Sanderson presents two cases where CWA is used, one where the focus was on interface design and the other focused on systems evaluation. Both cases presented systems where the final product were highly advanced systems, a nuclear power plant and an air defence platform.

The examples above are meant to put the case study performed at the plant, in general and CWA in particular in a wider perspective. It seems that CWA is a method that is very useful and that CWA and similar methods are sometimes a necessity as an analysis tool during system design, where the final solution supports dangerous or complicated work. Compared to a nuclear power plant an assembly environment is rather uncomplicated and the consequences of a failure are far less critical.

One important conclusion from the study at the reference company was that it is extremely time consuming to perform a CWA.

The first part of the analysis was to gather data regarding the use of the existing system, this during 21-hour (This is not possible of course if the final product is a first-of-a-kind system). The next step was to use the syntax that CWA provides and by doing that, evaluate if the problems encountered could be found by using CWA.

The aim of CWA is to find situations and strategies for unanticipated events that the human operator may face, and to identify these situations/strategies in the interface design phase. This makes it possible to design information systems that support the operator better. One fundamental idea behind CWA is that constraints in the work environment and the cognitive competences of the operator shape work activities (Sanderson, 2003). But, it is also a method that is useful when discrepancies are observed between design intentions and actual use of a system (Rasmussen et al., 1990). This fundamental idea includes seven dimensions; "Work Domain, Task Space", "Activity

Analysis in Domain Terms", "Decision Analysis in Information Terms", "Information Processing Strategies", "Allocation of Decision Roles", "Management Structure", "Mental Resources, Competency, and Preferences of the Individual Actor" (Rasmussen et al., 1990). These dimensions form the five-phased framework of CWA. The five phases are (Sanderson, 2003; Fidel and Pejtersen, 2004):

- 1. Work Domain Analysis (WDA): Identification of physical and purposive (defines why technical system exists) constraints for the workplace where the activities take place, in this case an assembly station.
- 2. Control Task Analysis (CoTa): Defines what needs to be done within a work domain independently of how the task is to be achieved or by whom.
- 3. Strategies Analysis (SA): Identification of different strategies for accomplishing the task or work defined in the CoTa.
- 4. Social-organizational analysis (SOA): Identifies how the responsibility for the execution of different strategies is divided among humans and between humans and automation.
- 5. Worker Competencies analysis (WCA): Identifies the worker competence needed to accomplish the identified tasks.

There is no hierarchical or strict sequential order between the different dimensions. Therefore, the user of CWA does not need to follow a strict execution order when performing the CWA. Instead, the user can define a path and method suitable for a specific problem (Fidel and Pejtersen, 2004).

The data gathered should be mapped and used in the five phases presented above. In CWA this mapping is done with support from different diagrams, see figure 4.1 and 4.2 for examples.

WDA	, Assembly			Bilaga	
Total System		Sub System	Components		
Functional Purpose	Turbo montering				
Abstract Function		Montering			
Generalized Function		Artikel- montering	Turbo- montering Grenrör- montering Del- montering		
Physical Function			Motor Turbo Grenrör Bult Packning Skruv- dragare		
Physical Form			Artikel- nummer Borrhål Dimension Längd Form Hylsa		

Figure 4.1: Example of WDA. This example is taken from the case study and views the different components at the specific assembly station.



Figure 4.2: Example of flow chart that supports the mapping of CoTa.

These diagrams, flow charts etc. are very useful during the analysis work and without them the work of creating meaning from the vast amount of data gathered is difficult.

The problem with CWA is the time it takes to do the analysis work and create the documentation. The time it takes to gather all the data needed and to create the documentation is far too long when it comes to system updates.

4.3 EVALUATION NEEDS DURING PRODUCT LIFECYCLE

There is no doubt that it is of great importance that an analysis method is used during the design stages of a system as well as during the entire system lifecycle (Bias, 1994; Nielsen, 1994; Wharton et al., 1994; Sutcliffe, 1995). The problem is that an analysis method like CWA is far too advanced to be used by technicians in their daily work. As a method used to create a system, yes, to use it as a method to decide what to view on the interface used by the assembly worker, no. At this point, it might be appropriate to define what a change is. A change is not a change that affects the interface itself. In this case, it is a change in the final product in a way that affects the information presented on the computer screen. This change can be minimal from a product perspective, the product's shape is the same, but the characteristics are changed. This can trigger the creation of a new part number and often that is the only visual change made to the product. These changes are a natural part of a product evolution process. An existing information system has to be able to handle this kind of change in an effective way. An IT/IS-system has to be easy to update when this change occurs, but that is not of interest in this case, the important part is how the technician can understand in an easy, fast way how a change will affect the assembly worker and the system's ability to handle the changes.

The methods above all evaluate a work environment, an existing or a future one, from a human perspective where human factors is in focus. This makes the analysis work rather complex and time consuming and the results provide the analyst with a vast amount of information. These are the fundamentals that the new system is based on, and they exist and continue to exist during the lifetime of a system. The conclusion from the analysis is based on the knowledge of how humans act and react due to our strengths and limitations. These human characteristics will not change during the system's lifecycle, i.e. the results and findings that the initial analysis gave in the design phase of a support system will be valid during the whole system's lifecycle, if we assume that the initial analysis work was performed in a correct way.

If one assumes that the human has the same abilities and limitations during a system's lifecycle, then one can assume that a system update is not required due to changes in those abilities and therefore a new CWA analysis should be unnecessary. With this in mind, one has to ask, if the human is a constant, what is not? The assembly system includes some fundamental parts:

- The human.
- The information system.
- The product.
- The process. The process includes such things as:
 - The assembly process.
 - The material facades.
 - The product transport system, in this case a carrier.

The human could be seen as a constant. The information system is based on the product and process demands. That is, if there are changes to the product or in the process, this can create a demand for an update of the information system. The information system itself can go through changes due to for example IT-platform changes, but these changes are connected to the IT-developer and not to the production technicians.

The product and process changes that affect the information system are the ones of interest in this case. The product changes constantly, from small changes that do not affect the assembler in any way, to larger changes in the work environment that can demand training activities, new user interfaces, new material racks etc. These changes, initiated by a Design Change Note (DCN), see Section 2.2.3, have increased in recent years and the management have the opinion that these changes will be at least constant in number, and probably increase in the future. A change in the product can also be initiated by new product introductions, such as product introductions in new markets, new in brand (AB Volvo companies) customers, changes in the environment legislations etc. Continuous improvement work (Kaizen), see for example Berger (1997) and Doolen (2008) can affect both the product (if a DCN is created) and process, and by doing this, the information needed in the work environment can change. It is possible to indentify two major "stakeholders" that can trigger an information need: *product* and *process*.

These stakeholders and their needs/demands can be described as follows:

Product

The product can create a demand on the information system due to the need for an assembler to choose the right parts that should be assembled. If there is only one part variant to assembly at a specific time, the need originates from the process rather than the product. If a number of part variants exist, the need originates from the part itself but it still can affect the process. This need creates a *decision point*, i.e. a decision demand is

put upon the assembler where he or she has to decide what part to assembly on the specific engine.

Process

A variant dependent process can include different parts to assemble and/or include different amounts of parts to assemble, but it can also differ as to the order in which the part should be assembled. The later is less likely if a Design For Assembly (DFA) Dalgleish (2000), process is used to verify the assemblability of the product. This creates a *process support* information demand where the information presents how the part should/must be assembled and is different from the above where the information system must support the decision on what to assembly.

The keywords here are "what" and "how", what part to assembly and how to assemble the parts. This is the basic knowledge that a production technician has to possess to be able to perform evaluations regarding changes in the information need. The technician does not need to understand exactly what tasks or activity an assembly operation contains, i.e. he/she does not need to analyse exactly what activity an assembler needs to perform during an assembly task, he/she only needs to evaluate from a product perspective if a "what" and a "how" decision is triggered by the product change.

The product changes trigger an evaluation on how the changes affects the work environment from an information needs perspective. This evaluation process needs different methods depending on where in the product lifecycle the change takes place. A typical product industrialisation project process (Figure 4.3) at the reference assembly plant contains a number of steps; each of them proceeded by a gate. A number of issues have to be solved before a gate opens that starts the next step in the development process. Depending on project stage, the amount, form, and quality of the available data¹⁵ varies.

Pre Study Concept Study Developmen	Final Development	Industrial- sation	Deploy- ment	Follow-up)
---------------------------------------	----------------------	-----------------------	-----------------	-----------	---

Figure 4.3: Typical project flow at the reference assembly plant.

¹⁵ Data in this case are date that can be used to evaluate information needs. For example number of variants, assembly process etc.
This process is strongly connected to the project. After the project is finished, the production part of a product lifecycle starts. The responsibility for the product and its assembly process is transferred from the project, which no longer exists, to the production department. During the lifecycle, changes are triggered that create needs for assembly process updates. These triggers vary and can for example be quality problems, product updates, legislation changes, environmental issues, etc. They can influence the impact (amplitude) (Figure 4.4) of the amount of work needed for the development of a solution.



Figure 4.4: Visualisation of impact from changes during a product lifecycle, an example.

That is, in the industrialisation project, the available data can vary, during the product lifecycle and the impact and availability of data from a change can vary. These variations demand different evaluation methods as well as different mindsets. During the project stages the human must be in the centre of the evaluation. That is, the evaluation methods should focus on human abilities, such as cognition, human behaviour and psychology. During the production phase of the lifecycle, the evaluation methods should focus on the variable that goes through changes, i.e. the product.

The development process¹⁶ at the reference company includes the steps presented in Figure 4.3. The philosophy behind this method is that:

¹⁶ The description of the process is based on company reference material.

- It focuses on process and business needs.
- It describes what to do (activities) and when to do (phases and gates), but not how to do it.

Four key areas need to be managed by the process. They are:

- Promote issues to raise critical risks before they become a problem. Focus on prediction and prevention.
 - o Business objectives management.
 - Understand and agree on the problem definition based on a diagnosis.
 - Identify the business value and set business targets.
 - Develop the vision, scope and business case to be used throughout the project.
- Solution management
 - Develop processes that are in line with business objectives.
 - Provide the technology necessary to reach the business objectives.
 - Find a technical solution that matches the new or modified processes.
 - Perform verification and validation.
- Business change management.
 - Drive the organizational transformation needed, through communication, training and roll-out.
 - Transform business targets into matching business results by active involvement of all stakeholders.
 - Ensure that technical solutions, processes and organizational changes meet to form a complete solution.
- Project control.
 - Integrate above areas.
 - Balance quality, time, cost and content.

All the above must be taken into consideration. However, at the moment it is possible to neglect the issues connected to project management, and focus on aspects such as prediction and prevention, input to technical solution, process (for example assembly process) and training.

The development process does not describe how to analyse the risks etc. However, it gives support regarding what issues need to be solved before it is possible to enter the

next stage of the development process. The issues are an input to the gates. These gates (Figure 4.5) and their content all control the process by providing a steering committee with information so that it is possible to open the gate to the next process stage.



Figure 4.5: Development process including decision gates.

Gates description:

- Change Initiation Gate (CIG): approve the business value and formally start a prestudy.
- Vision Gate (VG): approve the project vision and the diagnosis.
- Concept Study Gate (**CSG**): decide which solution to investigate further. The gate marks the end of the pre-study phase and starts the project.
- Development Gate (**DG**): choose one solution, and approve its ways of working in combination with technical concept.
- Final Development Contract Gate (**FDCG**): freeze the overall solution and sign the contract.
- User Launch Gate (**ULG**): approve that the solution is ready for user validation tests.
- Release Gate (**RG**): approve that the solution is ready for deployment and the organisation is ready to receive it.
- End Gate (EG): approve that solution contents and deployment are achieved according to the contract, hand over the responsibility to the maintenance organisation, and close the project.
- Follow Up Report (**FUR**): validate that the business objectives have been achieved and, if needed, decide action plans and further change management activities.

In the **CIG** (Figure 4.5), a high level vision for the project is presented. At this stage of the process the focus is on pre-study business objectives and business change management, and project control. At this stage, the focus on solution management is at a

minimum. Between the CIG and the VG (Figure 4.5), the actual evaluation process, in this case called "Diagnosis", is conducted. Issues such as interviews, measurement of current system (if such exists), best practice review, SWOT analysis, white book review (lessons learned from previous projects), and presentation of possible improvement areas are conducted and reviewed. The output from the diagnosis is the needs identified. These needs can be for example improvement of quality of the information stored, to secure the information flow in the assembly lines etc. During the work between the VG and the CSG (Figure 4.5), an evaluation method must be used to ensure high-level verification of information demands on future systems. The output from Pre-study is a decision regarding one or more solutions that will be further detailed in the next stage of the development process, the concept study phase. Before opening of DG (Figure 4.5), a decision has to be made regarding one solution that will be further developed in the Development Phase. In the Development Phase of the project the focus is on development of details necessary to freeze the solution. FDCG (Figure 4.5) opens the Final Development phase, which aims to develop the technical solution and prepare for deployment. Between FDCG and ULG (Figure 4.5) the project is in its final development phase. The major development is done during this phase. The confirmation of deployment and validation test plans is also an important part of this phase. The industrialisation phase with its RG (Figure 4.5) includes validation of tests performed on the processes and system/-s. The results from them influence the approvals of the deployment plans. EG (Figure 4.5) is open if there is approval regarding the functionality of the process and system and that they are fully operational. During the Follow-up phase ending with FUR (Figure 4.5) focus is on confirmation that the process and systems are working and delivering expected results.

This was a brief description of the phases and gates included in the development process. Absent in this description is business values. This is always the main driver, at a high level, for a development project. However, in this case, the interesting part is the analysis of information need, and therefore the business values are excluded from this discussion, even though they are related through information need influencing the product quality and hence the overall business objectives.

It is possible to visualise the evaluation needs and its connection to available data during the development project (Figure 4.6).



Figure 4.6: Visualisation of a relationship during the development process, between the available data, abstraction levels, and evaluation needs.

The data available (small amount), the abstraction levels (high abstraction), and the demand during the initial project phases have an impact on the evaluation that will be performed at these stages. However, the evaluation demands increase significantly during the later part of the pre-study phase to the development phase. The evaluation must be finished in the development phase so that it is possible to open FDCG. The need for an evaluation should be significantly lower during the later part of the process, but a need for evaluations can again be higher in the follow-up phase if problems occur. The available data will increase during the project, and will continue to increase during the product lifecycle.

One of the problems connected to development, product as well as process, is that it is extremely important to do evaluations as early as possible in the project. There are problems connected with this need:

- 1. As visualised in Figure 4.6, in the beginning of the project the available data is at an abstraction level and at an amount that can make it difficult to do in-detail evaluations.
- 2. The available data, amount and abstraction level, will increase in later stages of the process, with its peak in the development phase.

That is, in the beginning of the process, the lack of data lowers the possibility of influencing the final product. In later stages of the process it is possible to do in-detail analysis, but the possibility of influencing the final product decreases due to time limits for the project.

The evaluation work during these early stages should focus on human abilities and limitations and their impact on the process. The work during the later part of the product lifecycle should focus on demands originated from the product.

4.4 PROPOSED WORK METHOD

A work method that spans a product lifecycle should emphasis different aspects depending on where in the lifecycle an analysis is conducted. It is of great importance that it is simple to use and gives the production technicians the information needed to chose a suitable user interface from productivity as well as quality perspectives.

It is possible to divide the lifecycle into these major stages (Ryan and Riggs, 1997):

- Introduction.
- Growth.
- Maturity.
- Decline.

From a manufacturing point-of-view the project phase, i.e. the development project should be included in the product lifecycle.

In this case, it is possible to divide the PLC into two major parts:

- Development phase: During this part the project flow presented in Figure 4.3 should be followed and the focus should be on humans.
- Production phase: During this part should focus be on the product and its demands.

Development phase

The work process at this stage is relatively straightforward and should use one of the existing evaluation work methods, for example Cognitive Work Analysis. When, how and to what extent must be determined by company procedures. In the example presented in Section 4.3 the evaluation work must be done before the Final Development Contract Gate (Figure 4.5).

Production phase

The analysis work during the production stage of the PLC is triggered by different events. Events can be:

- Quality issues.
- Productivity demands.
- New product.
- New product variant.

The quality and productivity issues are often connected to the introduction of new products and product variants.

The proposed evaluation process in the production stage is much more simplified in comparison with the one used during the development work.

The proposed process includes steps according the workflow shown in Figure 4.7.



Figure 4.7: Proposed work process/method for the Production Phase.

Process steps:

- Start evaluation process: triggered by, for example, a design change note (DCN) (Section 2.2.3).
- New variant: evaluate if a new variant is introduced. More than one variant of a part creates a need for decision support.
 - Choose user interface: in this proposed work process a number of interfaces have been identified. Some of them are used at the plant in other applications:
 - Paper order form. The paper contains order number and the amount. An instruction form that "translates" the order number to part numbers is found at the assembly station and includes part numbers and assembly

instructions. Often used in pre-assembly stations without computer support.

	Orde	er id:	99	258542				VOLI	/O IT, 1	SKÖV
Header				No.	Instructi	on text	Bcode	Continu	Tight.	QυT
364025				1			\checkmark		\checkmark	хЗ
946671 SCREW M8*30				3						
20411113 CONSOLE				1					0	x2
1547252 SPACER				1						
1546531 LOCKING SHIN	1			1						
977977 CLIP				1						
946440 SCREW M8*16				2						
STATION COMPLETED								\bigcirc		
STATION COMPLETED :										
F2) Status CTRL (F3) Send Res. (F4)	Man.Input	an.Isput (F5) Re Do (F6) Man.Corr. (F7) Reject (F8) Msgs (F9) OK (F10) Station							list	

• Interface presenting part number and assembly order (Figure 4.8).

Figure 4.8: Interface with part number and assembly order.

Pick lights (pick-by-light). A light (Figure 4.9) indicates what part to assemble. It is possible to add functionality that makes it possible to verify that the part has been picked. This can be done via a photo cell that indicates that the assembler has picked the part. It is possible to use pick lights in at least two ways: 1, parts are identified one by one and the picking is registered by the system. 2, all parts are identified simultaneously by the system and the assembler can choose which part to pick first.



Figure 4.9: Light indicating what part to pick next. This specific installation includes a photocell that when affected, registers when a part is picked.

 Colour Coding. A colour marking (Section 3.2 and Figure 4.10) is used to guide the assembler to pick the right part.



Figure 4.10: Example of colour coding in an assembly environment.

- Voice picking. The assembler uses a headphone where he/she receives a number that is connected to a location in the material racks.
- Sequencing and kitting. These techniques are connected to logistics and how the material is presented to the assembler at the assembly station. In sequencing (Figure 4.11) the part is presented to the assembler so that he or she does not need to decide which part should be assembled next.
 Figure 4.12 illustrates a kit where parts are gathered and presented to the assembler all at once.



Figure 4.11: Sequencing example. In this case is it water pumps prepared for assembly.



Figure 4.12: Kit presented at the assembly station. When the parts in the box are assembled the box is removed and a new box is presented to the assembler.

- Create training material. The training material should be based on the information presented in the chosen interface.
- Define information. The information needed is defined and prepared for the chosen interface.
- Influence process. If the change in the production influences the assembly process, an update is required for the process information.
 - Define process. Define new assembly process.
 - Choose user interface: in this proposed work process a number of interfaces have been identified. Some of them are used at the plant in other applications.
 - Paper instructions. The process is presented in a paper instruction available at the assembly station.
 - Video recording. A video recording is made that shows the assembly task. (This has to be done after the new changes have been made in the assembly environment). The recording is presented on a computer screen at the assembly station.
 - 3D-recording. A 3D-recording is made that shows the assembly task.
 The recording is presented on a computer screen at the assembly station.

- Pictures. Pictures are present on paper or on a computer screen. The pictures show the critical parts of an assembly task.
- Create training material. The training material should be based on the information presented in the chosen interface.
- Define information. The information needed is defined and prepared for the chosen interface.

One of the problems a production technician will face during the evaluation is to understand the consequences of their choice of interface. The interfaces presented earlier in this section all have their pros and cons. Although tests have not been performed on them all, some characteristics can be identified for each of them. These characteristics must be taken into consideration during interface choice.

The characteristics for:

Paper instructions

The benefit with paper instructions is obvious in that there is no need for a computer in the work context. This makes it possible to quickly distribute information and set up new assembly stations if needed. The drawbacks are that it is almost impossible to determine if the instruction, in this case, is the latest review or not and it needs to be physically changed when an update is required. It has also been noted that the assembly personnel do not use them in an effective way, and sometimes not at all. One problem with paper is where to put it, in an assembly context that is. It is also difficult for an assembler to understand when to use paper instructions.

Video recordings

There are different ways to use video recordings. At the reference plant recordings are used in the rebalancing¹⁷ work for the assembly stations. A video recording is made that can be used in conjunction with software to find and define wastes, necessary activities etc. This recording can also be used during the introduction of new personnel. It is possible to use the recordings as information and process support directly in the assembly context. The major drawback is that it is not possible to create a recording before the assembly context and the product physically exists in the assembly context.

¹⁷ Balancing an assembly station is to make sure that the assembly time for all stations in an assembly line has the same station time. This is, in short, managed by moving parts and tasks between the assembly stations. A rebalancing task is initiated for example when a new engine variant is introduced.

Consequently, it is not possible to create e.g. training material and therefore is it not possible to start the training at a stage where there is no effect on the production.

3D-recordings

A 3D-recording, based on video or pictures, is created in a computer environment where the basis for it is the CAD-data showing the product and equipment. The benefit of using 3D-material is that it is possible to create instructions, training material etc. earlier than is possible when the information is based on physical objects. Consequently, it is possible to prepare an assembly station with the instructions needed before start of production. It is also possible to start training activities earlier compared with training on physical objects. The major drawback is the need for deep computer knowledge in the field of virtual manufacturing. It is both time consuming and skill demanding to create the virtual environment needed to create the information. However, it is possible to see instruction and training material as a spinoff benefit from other analysis needed in the development phase of new assembly stations. That is, it is easy to create the material if the virtual environment already exists. There is a need for early ergonomic evaluations and a decrease in physical mock-ups (Bäckstrand et al., 2005b; Bäckstrand et al., 2006b; Bäckstrand et al., 2007; Högberg et al., 2007; Bäckstrand et al., 2008; Högberg et al., 2008). The output from that analysis, evaluation data and virtual environment is very suitable as a foundation for instructions, training material and process support information.

Pictures

Pictures of specific task or assembly related details are a good and simple way of distributing information in an assembly context. It is easy to use a camera, most people can handle a camera after a short introduction. Photographs are easy to distribute, both on paper and via a computer. The major drawback is the same as for video recordings in that it is not possible to take a photo without physical objects.

Interface presenting part number and assembly order

A graphical user interface is an effective way of distributing information and creating knowledge. Computer support makes it possible to distribute information at the right time and place. It is possible to trace where and when the information is viewed, and this is important regarding traceability e.g. in quality work. The major drawback is cost. The cost for the development or purchase of software, license cost, hardware, installation, maintenance etc. can be very high and sometimes hard to estimate in early project stages. Compared with paper instructions the payback time can be long.

Pick lights (pick-by-light)

Pick lights can be an effective way to influence productivity in a positive way. It is easy to understand, just pick the part where the light is lit. It can be easy to view the assembly order, and it is possible at the same time to view the amount of parts to pick. Compared with the other interfaces a pick light system is the least demanding together with Colour Coding, on the short-term-memory as well as on attention levels. The major drawback is that it is rather inflexible. If the assembly environment is changed the pick light system might need to be reprogrammed and moved etc. It also needs qualified personnel that can maintain the system.

Colour coding

There are many resemblances between Colour coding and pick lights. Colour coding is one way of viewing decision point information without using computer support, although it can be more efficient using computers. The benefit of using colours, or symbols, is that it makes it possible for the assembler to find the parts needed with less attention compared with using systems where part numbers is the information carrier. It is inexpensive but needs to be attended to so that no mistakes are made, e.g. the wrong colour on a box. This problem is not unique to Colour coding in any way, the same problem can also occur in a computer supported system. The major drawback is that a believed upper limit for the number of coloured fields used at an assembly station is four, after that a pick light is preferred. This has not been tested, but if the number of fields exceeds four, the problem of keeping track of where in the process the assembly takes place increases. Case study 2 (Section 3.3) indicates that this is the case.

Voice picking

Like Pick lights and Colour coding the benefit with Voice picking is that the user does not need to know what to assembly next. A position in space is presented to the user via a headset. The part to assembly next can be found at this location as well as a code. This code is read by the assembler and registered by a microphone and compared with a code stored in a computer. If the assembler reads the right code the system continues to present the next part to assemble. If the assembler reads the wrong code, he/she receives an error message and is urged to redo the task. The benefit is that it is possible to distribute information to locations where no computer exists, although a wireless network is needed. It is easy to use, no specific training is needed in the use of information. A drawback is that, according to initial tests at the company, the user feels isolated from their colleagues etc. It is too early to say how this affects human in the assembly context but if further tests initiated this is one of the issues.

Sequencing and kitting

Sequencing and kitting have been described earlier in this text, but it is possible to add that both sequencing and kitting lower the workload for the assembly personnel. However, an important issue is that the burden on the preparation personnel increases as well as their information need. The preparation personnel need the same information that the assembly personnel needs. Therefore, the techniques presented earlier, e.g. pick lights, although the assembly personnel have no use for them in this case, are needed by the preparation personnel.

4.4.1 Prototype User Interface

In order to test the exemplified work method, a prototype user interface was designed where the work method worked as a foundation for the design. The first view (Figure 4.13), the assembly view, is the main part the interface where the assembler can find decision support information about what to assembly next. The three (coloured) fields, "Rör Nedre", "VCB-Mellandel" and "Servopump" are where the assembler can find two types of information:

- In what order the parts should be assembled, reading from left to right.
- What part variant should be assembled.



Figure 4.13: Assembly view.

A feature that was appreciated by the personnel during Case study 1 was the possibility to see what engine variant was approaching the assembly station. This feature is handled in the prototype by the field "Pågående kö" (current queue), located in the upper part of the interface (Figure 4.13). This field contains information regarding what type of engine variant approaching the station. In this specific case it is the view seen at station S0800 that is shown, the stations viewed in the approaching field are, from right to left, station S0700 to S0300. The interface is updated each time a carrier is moved to the next assembly station.

If this design was "a real case" this functionality should have been evaluated in the development phase where the humans' abilities are in focus. The three (coloured) fields however, should be evaluated by the technician in the production phase, where he or she should determine if:

- 1. More than one variant exists.
- 2. Number of different variants. This determines the number of different colours.
- 3. Number of decision points. There are three decision points in the prototype. This is evaluated and defined by using the work method described in Section 4.4.

However, the interface described above only handles decision points. In Figure 4.14, the interface is updated with a process flow description, a view that shows the production status, and a control instruction view that shows how a control process should be performed.



Figure 4.14: Updated interface including for example process support information.

The main issue is that the user should be triggered by an event. This could be the signal to the user that a new variant has arrived at the station, but it could also signal to the user that, for example a change has been made in the assembly process. Figure 4.15 illustrates how a change in a control instruction can be viewed in the prototype interface.



Figure 4.15: Illustration of change control instruction event.

In a real case scenario the updated control instruction is put into focus automatically. It is not possible for the user to continue to assemble the engine until he/she has verified that they have observed and read the new instruction (done by changing the interface back to the assembly view, see Figure 4.14).

4.5 CONCLUSIONS

The conclusion regarding the proposed work method is that it is important to use different ways of working depending on where in the product lifecycle an evaluation is needed. There is more than one reason for this.

- Time: A work process that is too time consuming is not an alternative. Therefore, it is important to have a balance between the output from an analysis, in the form of knowledge, and the time it takes to conduct it.
- Competence: It is not possible to have cognitive specialists working in the later part of the PLC. The cost for this will be too high. However, the need for having a specialist to consult still exists at this stage of the process.

• Simplicity: A work process must be easy enough to use so that a non-specialist can use it.

It is possible put considerable effort into the decision-making process regarding what method is the best to use in the development phase of the product lifecycle. This research never had the aim of evaluating different work methods used in the development phase. However, there is no doubt that an evaluation of information needs in the development phase is a necessity. This will create a demand to find "the best" evaluation method. This demand could trigger a decision-making process regarding what evaluation method should be chosen. As mentioned earlier (see e.g. Section 4.2) CWA has been used during this research on at least one occasion. During its use it has become clear that the method is very time consuming to use, but will return a vast amount of data. Time is relative. In this case the use is too time consuming for the daily work by the production technicians at the assembly plant, but, as an evaluation method in earlier stages of the PLC the method is very interesting. Characteristics of CWA are:

- Easy to use (this can seem as a contradiction to earlier statements made in the thesis), for a person experienced in the use of CWA.
- Documentation is created at the same time as the evaluation of the data is made.
- Assembly process: the documentation can be used as a foundation for the assembly process made in the IS-GPD, before the Release Gate (RG) (Section 4.3 and Figure 4.5). For example the Work Domain Analysis (WDA) (Figure 4.1) and the Control Task Analysis (CoTa) (Figure 4.2) include information that the production technician would need to gather or create before RG. If CWA is used, some of the work has already been done and in an earlier stage of the development process.
- Easy to audit the assembly personnel regarding the assembly process. That is, do they follow the proscribed assembly process or not. This is an important part of the quality work as well as the work regarding productivity.

The conclusion is that at the right time in the product lifecycle CWA is a method that is well suited to providing a project with information regarding information needs in an assembly line.

The derived work method has been applied in the design of the prototype user interface, but has not yet been applied in the daily assembly preparation work at the

reference company. The interface has been presented to and accepted by the assembly personnel, but further testing and implementation has to be performed before an implementation is a fact.

CHAPTER 5 DISCUSSION

This chapter contains the final discussion of this research and summarises some of the findings. It also presents and discusses the findings related to the two hypotheses presented in Chapter 1.

5.1 GENERAL REFLECTIONS

The research shows that just information is not enough to prevent quality problems from occurring and that one of the issues connected to the specific quality problem is attention. The lack of attention at a decision point is a major contributor to the quality problems registered at the reference company. A higher level issue briefly discussed in Section 1.2.1 was that there are potential problems if the personnel use information sources other than the ones that are a part of the general information system. It has been obvious during the work that in some cases the personnel use the product as an information source. This could lead to a quality problem due to the connection between this and the prognostication¹⁸ that the personnel sometime use of what products might arrive next at the assembly station. It is also very difficult to see the difference between parts, and for example in the case of the turbo sometimes only the part number can identify a difference. Although this is connected to the information presentation timing (Section 3.2.4, 3.4 and 4.4.1) it is still interesting and important to understand why a person chose to use one source in preference to another. This issue has not been discussed in depth in the thesis but is still a major issue that should be addressed in further work.

It is important to state that the hypothesis is mainly a result from the research and that the validation of the two and the evaluation method will be addressed in future work.

The following two sections aim to outline what conclusions can be drawn from the research in respect to Hypothesis 1 and 2.

¹⁸ Prognosis: a forecast of the probable development of (something) Hornby, A. S. (1997). Oxford Advanced Learner's Dictionary of Current English. In Crowther, J. (ed.). Oxford: Oxford University Press. In this case is it a forecast regarding what engine is to arrive.

5.2 HYPOTHESIS 1

According to Hypothesis 1 - 'Information seeking behaviour'

The degree to which Active Information Seeking behaviour is supported/triggered has a large influence on the number of internal rejects.

As noted in Chapter 1, this hypothesis is based on the fact that in most of the cases the assembly personnel have enough information present in their working environment to be able to perform a correct task. Therefore, it is more likely that human behaviour in general, and Information Seeking Behaviour specifically, influence the internal reject rate, and not the amount of data or information.

By presenting information in a way that supports active attention, an Active Information Seeking process can take place that positively affects the internal reject rate (Section 3.2.4). The information seeking behaviour model first presented in Section 1.2.2 and later elaborated in Section 2.5 is by nature connected to Hypothesis 1.

The main issue as mentioned earlier in the thesis is the effect attention has on the information search process. There is no doubt that attention is an important factor in the assembly environment, both case studies have showed this. The interesting work actually starts after the last page of this thesis is written. That is, this thesis presents an explanation model that states that attention is important for the outcome of an assembly task, therefore it is the only natural step to put this explanation model into practice.

5.3 HYPOTHESIS 2

According to Hypothesis 2 - 'Use of evaluation methods'

The use of an evaluation work method in the design phase of an information flow will affect the internal reject rate in a positive way.

This hypothesis is based on the assumption that the assembly personnel always strive for a performance outcome that lives up to the quality demands and targets. A common opinion among management and technicians is that the amount and the relevance of the data/information present in the assembly environment is enough to perform a correct task. This indicates that there is a deviation between the need and demand connected to a

specific task and the believed need/demand from the designers' point-of-view. The hypothesis states that a method used in the design phase is important and can positively affect the internal reject rate. The conclusion from this research is that evaluation methods are of great importance during a product's entire lifecycle rather than just during the design phase.

A company can only survive if it makes money. Therefore is it very important that every activity tries in some way to lower the cost, short term or long term. A conclusion from this research is that an evaluation method is needed in almost all of the stages in the product life cycle, and as a result money will be saved. One interesting, but missing parameter that all of the evaluation methods studied, briefly or more in depth, is cost. It is not that easy to estimate cost for a solution based on results from the evaluation. The parameter cost and the "equation" benefit/cost = savings ratio will be an important part of the decision making process regarding how a user face should be design and used.

CHAPTER 6 CONCLUSIONS AND FURTHER WORK

This chapter presents a brief summary of the research work and the relevance of the results and suggestions in this thesis, and finally, recommendations for further work.

6.1 THESIS SUMMARY

This thesis focuses on the connections between the use of information and quality in a manual assembly context. The main motivation for this research was the desire to understand why the existence of information itself was no guarantee of flawless production. As a consequence of this, the desire was to understand if a way to prevent the occurrence of these quality problems existed and if so, was it possible to incorporate this in the daily work as well as in the development of an information system for industry use. The main objectives were to understand how attention affected the internal reject rate, and to investigate and propose a suitable evaluation method during the design stage of an information system. During the research the knowledge gained regarding attention, evaluation methods and information use have made it clear that attention, active, passive goal-driven, stimulus driven etc. is an extremely important part of the use of an information system. The literature review confirmed the importance of active attention initiated by events in our surroundings as a driver for a successful use of information in an assembly context. This review contributed largely to the design of the case studies performed. During these case studies different information providers were evaluated from a quality as well as a productivity perspective. The results from the studies revealed the importance of the presentation of the information, where the information itself triggers the right search process. The way the information was presented during the studies showed that it is possible to influence quality as well as productivity by addressing the information in the right way and at the right time. Furthermore, the research has shown that it is possible to use information effectively so that unnecessary waiting, due to discreet information flows, is eliminated in the assembly line. It is argued that by using an evaluation method in the design stage of an information system these types of problems could be avoided and eliminated before they occur. The research has brought to light some of the problems connected to existing evaluation methods that focus on cognitive analysis issues. These methods are exceptional regarding the

knowledge output gained from the use of the same. However, it has become clear that these evaluation methods have drawbacks that make it difficult to use by other than experts in cognitive analysis. This influences the use in later parts of the product lifecycle. Therefore, a method is suggested that makes it possible for non-experts to evaluate assembly environments and to find problems related to cognitive issues connected to decision support and process support.

6.2 CONTRIBUTION

The overall research objectives were to:

- Understand how information affects the internal reject rate (Section 1.3.2).
- Understand how the assembly personnel interact with the current information system (Section 1.3.2).
- Investigate and propose evaluation methods suitable for work place design at different phases of the product lifecycle (Section 1.3.2).

These objectives are realised through the following contributions:

- Knowledge gained regarding how attention, human errors and situation awareness influence the assembly worker during the information search process and their assembly tasks, and how this is connected to internal reject rates (Section 2.4).
- A proposed work method that is based on the case studies performed. It also takes into consideration the feedback gained from the interviews and survey that were conducted among the assembly personnel (Section 3.2.4). The method is based on the conclusion that the user needs support regarding the decision point (*decision support*) and assembly process (*process support*).
- A user interface prototype was created with help from the evaluation method. The interface is designed based on three principles: 1. identify decision points, 2. present the information at the right time, 3. connect the information to training. That is, use the same material for viewing the assembly process in the assembly line and to train the personnel (Section 4.4.1).
- Finally, the research has contributed to knowledge regarding the importance of presenting the information at the right time. And this from both a quality perspective as well as from a human workload perspective (Section 3.2).

6.3 CONSIDERATIONS FOR FURTHER WORK

This research ranged over an area including cognitive science, ergonomics, psychology, human behaviour, engineering and in some sense computer science. The scope for further work is wide, but it should focus on a more integrated perspective within the field of Human Factors. During this research some avenues for further research and future development were identified. It is recommended that new studies and future development are addressed to:

- The connection between information systems and training needs. Case study 2 showed that there could be a connection in how the information is presented and the learning curve. This might not come as a surprise, but it is important that if this is the case, it should be taken into consideration during the design of an information system.
- Put the explanation model into practice, i.e. to test and evaluate user interfaces for decision support, for example pick-lights, voice picking, kitting, sequencing etc. to determine when to use a particular interface.
- Test and evaluate process information user interfaces. During this research the need for an effective way to provide process information was identified. A system with connections to training was evaluated and the conclusions were that this system and the information provided from the system could be used as a process information provider within the assembly line.
- Information sources. Why do the assembly personnel use information sources that are not part of the information system, in this case the product? And is that a quality risk? Is it possible to use this as an information source and thereby incorporate the product in the information system?
- Feedback to product design. The need for information and instructions in an assembly plant must always be seen as a waste, and it is a cost to provide it, not a saving. With this as a fundamental way of looking at information, it is of great importance that the work does not end with the design of an information system. The important work with eliminating this need starts and it should start with immediate feedback to the production designers that are responsible. This requires a work process that continues to evaluate and give continuous feedback to product design so that the process of eliminating information needs can start. The work with this process will start in the near future at the reference company.

• Framework for user interface design connected to *decision support* and *process support* needs. Different needs should create different demands on user interface design, presentation of information and on what information the assembler needs. It is therefore important to investigate further in detail how information should be presented and what type of information the assembler needs to handle situations connected to decision support and process support.

6.4 CONCLUSIONS

This research was initiated and motivated by industry demands regarding the need for information and its influence on quality as well as on humans, in this case the assembly personnel. It has been shown that:

- There is a strong connection between attention and the internal reject rates, where a system which demands an active attention mode is more vulnerable to quality and productivity issues.
- An information system must provide information such that the user understands that there are changes in their work task. A trigger that changes the attention mode should be present. If the trigger is absent or neglected there is an increased risk of quality problems.
- An information system (as presented in Section 3.2) where the focus for the presented information is on decision support can affect the internal reject rates in a positive way.
- The timing of presentation of information to the user is of importance in determining how the assembler acts within the assembly process. This directly affects the risks to quality in the assembly context.
- The way in which information is structured and viewed has a large influence on productivity.
- A connection has been established between productivity and the number of variants. This is not a consequence of product complexity, but instead is a consequence of the information seeking process.
- Design flows can be found by using an evaluation method during the design of an information system.

• An evaluation work process that is too time consuming or too complex is not an alternative during day to day work. The process for evaluation demands on existing information systems must be manageable by the manufacturing technician.

REFERENCES

Alter, S. (1999). *Information Systems: A Management perspective*. Addison-Wesley Educational Publishers Inc.

Andersen, E. S. & Schwencke, E. (2002). *Projektarbete-en vägledning för studenter (in Swedish)*. Lund: Studentlitteratur.

Axelsson, J. R. (2000). *Quality and Ergonomics - Towards successful integration*. University of Linköping. Linköping. Department of Mechanical Engineering. Doctorial Thesis.

Badre, D. & Wagner, A. D. (2007). Left ventrolateral prefrontal cortex and the cognition control of memory. *Neuropsychologia*, 45, 2883-2901.

Beach, L. R. & Connelly, T. (2005). *The Psychology of Decision Making. People in Organizations*. Thousand Oaks: SAGA Publications.

Beach, R., Muhlemann, A. P., Price, D. H. R., Paterson, A. & Sharp, J. A. (2001). The role of qualitative methods in production management research. *International Journal of Production Economics*, 74 (1-3), 201-212.

Bellorini, A. & Decortis, F. (1995). A stress-based analysis in Air Traffic Control. In *Proceedings of the 21st Conference of the European Association for Aviation Psychology (EAAP)*, Dublin. pp. 95-101.

Berger, A. (1997). Continuous improvement and kaizen: standardization and organizational designs. *Integrated Manufacturing Systems*, 8 (2), 110-117.

Besnard, D. & Greathead, D. (2003). A cognitive approach to safe violations. *Cognition, Technology & Work*, 5, 272-282.

Bias, R. G. (1994). The Pluralistic Usability Walkthrough: Coordinated Empathis. In Mack, L. R. & Nielsen, J. (eds.). *Usability Inspection Methods*. New York: John Wiley & Sons, Inc.

Blasch, P. E. & Plano, S. (2002). JDL Level 5 Fusion Model "User Refinement" Issues and Applications in Group Tracking. In *SPIE Aerosense 2002*.

Brandt, D. S. & Uden, L. (2003). Insight into mental models of novice Internet searchers. *Association for Computing Machinery. Communications of the ACM*, 46 (7), 133-133.

Bratthall, L. & Jörgensen, M. (2002). Can you Trust a Single Data Source Exploratory Software Engineering Case Study? *Empirical Software Engineering*, 7 (1), 9-26.

Breakwell, G. M. (2002). Interviewing. In Breakwell, G. M., Hammond, S. & Fife-Schaw, C. (eds.). *Research Methods in Psychology*. pp. 239-250. London: SAGE publications LTD.

Brewster, S. A. (1997). Using non-speech sound to overcome information overload. *Displays*, 17 (3-4), 179-189.

Byström, K., Murtonen, K. & Järvelin, K. (1995). Task Complexity Affects Information Seeking and Use. *Information Processing & Management* 31 (2), 191-213.

Bäckstrand, G., De Vin, L., Högberg, D. & Case, K. (2005a). Parameters affecting quality in manual assembly of engines. In *Proceedings of the 2nd International Manufacturing Conference: Challenges Facing Manufacturing*, Dublin. pp. 395-402.

Bäckstrand, G., De Vin, L. J., Högberg, D. & Case, K. (2006a). Attention, interpreting, decision-making and acting in manual assembly. In *Proceedings of the Innovations in Manufacturing, 23rd International Manufacturing Conference*, Belfast. pp. 165-172.

Bäckstrand, G., Högberg, D., De Vin, L. J., Case, K. & Piamonte, P. (2006b). Ergonomic Analysis in a Virtual Environment. In *Proceedings of the 23rd International Manufacturing Conf.*, University of Ulster, Jordanstown, Northern Ireland. pp. 165-172.

Bäckstrand, G., Lämkull, D., Högberg, D., De Vin, L. J. & Case, K. (2007). Reduction of Design Flaws Through Virtual Methods. *International Journal of Manufacturing Research*, 2 (2), 198-208.

Bäckstrand, G., Möller, S., Högberg, D., De Vin, L. J., Sundin, A. & Case, K. (2005b). A Roadmap towards Cost Calculation Methods Connected to Ergonomics Analysis and Simulation. In *37th Ann. Conf. of the Nordic Ergonomics Society (NES)*, Oslo, Norway. pp. 312-316.

Bäckstrand, G., Thorvald, P., De Vin, L. J., Högberg, D. & Case, K. (2008). Information impact on work environment and product quality: a case study. In *Proceedings of the Nordic Ergonomics Society Conference (NES)*, Iceland.

Cacioppo, J. T. & Petty, R. E. (1982). The Need for Cognition. *Journal of Personality* and Social Psychology, 42 (1), 116-131.

Coull, J. T., Frith, C. D., Buchel, C. & Nobre, A. C. (2000). Oriented attention in time: behavioural and neuroanatomical distiction between exogenousand endogenous shifts. *Neuropsychologia*, 38, 808-819.

Coull, J. T. & Nobre, A. C. (1998). Where and When to Pay Attention: The Neural Systems for Directing Attention to Spatial Locations and to Time Intervals as Revealed by Both PET ans fMRI. *The Journal of Neuroscience*, 18 (18), 7426-7435.

Crawford, L. E. & Cacioppo, J. T. (2002). Learning Where to Look for Danger: Integrating Affective and Spatial Information. *Psychological Science*, 13 (5), 449-454.

D'Souza, M. E. & Greenstein, J. S. (2003). Listening to users in a manufacturing organization: a context-based approach to the development of a computer-supported

collaborative work system. *International Journal of Industrial Ergonomics*, 32 (4), 251-264.

Dalgleish G.F, Jared G.E.M & Swift K.G. (2000). Design for assembly: influencing the design process. *Journal of Engineering Design*, 11 (1), 17-29.

Dantes. (2008). *The Dantes Project*. <<u><htp://www.dantes.info/Projectinformation/Glossary/Glossary.html>></u>. [2008-01-22.

Decrop, A. (1999). Triangulation in Qualitative Tourism Research. *Tourism Management*, 20, 157-161.

Dencker, B., Balzer, H. J., Theuerkauf, W. E. & Schweres, M. (1999). Using a production-integrated video learning system (PVL) in the assembly sector of the car manufacturing industry. *International Journal of Industrial Ergonomics*, 23 (5-6), 525-537.

Denzin, N. K. & Lincoln, Y. S. (2000). The Discipline and Practice of Qualitative Research. In Denzin, N. K. & Lincoln, Y. S. (eds.). *Handbook of Qualitative Research*. Thousand Oaks: SAGE Publications Inc.

Dix, A., Ramduny, D. & Wilkinson, J. (1998). Interaction in the Large. *Interacting with Computers*, 11, 9-32.

Doolen, T. L., Aken, E. M. V., Farris, J. A., Worley, J. M. & Huwe, J. (2008). Kaizen events and organizational performance: a field study. *International Journal of Productivity and Performance Management*, 57 (8), 637-658.

Downing, P. E. (2000). Interaction between visual working menory and selective attention. *Psychological Science*, 11 (6), 467-473.

Drury, C. G. (2000). Human factors and quality: Integration and new directions. *Human Factors and Ergonomics in Manufacturing*, 10, 45-59.

Dubischar, D., Gundlach, V. M., Steinkamp, O. & Khrennikov, A. (1999). A p -Adic Model for the Process of Thinking Disturbed by Physiological and Information Noise. *Journal of Theoretical Biology*, 197 (4), 451-467.

Duncan, J. (1995). Selection of inputs and goals in the control of behaviour. In Baddeley, A. & Weiskrantz, L. (eds.). *Attention. Selection, Awareness, & Control.* Oxford: Oxford University Press.

Edmunds, A. & Morris, A. (2000). The problem of information overload in business organisations: A review of the literature. *International Journal of Information Management*, 20 (1), 17-28.

Eklund, J. (1997). Ergonomics, quality and continuous improvement--conceptual and empirical relationships in an industrial context. *Ergonomics*, 40 (10), 982-1002.

Ellis, D. (1989). A behavioural approch to information retrival design. *Journal of Documentation*, 45 (3), 171-212.

Ellis, D., Cox, D. & Hall, K. (1993). A Comparison of the Information Seeking Patterns of Researchers in Physical and Social Sciences. *Journal of Documentation*, 49 (4), 356-369.

Endsley, M. R. (1996). Automation and Situation Awareness. In Parasuraman, R. & Mouloua, M. (eds.). *Automation and Human Performance: Theory and Applications*. pp. 163-181. New Jersey: Lawrence Erlbaum.

Endsley, M. R. (1999). Situation Awareness and Human Error: Designing to Support Human Performance. In *Proceedings of the High Consequence Systems Surety Conference* Albuquerque, NM.

Endsley, M. R. (2000a). Situation Models: An Avenue to the Modeling of Mental Models. In *Proceedings of the 14th Triennial Congress of the Int. Ergonomics Association*. pp. 61-64.

Endsley, M. R. (2000b). Theretical Underpinnings of Situation Awareness: A critical Review. In Endsley, M. R. & Garland, D. J. (eds.). *Situation Awareness Analysis and Measurement*. pp. 3-33. New Jersey: Lawrence Erlbaum Associates, Inc.

Endsley, M. R. (2001). Designing for Situation Awareness in Complex System. In *Proceedings of the Second International Workshop of Symbiosis of Humans, Artifacts and Environment*, Kyoto, Japan.

Endsley, M. R. & Robertson, M. M. (2000). Training for Situation Awareness in Individuals and Teams. In Endsley, M. R. & Garland, D. J. (eds.). *Situation Awareness Analysis and Measurement*. pp. 349-365. New Jersey: Lawrence Erlbaum Associates, Inc.

Engstrom, T., Jonsson, D., Medbo, L. & Medbo, P. (1995). Inter-relations between product variant codifications and assembly work for flexible manufacturing in autonomous groups. *Journal of Materials Processing Technology*, 52 (1), 133-140.

Feng, Y.-H., Teng, T.-H. & Tan, A.-H. (2009). Modelling Situation Awareness for Context-aware Decision Support. *Expert Systems with Applications*, 36, 455-463.

Fidel, R. & Pejtersen, A. M. (2004). From information behaviour research to the design of information systems: the Cognitive Work Analysis framework. *Information Research: an international electronic journal*, 10 (1), 210-210.

Forza, C. (1995). The impact of information systems on quality performance: an empirical study. *International Journal of Operations & Production Management*, 15 (6), 69-83.

Galaktionova, Y., A., Ibraqimova, E., K. & Bekisheva, E., V. (2006). On product qualtiy control. *Fibre chemistry*, 38 (1), 79-82.

Geyer, T., Muller, H. J. & Krummenacher, J. (2008). Expectancies modulate attentinal capture by saliant color singletons. *Vision Research*, 48, 1315-1326.

Greenall, B., Associates, & AB Volvo. (2004). *Volvo Six Sigma Operational Black Belt - Module 1*. Milton Keynes: Greenall Barnard Associates & AB Volvo.

Greenough, R., Fakun, D. & Kay, J. (2001). Development of a digital manual for a manufacturing system - a case study. *Integrated Manufacturing Systems*, 12 (6), 387-393.

Greenough, R. M. & Tjahjono, B. (2007). An interactive electronic technical manual for an advanced aerospace assembly machine. *International Journal of Advanced Manufacturing Technology*, 33, 1045-1055.

Gustavsson, M. (2000). *Potential for Learning in Process Operator Work- A study of Operator Work in a Highly Automated Process Industry*. Linköping: Linköpings Universitet. (Linköping Studies in Education and Psychology No. 71).

Hartley, R. V. L. (1927). Transmission of Information. *The Bell System Technical Journal*, 7.

Hicks, B. J. (2007). Lean information management: Understanding and eliminating waste. *International Journal of Information Management*, 27 (4), 233-249.

Himma, K. E. (2007). The concept of information overload: A preliminary step in understanding the nature of a harmful information-related condition. *Ethics and Information Technology*, 9 (4), 259-272.

Hoffman, E. J. (1998). Visual Attention and Eye Movements. In Pashler, H. (ed.). *Attention*. Hove: Psychology Press Ltd.

Holden, J. D. (2001). Hawthorne effects and research into professional practice. *Journal of Evaluation in Clinical Practice*, 7 (1), 65-70.

Hong, K., Nagarajah, R., Iovenitti, P. & Dunn, M. (2007). A sociotechnical approach to achieve zero defect manufacturing of complex manual assemblies. *Human Factors and Ergonomics in Manufacturing*, 17 (2), 137-148.

Hornby, A. S. (1997). Oxford Advanced Learner's Dictionary of Current English. In Crowther, J. (ed.). Oxford: Oxford University Press.

Hoshino, O. (2003). Coherent interaction of dynamical attractors for object-based selective attention. *Biological Cybernetics*, 89 (2), 107-118.

Houston, B. K. (1969). Noise, task difficulty, and Stroop color-word performance. *Journal of Experimental Psychology*, 82, 403-404.

Högberg, D., Bäckstrand, G., Lämkull, D., De Vin, L. J., Case, K., Örtengren, R., Hanson, L. & Berlin, C. (2007). Towards Dynamic Ergonomics Analysis of Work Sequences in Virtual Environments. In *17th International Conf. on Flexible Automation and Intelligent Manufacturing (2007 FAIM)*, Philadelphia, USA. pp. 581-588.

Högberg, D., Bäckstrand, G., Lämkull, D., Hanson, L. & Örtengren, R. (2008). Industrial customisation of digital human modelling tools. *International Journal of Services Operations and Informatics*, 2 (1), 53-70.

IEA. (2009). *International Ergonomics Association* <<u>http://www.iea.cc/browse.php?contID=what_is_ergonomics></u>. [2009-09-22].

Jinsong, Z., Qifu, W., Li, W. & Yifang, Z. (2005). Configuration-oriented product modelling and knowledge management for made-to-order manufacturing enterprises. *International Journal of Advanced Manufacturing Technology*, 25 (1), 41-52.

Johnstone, D. & Tate, M. (2004). Bringing human information behaviour into information systems research: an application of systems modelling. *Information Research: an international electronic journal*, 9 (4), 191-191.

Jonides, J. (1981). Voluntary versus automatic control over the mind's eye's movement. In *Attention and Performance IX*. pp. 187-203. Lawrence Erlbaum.

Julien, H. & Barker, S. (2009). How high-school students find and evaluate scientific information: A basis for information literacy skills development. *Library & Amp; Information Science Research*, 31 (1), 12-17.

Järvelin, K. & Wilson, T. D. (2003). On conceptual models for information seeking and retrieval research. *Information Research: an international electronic journal*, 9 (1), 163-163.

Kaplan, F. & Hafner, V. V. (2006). The Challenges of Joint Attention. *Interaction Studies*, 7 (2), 135-169.

Kasvi, J. J. J., Vartiainen, A., Pulkkis, A. & Nieminen, M. (2000). The Role of Information Support Systems in the Joint Optimization of Work Systems. *Human Factors and Ergonomics in Manufacturing*, 20 (2), 193-211.

Kelly, D. & Fu, X. (2007). Eliciting better information need descriptions from users of information search systems. *Information Processing & Management*, 43, 30-46.

Kida, T., Wasaka, T., Nakata, H., Akatsuka, K. & Kakigi, R. (2006). Active attention modulates passive attention-related neural responses to sudden somatosensory input against a silent background. *Experimental Brain Research*, 175 (4), 609-617.

Kilov, H. & Sack, I. (2009). Mechanisms for Communication Between Business and IT Experts. *Computer Standards & Interfaces*, 31, 98-109.

Kuhlthau, C. C., Heinström, J. & Todd, R. J. (2008). The 'Information Search Process' Revisited: is the Model Still Useful? *Information Research*, 13 (4).

Kwasitsu, L. (2003). Information-seeking behavior of design, process, and manufacturing engineers. *Library and Information Science Research*, 25 (4), 459-476.

LaBerge, D. (1995). *Attentional Processing - The Brain's Art of Mindfulness*. London: Harvard University Press. (Perspectives in Cognitive Neuroscience).

LaBerge, D. & Samuels, J., S. (1974). Towards a Theory of Automatic Information Processing in Reading. *Cognitive Psychology*, 6, 293-323.

Lakusta, L. & Landau, B. (2005). Starting at the end: The importance of goals in spatial language. *Cognition*, 96, 1-33.

Langan-Fox, J., Wirth, A., Code, S. & Langfield-Smith, K. (2001). Analyzing shared and team mental models. *International Journal of Industrial Ergonomics*, 28 (2), 99-112.

Leech, N. & Onwuegbuzie, A. (2009). A typology of mixed methods research designs. *Quality & Quantity*, 43 (2), 265-276.

Liljegren, E. (2006). Usability in a medical technology context assessment of methods for usability evaluation of medical equipment. *International Journal of Industrial Ergonomics*, 36 (4), 345-352.

Lin, L., Drury, C. G. & Kim, S. W. (2001). Ergonomics and quality in paced assembly lines. *Human Factors and Ergonomics in Manufacturing*, 11 (4), 377-382.

Losee, R., M. (1997). A discipline independent definition of information. *Journal of the American Society for Information Science*, 48 (3), 254-269.

Losee, R. M. (1990). *The Science of Information; Measurement and Applications*. San Diego, California: Academic Press, Inc.

Lou, H. H. & Huang, Y. L. (2003). Hierarchical decision making for proactive quality control: system development for defect reduction in automotive coating operations. *Engineering Applications of Artificial Intelligence*, 16 (3), 237-250.

Lyons, E. (2002). Qualitative data Analysis: data Display Model. In Brewster, S. A., Hammond, S. & Fife-Schaw, C. (eds.). *Research Methods in Psychology*. London: SAGE Publications.

Machlup, F. (1980). *Knowledge: Its Creation, Distribution and Economic Significance*. Princeton, New Jersey: Prinston University Press.

Mack, L. R. (1994). Executive Summary. In Mack, L. R. & Nielsen, J. (eds.). *Usability Inspection Methods*. pp. 1-23. New York: John Wiley & Sons, Inc.

Marcum, J. W. (2002). Rethinking Information Literacy. Library Quarterly, 72 (1), 1-27.

Marsden, P. & Green, M. (1996). Optimising procedures in manufacturing systems. *International Journal of Industrial Ergonomics*, 17 (1), 43-51.

Meadow, C. T. & Weijing, Y. (1997). Measuring the impact of information: Defining the concepts. *Information Processing & Management*, 33 (6), 697-714.

Miller, B., Malloy, M. A., Masek, E. & Wild, C. (2001). Towards a Framework for Managing the Information Environment. *Information Knowledge Systems Management*, 2 (4), 359-385.

Millikan, R. G. (2001). The language-thought partnership: a bird's eye view. *Language & Communication*, 21, 157-166.
Minitab. (2009). <<u>http://www.minitab.com/en-SE/default.aspx</u>>.

Mital, A. & Pennathur, A. (2004). Advanced technologies and humans in manufacturing workplaces: an interdependent relationship. *International Journal of Industrial Ergonomics*, 33 (4), 295-313.

Moray, N. (1998). Identifying mental models of complex human-machine systems - an ergonomist-psychologist dialogue. *International Journal of Industrial Ergonomics*, 22 (4-5), 293-297.

Mursu, A., Luukkonen, I., Toivanen, M. & Korpela, M. (2007). Activity Theory in Information Systems Research and Practice: Theoretical Underpinnings for an Information Systems Development Model. *Information Research*, 12 (3).

Narasimhan, S., Tripathy, S. P. & Barrett, B. T. (2009). Loss of positonal information when tracking multiple moving dots: The role of visual memory. *Vision Research*, 49, 10-27.

Nationalencyklopedin. (2005). <<u>http://ne.se/jsp/search/article.jsp?i_art_id=150817></u>>. [Link available 2006-01-05

Nembhard, D. A. & Uzumeri, M. V. (2000). Experiential learning and forgetting for manual and cognitive tasks. *International Journal of Industrial Ergonomics*, 25 (4), 315-326.

Niebel, B. & Freivalds, A. (2003). Methods, Standards and Work Design. McGrew-Hill.

Nielsen, J. (1994). Heuristic Evaluation. In Mack, L. R. & Nielsen, J. (eds.). *Usability Inspection Methods*. pp. 25-62. New York: John Wiley & Sons, Inc.

Norman, A. D. (1990). *The Design of Everyday Things*. London, England: The MIT Press

Olshausen, B. A., Anderson, C. H. & Van Essen, D. C. (1993). A Neurobiological Model of Visual Attention and Invariant Pattern Recognition Based on Dynamic Routing of Information. *The Journal of Neuroscience*, 13 (11), 4700-4719.

Park, K. S. & Lee, J. I. (2007). A New Method for Estimating Human Errors Probabilities: AHP-SLIM. *Reliability Engineering and System Safety*, 93, 578-587.

Pashler, H. & Johnston, J. C. (1998). Attentional limitations in dual-task performance. In Pashler, H. (ed.). *Attention*. pp. 155-188. Hove: Psychology Press Ltd.

Patton, M. Q. (1980). *Qualitative Evaluation and Research Methods*. Newbury Park: SAGE Publications Inc.

Paz Barroso, M. & Wilson, J. R. (2000). Human Error and Disturbance Occurrence in Manufacturing Systems (HEDOMS): A Framework and a Toolkit for Practical Analysis. *Cognition, Technology & Work*, 2 (2), 51-61.

Pennathur, A. & Mital, A. (2003). Worker mobility and training in advanced manufacturing. *International Journal of Industrial Ergonomics*, 32 (6), 363-388.

Proctor, R. W. & Dutta, A. (1995). *Skill Acquisition and Human Performance*. Thousand Oaks: Sage Publications, Inc. (Advanced Psychology Texts).

Rafel, R. & Henik, A. (1994). The Neurology of Inhibition: Integrating Controlled and Automation Processes. In Dagenbach, D. & Carr, T. H. (eds.). *Inhibitory, Processes in Attention, Memory, and Language*. pp. 1-44. San Diego: Academic Press Inc.

Rasmussen, J., Pejtersen, A. M. & Schmit, K. (1990). Taxonomy for Cognitive Work Analysis *CiteSeer*.

Rasmussen, J., Pejtersen, M. A. & Goodstein, L. P. (1994). *Cognitive Systems Engineering*. John Wiley & sons, Inc.

Reason, J. (1990). Human Error. Cambridge University Press.

Reed, R. (2001). Aligning the plant floor to the board room. IIE Solutions, 33 (7), 34-37.

Rich, R. F. (1997). Measuring knowledge utilization: Processes and outcomes. *Knowledge & Policy*, 10 (3), 11-23.

Richard, C. M., Wright, R. D. & Ward, L. M. (2003). Goal-driven modulation of stimulus-driven attentional capture in multiple-cue displays. *Perception & Psychophysics*, 65 (6), 939-955.

Rookes, P. & Willson, J. (2000). *Perception : theory, development, and organisation*. London: Routledge.

Ryan, C. & Riggs, W. E. (1997). Redefining the Product Life Cycle: The Five-Element Product Wave. *Journal of Product Innovation Management*, 14 (3), 227-228.

Sander, P. C. & Brombacher, A. C. (2000). Analysis of quality information flows in the product creation process of high-volume consumer products. *International Journal of Production Economics*, 67 (1), 37-52.

Sanderson, P. M. (2003). Cognitive Work Analysis In Carroll, J. M. (ed.). *HCI Models, Theories, and Frameworks: Toward a Multidisciplinary Science*. pp. 225-264: Morgan Kaufmann Publishers.

Sawaragi, T. & Murasawa, K. (2001). Simulating behaviors of human situation awareness under high workloads. *Artificial Intelligence in Engineering*, 15 (4), 365-381.

Schubö. (2009). Salience detaction and attentinal capture. *Psychological Research*, 73, 233-243.

Shannon, E. C. & Weaver, W. (1963). *The Mathematical Theory of Communication*. University of Illinois Press.

Shaughnessy, J. J., Zechmeister, E. B. & Zechmeister, J. S. (2003). *Research Methods in Psychology*. New York: McGraw-Hill.

Sheridan, T. (2000). HCI in Supervisory Control: Twelve Dilemmas. In Elser, P. F., Kluwe, R. H. & Boussoffara, B. (eds.). *Human Error and System Design and Manufacturing*. pp. 1-12: Springer-Verlag London Limited.

Sheridan, T. (2002). *Humans and Automation:System Design and Research Issues*. Santa Monica: John Wiley & Sons, Inc., Publication. (HFES Issues in Human Factors and Ergonomics Series).

Sireteanu, R. & Rettenbach, R. (1994). Perceptual Lerning in Visual Search: Fast, Enduring, But Non-specific. *Vision Research*, 35 (14), 2037-2043.

Solme. (2008). *Analysis in AVIX - Step By Step* Solme AB. Skeppsbron 4. SE-411 21 Göteborg, SWEDEN.

Stake, R. E. (2000). Case Studies. In Denzin, N. K. & Lincoln, Y. S. (eds.). *Handbook of Qualitative Research*. Thousand Oaks: SAGE Publications Inc.

Sutcliffe, A. G. (1995). Human Computer Interface Design. Macmillan Press LTD.

Sörqvist, L. (1998). *Kvalitetsbristkostnader-Ett hjälpmedel för verksamhetsutveckling*. Lund: Studentlitteratur. In Swedish.

Tang, R. & Solomon, P. (1998). Toward an understanding of the dynamics of relevance judgment: An analysis of one person's search behavior. *Information Processing & Management*, 34 (2/3), 237-256.

Thagard, P. & Millgram, E. (1997). Inference to the Best Plan: A Coherence Theory of Decision. In Ram, A. & Leake, D. B. (eds.). *Goal-driven learning*. pp. 439-454. Cambridge, MA: MIT Press.

Vicente, K. J. (1999). *Cognitive work analysis, Toward Safe, Productive, and Healthy Computer-Based Work*. Mahwah: Lawrence Erlbaum

Vicente, K. J. (2006). The Human Factor. Routledge.

Vierck, E. & Miller, J. (2005). Direct selection by color for visual encoding. *Perception & Psychophsics*, 67 (3), 483-494.

Wang, Y. D. & Forgionne, G. (2006). A decision-theoretic approach to the evaluation of information retrieval systems. *Information Processing & Management*, 42 (4), 863-874.

Wharton, C. & Lewis, C. (1994). The Role of Psychological Theory in Usability Inspection Methods In Nielsen, J. & Mack, L. R. (eds.). *Usability Inspection Methods* New York: John Wiley & Sons, Inc. .

Wharton, C., Rieman, J., Lewis, C. & Polson, P. (1994). The Cognitive Walkthrough Method: A Practitioner's Guide. In Mack, L. R. & Nielsen, J. (eds.). *Usability Inspection Methods*. New York: John Wiley & Sons, Inc. Wickens, C. D. (1992). *Engineering Psychology and Human Performance*. . HarperCollins Publishers Inc.

Wickens, C. D. & Hollands, J. G. (2000). *Engineering Psychology and Human Performance*.: Prentice-Hall.

Wilson, T. D. (1997). Information behaviour: An interdisciplinary perspective. *Information Processing & Management*, 33 (4), 551-572.

Wilson, T. D. (1999). Models in information behaviour research. *Journal of Documentation*, 55 (3), 249-270.

Wilson, T. D. (2000). Human Information Behaviour. *Informing Science The International Journal of an Emerging Transdiscipline*, 3 (2), 49-56.

Wilson, T. D. (2001). Information Overload: Implications for Health-care Services. *Health Informatics Journal*, 7 (2), 112-117.

Wilson, T. D. (2006a). On User Studies and Information Needs. *Journal of Documentation*, 62 (6), 658-670.

Wilson, T. D. (2006b). A Re-examination of Information Seeking Behaviour in the Context of Activity Theory. *Information Research*, 11 (4).

Yantis, S. (1998). Control of Visual Attention. In Pashler, H. (ed.). *Attention*. pp. 223-252. Hove: Psychology Press Ltd.

Yoon, K. (2007). A study of interpersonal information seeking: the role of topic and comment in the articulation of certainty and uncertainty of information need. *Information Research*, 12 (2).

Appendices

Appendix A: Layout assembly



Basic Assembly part 2.



Basic Assembly part 1.



Appendix B: Case study 2 - Instructions









Appendix C: Questionnaire survey (translated from Swedish)

Survey case study regarding information handling

Place: Date:	Final Assembly, Volvo Powertrain, Skövde 2007-08-23
Study performed by:	Gunnar Bäckstrand

Please be as accurate as possible when you fill in the form. The purpose with the survey is to get an accurate picture of your work situation, from an information handling point-of-view.

Part 1	Background da	Background data		
1.1	Year of birth 19			
1.2	Female	Male		
1.3	I have worked at the company for	Dr:		
	0-1 year			
	1-2 years			
	2-5 years			
	5-10 years			
	More than 10 years			
1.4	I have worked in the assembly p	lant for:		
	0-1 year			
	1-2 years			
	2-5 years			
	5-10 years			
	More than 10 years			

1.5 I work mainly:

	Day		
	Eve	ning	
	Nigh	nt	
	Wee	ekends	
	Othe	er	Пном:
1.6	My main assignme	ent is:	
		Assembly	
		Quality	
		Management	
		Other	□
1.7	I have worked on	station S0800 o	during the case study:
		Yes	
		No	
1.8	I have worked on	station S1100 o	during the case study:
		Yes	
		No	

If "No" in question 1.7 and 1.8, continue to question 2.8

Part 2	Information	
2.1	Do you believe that the colour coding the quality in a positive way?	g used on station S0800 and S1100 influenced
	To a very high degree	
	To a rather high degree	
	In some way	
	In a rather insignificant way	y 🗆
	Not at all	
	Negatively	
2.2	Did you feel secure with the colour c	oding that was used in station 800 and 1100?
	Yes]
	No]
	Sometimes When?]
	Why?	
	Sometimes not When?]
	Why?	
2.3	Is it your opinion that the information information you needed to perform a	n you received via colour coding was the a correct assembly task?
	Yes]
	No]
	Sometimes]
	Sometimes not Why?]
2.4	Using colour coding on station S080	0 and S1100 worked very well.

Disagree Entirely

Neither or

Agree Entirely

2.5	It was faster to use colour coding when I assembled.			
	Agree Entirely	Neither or	Disa	agree Entirely
2.6	Colour coding made it e next engine.	asier for me to	better keep tra	ack of what to assembly in the
	Agree Entirely	Neither or	Disa	agree Entirely
2.7	By using colour coding i influencing the quality n	t was possible egatively.	to take away s	some of "blue buttons" without
	Agree Entirely	Neither or	Disa	agree Entirely
2.8	Which information sourc	e would you sa arding what to a	ay is the most assemble in a	important when it comes to specific workstation?
	My colleagues			
	Computer screen on carrier			
	The part to assemble			
	The engine			
	Paper instructions			
	Other sources Which?			-

Which information source would you say should be the most important when it comes to viewing information regarding what to assembly in a specific workstation?

My colleagues	
Computer screen on carrier	
The part to assemble	
The engine	
Paper instructions	
Other sources Which?	

2.10	According to you WHEN do you need the inform most important.		I need the inforr	rmation? Rank them, where 1 is the		
	When the carr	ier has stopped on a st	ation			
	Before the car	rier has stopped on a s	tation			
	When it is time	e to assemble the part o	on the engine			
2.11	Do you believ	ve that information	can affect prod	uctivity?		
		Yes, a lot				
		Yes, some				
		No				
		Don't know				
2.12	I believe that	the information I r	eceive can affeo	ct the quality		
		Positively				
		Negatively				
		Not at all				
2.13	I believe that	too much informat	tion can affect th	ne quality		
		Positively				
		Negatively				
		Not at all				
2.14	Can the infor the quality?	mation you have a	ccess to, via IT-	-systems and instructions, influence		
		Yes				
		No				
		Don't know				

2.15	The information you have account	ess to is a:		
	Support			
	An obstacle			
	Annoying			
	A prerequisite for a	successful a	ssembly	
2.16	Is it your opinion that the infor	mation is a	ccessible when you most need it?	_
	Often			
	Sometimes			
	Seldom			
2.17	Is it easy to find the informatio	n when yo	u need it?	_
	Often			
	Sometimes			
	Seldom			
2.18	Is it easy to see the informatio	n present o	on the computer screens?	_
	Yes			
	No Why?:			
2.19	Is it your opinion that for exam information for you to quickly f	ple "20824 ind the par	1541 SERVOPUMP" is enough t in the material racks?	_
	Yes			
	No			
	Sometimes When?			
	Sometimes not When?			

2.20 You are working at station S0800 and are to assembly a part (for example the servo pump), where do you look for/find information regarding the part your are about to assembly?

	Get info from team leader		
	Engine		
	Computer screen		
	From colleague		
	Other When?		
2.21	Is there a quality risk with "prework"?		
	Yes		
	No		
	Sometimes When?		
2.22	The best way to get information is via the	e text on a computer screen.	
	Agree Entirely Neither or	Disagree Entirely	
2.23	Which of the following statements, accor	ding to your opinion, is correct.	
	"Colour coding is easier to unde	rstand than text on a computer screen"	
	"20824541 SERVOPUMP is eas	sier to understand then colour coding"	
	"Colour coding does not make n	ny job easier"	
	" Colour coding make my job ea	sier"	
	"None of the above"		
2.24	If I had access to colour coding on the st assemble.	tation where I work it would go fa	aster to
	Agree Entirely Neither or	Disagree Entirely	
2.25	I believe that colour coding could be use	d in more stations.	
	Agree Entirely Neither or	Disagree Entirely	

2.26 Some statements regarding your work.

	Totally True	Partly True	Partly False	Totally False	Don't know
The assembly goes faster when I "prework"					
"Prework" does not affect quality					
The risk to assemble wrong part increases with "prework"					
I look at the approaching engine and thereby see what I should assemble					
With right information at the right time could "prework" make my work easier					

2.27 The information I receive during the assembly contributes to the feeling of control regarding my work situation.

	Agree Entirely	Neither or	Disagree Entirely
2.28	When the carrie first to see what	r arrives at the works to assemble.	station I always look at the computer screen
	Agree Entirely	Neither or	Disagree Entirely

Other remarks: